THE ONE HUNDRED SIXTEENTH DAY

CARSON CITY (Thursday), June 2, 2011

Assembly called to order at 11:28 a.m.

Mr. Speaker presiding.

Roll called.

All present.

Prayer by the Chaplain, Reverend Bruce Henderson.

Each time we come together, we begin by talking to God. I thought it would be appropriate today to allow God to speak to us. Let us listen as He speaks from the Book of Jeremiah:

"For I know the plans I have for you, plans to prosper you and not to harm you, plans to give you hope and a future. Then you will call upon me and come and pray to me, and I will listen to you. You will seek me and find me when you seek me with all your heart." (Jeremiah 29:11-13)

Thank You, Lord.

AMEN.

Pledge of allegiance to the Flag.

Assemblywoman Kirkpatrick moved that further reading of the Journal be dispensed with, and the Speaker and Chief Clerk be authorized to make the necessary corrections and additions.

Motion carried.

REPORTS OF COMMITTEES

Mr. Speaker:

Your Committee on Ways and Means has had under consideration various budgets for the Department of Business and Industry, and begs leave to report back that the following accounts have been closed by the Committee:

Business and Industry Administration (101-4681)

Division of Industrial Relations (210-4680)

Occupational Safety and Health Enforcement (210-4682)

Safety Consultation and Training (210-4685)

Mine Safety and Training (210-4686)

Housing Division (503-3841)

Low Income Housing Trust Fund (101-3838)

Special Housing Assistance (101-3839)

Weatherization (101-4865)

Industrial Development Bonds (101-4683)

Manufactured Housing (271-3814)

Mobile Home Lot Rent Subsidy (630-3842)

Mobile Home Parks (271-3843)

Manufactured Housing Education/Recovery (271-3847)

JUNE 2, 2011 — DAY 116

5679

Also, your Committee on Ways and Means has had under consideration various budgets for the Division of Mental Health and Developmental Services of the Department of Health and Human Services, and begs leave to report back that the following accounts have been closed by the Committee:

MHDS Administration (101-3168)

MH Info Systems (101-3164)

Alcohol Tax Program (101-3255)

Substance Abuse Prev/Treatment (101-3170)

Family Preservation Prgm (101-3166)

Sierra Regional Center (101-3280)

Desert Regional Center (101-3279)

Rural Regional Center (101-3167)

Rural Clinics (101-3648)

No NV Adult MH Svc (101-3162)

So NV Adult MH Svc (101-3161)

Lake's Crossing (101-3645)

DEBBIE SMITH, Chair

Mr. Speaker:

Your Committee on Commerce and Labor, to which was referred Senate Bill No. 60, has had the same under consideration, and begs leave to report the same back with the recommendation: Do pass.

KELVIN ATKINSON, Chair

Mr. Speaker:

Your Committee on Ways and Means, to which were referred Senate Bills Nos. 443, 446, has had the same under consideration, and begs leave to report the same back with the recommendation: Do pass.

Also, your Committee on Ways and Means, to which was rereferred Assembly Bill No. 380, has had the same under consideration, and begs leave to report the same back with the recommendation: Amend, and do pass as amended.

Also, your Committee on Ways and Means, to which was rereferred Assembly Bill No. 503, has had the same under consideration, and begs leave to report the same back with the recommendation: Amend, and do pass as amended.

DEBBIE SMITH, Chair

COMMUNICATIONS

MESSAGES FROM THE GOVERNOR

OFFICE OF THE GOVERNOR

June 1, 2011

April 4, 2011

SPEAKER JOHN OCEGUERA, Nevada State Assembly, 401 SOUTH CARSON STREET, CARSON CITY, NV 89701

RE: Assembly Bill 135 of the 76th Legislative Session

DEAR MR. SPEAKER:

I am herewith forwarding to you, for filing within the constitutional time limit and without my approval, Assembly Bill 135, which is entitled:

AN ACT relating to probation; revising provisions concerning violations of probation; and providing other matters properly relating thereto.

This bill relates to the authority of courts to, upon determination that a person has violated a condition of probation, continue or revoke the probation or suspension of sentence. More specifically, the bill limits courts' discretion to revoke the probation and suspension of sentence to cases where: (1) imprisonment is necessary to protect the community from further criminal activity by the probationer; (2) the probationer is in need of treatment which can most effectively be provided if he or she is imprisoned; (3) the seriousness of the violation or the totality of violations by the probationer warrant revocation of probation and suspension of the sentence; or (4) the violation demonstrates the probationer cannot be supervised pursuant to practices and policies governing probation.

The bill further precludes courts from revoking probation and the suspension of the sentence based on the probationer's failure to pay court imposed administrative assessments, fees and expenses. In addition, before revoking probation and the suspension of the sentence, the bill requires courts to make findings to support the reasons for the revocation and state them on the record.

The requirement that courts state on the record the reasons for revocation of probation and the suspension of a sentence is reasonable. The limits imposed on the discretion of courts to revoke probation and the suspension of a sentence, however, are not. The effective administration of justice requires the proportionate enforcement of criminal sentences. Proportionality is not, though, easily obtained through the application of categorical methodology. Instead, courts require flexibility and discretion in fashioning the appropriate response to a probation violation. Thus, courts have traditionally been granted latitude in determining the appropriate mechanism by which such sentences are enforced. The revocation of probation and the suspension of a sentence are tools often employed toward that end.

Insofar as this bill limits the cases in which these enforcement mechanisms are available, it undermines the ability of courts to effectively enforce sentences. The Legislature's attempt to categorically define the circumstances under which revocation of probation and the suspension of a sentence are appropriate fails to adequately capture the scope of cases where such revocation is appropriate; therefore, I veto this bill and return it to you without my approval.

Sincere regards, BRIAN SANDOVAL Governor

OFFICE OF THE GOVERNOR

June 1, 2011

SPEAKER JOHN OCEGUERA, Nevada State Assembly, 401 SOUTH CARSON STREET, CARSON CITY, NV 89701

RE: Assembly Bill 253 of the 76th Legislative Session

DEAR MR. SPEAKER:

I am herewith forwarding to you, for filing within the constitutional time limit and without my approval, Assembly Bill 253, which is entitled:

AN ACT relating to occupational safety; revising certain fines for willful violations of the Nevada Occupational Safety and Health Act; authorizing citations and fines for violation of a settlement agreement; providing for a survey of salaries of safety and mechanical inspectors; and providing other matters properly relating thereto.

This bill relates to occupational safety. It proposes significant increases to the fines that may be imposed for willful violations of the Nevada Occupational Safety and Health Act (OSHA). The bill also revises the punishment for a willful violation that results in the death of an employee.

There is some merit to the argument that the bills increased fines may help reduce the number of willful OSHA violations occurring in Nevada's workplaces. However, a more innovative and proactive approach is warranted to improve workplace safety and change behavior before it

JUNE 2. 2011 — DAY 116

5681

results in workplace injuries or death. I therefore exercise my constitutional grant of authority to veto AB 253 and return it to you without my approval.

Sincere regards, BRIAN SANDOVAL Governor

OFFICE OF THE GOVERNOR

CARSON CITY, NEVADA, June 1, 2011

SPEAKER JOHN OCEGUERA, Nevada State Assembly, 401 SOUTH CARSON STREET, CARSON CITY, NV 89701

RE: Assembly Bill 254 of the 76th Legislative Session

DEAR MR. SPEAKER:

I am herewith forwarding to you, for filing within the constitutional time limit and without my approval, Assembly Bill 254, which is entitled:

AN ACT relating to occupational safety; revising provisions governing the grounds for the issuance of a citation for certain occupational safety and health violations; providing for the issuance of a citation for certain occupational safety and health violations upon a determination by the Administrator of the Division of Industrial Relations of the Department of Business and Industry or the Administrator's authorized representative that any employee has access to a hazard; and providing other matters properly relating thereto.

This bill relates to occupational safety. Like Assembly Bill 253, which I have already vetoed, Assembly Bill 254 unnecessarily increases the complexity and cost of operating a business in Nevada. The bill expands the scope of behavior for which a citation may be issued to an employer for violating the Nevada Occupational Safety and Health Act (OSHA) by allowing the issuance of a citation based upon a determination that an employee "has access to a hazard" in the workplace.

Legislation passed in 2009 has significantly improved jobsite safety and enforcement of OSHA violations in Nevada, and I am a strong supporter of continued improvement in these areas. However, this bill creates ambiguous and misguided new requirements that will be difficult to enforce and burdensome to comply with. Because this bill will not proactively improve the conditions and safety of our workplaces, I veto it and return it to you without my approval.

Sincere regards, BRIAN SANDOVAL Governor

OFFICE OF THE GOVERNOR

June 1, 2011

SPEAKER JOHN OCEGUERA, Nevada State Assembly, 401 SOUTH CARSON STREET, CARSON CITY, NV 89701

RE: Assembly Bill 456 of the 76th Legislative Session

DEAR MR. SPEAKER:

I am herewith forwarding to you, for filing within the constitutional time limit and without my approval, Assembly Bill 456, which is entitled:

AN ACT relating to education; authorizing certain pupils to receive a standard high school diploma without passing all subject areas of the high school proficiency examination under certain circumstances; authorizing the board of trustees of a school district to adopt a policy that allows certain pupils enrolled in high school the opportunity to make up credit; authorizing a juvenile court to impose certain orders

against the parent or legal guardian of a child who is adjudicated in need of supervision because the child is a habitual truant; revising provisions governing employment of minors; and providing other matters properly relating thereto.

This bill alters requirements for graduation from high school. In order to receive a standard high school diploma, students must pass all portions of the Nevada High School Proficiency Examination (HSPE) and meet certain other district and state requirements. Approval of Assembly Bill 456 would allow students to graduate from high school even if they continue to fail one subject area of the HSPE, so long as they earn at least a 2.75 grade point average, satisfy minimum attendance requirements, do not have any pending disciplinary actions, and obtain a cumulative passing score on the HSPE.

Supporters of the bill point out that some students cannot pass the math portion of the HSPE, but these students are otherwise proficient in all other subject areas and do well in school. While personally compelling, this argument lacks a broad policy implication; if the number of students is as small as has been represented, other remedies may exist besides a wholesale change in graduation requirements. The bill does not address the fact that Nevada's graduation rates and grade-level performance are amongst the worst in the nation. It similarly fails to address why an achievement gap exists for a few students on certain portions of the HSPE. Other measures already enacted this legislative session do more to change the status quo. For example, Assembly Bill 290 will allow the principal of a high school to postpone administration of the math and/or science portion of the HSPE for pupils who are not academically ready. These students will be enrolled in appropriate course work and participate in a program designed to help students pass the exam.

Although this bill may allow more students to graduate from high school, it represents diminished expectations for our students and lower standards for obtaining a high school diploma in Nevada. In my State of the State address, I said that our education system emphasizes too many of the wrong things. AB 456 is another example of this paradigm and would send the wrong message to our students.

I am committed to improving our education system and enhancing student achievement. Because this bill provides a way to hide or ignore a student achievement problem, rather than to fix it, I veto it and return it to you without my approval.

Sincere regards, BRIAN SANDOVAL Governor

MESSAGES FROM THE SENATE

SENATE CHAMBER, Carson City, June 1, 2011

To the Honorable the Assembly:

I have the honor to inform your honorable body that the Senate amended, and on this day passed, as amended, Assembly Bill No. 225, Amendment No. 860; Assembly Bill No. 229, Amendment No. 861, and respectfully requests your honorable body to concur in said amendments.

Also, I have the honor to inform your honorable body that the Senate on this day concurred in the Assembly Amendment No. 681 to Senate Bill No. 223; Assembly Amendment No. 628 to Senate Bill No. 236; Assembly Amendment No. 682 to Senate Bill No. 299; Assembly Amendment No. 672 to Senate Bill No. 323; Assembly Amendment No. 711 to Senate Bill No. 339.

SHERRY L. RODRIGUEZ
Assistant Secretary of the Senate

MOTIONS, RESOLUTIONS AND NOTICES

Assemblyman Atkinson moved that the action whereby the Assembly concurred to Senate Amendment No. 658 to Assembly Bill No. 199 be reconsidered.

Motion carried.

UNFINISHED BUSINESS

CONSIDERATION OF SENATE AMENDMENTS

Assembly Bill No. 199.

The following Senate amendment was read:

Amendment No. 658.

AN ACT relating to the practice of pharmacy; revising provisions governing the authority of a registered pharmacist to collaborate with a practitioner for the implementation, monitoring and modification of drug therapy; authorizing the State Board of Pharmacy to establish regulations relating to collaborative pharmacy practice; revising provisions governing the use of the letters "Rx" and "RX" by certain persons; revising provisions relating to the authority of a registered pharmacist to possess and administer controlled substances and dangerous drugs under certain circumstances; providing a penalty; and providing other matters properly relating thereto.

Legislative Counsel's Digest:

Existing law authorizes a registered pharmacist to collaborate with a practitioner to engage in the implementation and modification of drug therapy for a patient at a licensed medical facility or licensed pharmacy. (NRS 639.0124) Section 1 of this bill prescribes requirements for written guidelines and protocols which must be developed by a pharmacist who collaborates with a practitioner and requires those guidelines and protocols to be approved by the State Board of Pharmacy. Section 1 also authorizes the written guidelines and protocols to set forth provisions for a pharmacist to implement, monitor and modify the drug therapy of a patient in a medical facility or a setting [other than] that is affiliated with a licensed medical facility. [or licensed pharmacy.] Sections 10.3 and 10.7 of this bill revise the authority of a pharmacist to possess and administer controlled substances and dangerous drugs under certain circumstances for the care of a patient in accordance with the written guidelines and protocols developed pursuant to section 1.

Existing law prohibits a person operating a business from using the letters "Rx" or "RX" if the person does not have a license from the Board. (NRS 639.230) Section 10 of this bill authorizes persons who are

not subject to the laws governing the practice of pharmacy to use those letters if the person obtains approval from the Board.

A person who violates any provision of chapter 639 of NRS governing pharmacists and pharmacies, including any provision of this bill, is guilty of a misdemeanor. (NRS 639.310)

THE PEOPLE OF THE STATE OF NEVADA, REPRESENTED IN SENATE AND ASSEMBLY, DO ENACT AS FOLLOWS:

Section 1. Chapter 639 of NRS is hereby amended by adding thereto a new section to read as follows:

- 1. Written guidelines and protocols developed by a registered pharmacist in collaboration with a practitioner which authorize the implementation, monitoring and modification of drug therapy:
- (a) May authorize a pharmacist to order and use the findings of laboratory tests and examinations.
- (b) May provide for implementation, monitoring and modification of drug therapy for a patient receiving care [in]:
 - (1) In a licensed medical facility; or [; if]
- (2) If developed to ensure continuity of care for a patient, in any setting [outside] that is affiliated with a medical facility where the patient is receiving care. A pharmacist who modifies a drug therapy of a patient receiving care in a setting that is affiliated with a medical facility shall, within 72 hours after implementing or modifying the drug therapy, provide written notice of the implementation or modification of the drug therapy to the collaborating practitioner or enter the appropriate information concerning the drug therapy in an electronic patient record system shared by the pharmacist and the collaborating practitioner.
- (c) <u>Must state the conditions under which a prescription of a practitioner relating to the drug therapy of a patient may be changed by the pharmacist without a subsequent prescription from the practitioner.</u>
 - (d) Must be approved by the Board.
 - 2. The Board may adopt regulations which:
- (a) Prescribe additional requirements for written guidelines and protocols developed pursuant to this section; and
- (b) Set forth the process for obtaining the approval of the Board of such written guidelines and protocols.
 - **Sec. 2.** (Deleted by amendment.)
 - Sec. 3. (Deleted by amendment.)
 - **Sec. 4.** (Deleted by amendment.)
 - **Sec. 5.** (Deleted by amendment.)
 - **Sec. 6.** (Deleted by amendment.)
 - **Sec. 7.** (Deleted by amendment.)

- **Sec. 8.** (Deleted by amendment.)
- **Sec. 9.** NRS 639.0124 is hereby amended to read as follows:

639.0124 "Practice of pharmacy" includes, but is not limited to, the:

- 1. Performance or supervision of activities associated with manufacturing, compounding, labeling, dispensing and distributing of a drug, including the receipt, handling and storage of prescriptions and other confidential information relating to patients.
 - 2. Interpretation and evaluation of prescriptions or orders for medicine.
 - 3. Participation in drug evaluation and drug research.
- 4. Advising of the therapeutic value, reaction, drug interaction, hazard and use of a drug.
 - 5. Selection of the source, storage and distribution of a drug.
- 6. Maintenance of proper documentation of the source, storage and distribution of a drug.
- 7. Interpretation of clinical data contained in a person's record of medication.
- 8. Development of written guidelines and protocols in collaboration with a practitioner which are intended for a patient in a licensed medical facility or in a setting [outside] that is affiliated with a medical facility where the patient is receiving care and which authorize the implementation, monitoring and modification of drug therapy. The written guidelines and protocols [may authorize a pharmacist to order and use the findings of laboratory tests and examinations.] must comply with section 1 of this act.
- 9. Implementation and modification of drug therapy in accordance with the authorization of the prescribing practitioner for a patient in a pharmacy in which drugs, controlled substances, poisons, medicines or chemicals are sold at retail.
- → The term does not include the changing of a prescription by a pharmacist or practitioner without the consent of the prescribing practitioner, except as otherwise provided in NRS 639.2583.
 - **Sec. 10.** NRS 639.230 is hereby amended to read as follows:
- 639.230 1. A person operating a business in this State shall not use [the letters "Rx" or "RX" or] the word "drug" or "drugs," "prescription" or "pharmacy," or similar words or words of similar import, without first having secured a license from the Board. A person operating a business in this State which is not otherwise subject to the provisions of this chapter shall not use the letters "Rx" or "RX" without the approval of the Board. The Board must not unreasonably deny approval of the use of the letters "Rx" or "RX" by any person but may deny approval upon a determination that:
- (a) The person is subject to the provisions of this chapter but has not secured a license from the Board; or

(b) The use of the letters "Rx" or "RX" by the person is confusing or misleading to or threatens the health or safety of the residents of this State.

- 2. Each license must be issued to a specific person and for a specific location and is not transferable. The original license must be displayed on the licensed premises as provided in NRS 639.150. The original license and the fee required for reissuance of a license must be submitted to the Board before the reissuance of the license.
- 3. If the owner of a pharmacy is a partnership or corporation, any change of partners or corporate officers must be reported to the Board at such a time as is required by a regulation of the Board.
- 4. Except as otherwise provided in subsection 6, in addition to the requirements for renewal set forth in NRS 639.180, every person holding a license to operate a pharmacy must satisfy the Board that the pharmacy is conducted according to law.
- 5. Any violation of any of the provisions of this chapter by a managing pharmacist or by personnel of the pharmacy under the supervision of the managing pharmacist is cause for the suspension or revocation of the license of the pharmacy by the Board.
 - 6. The provisions of this section do not prohibit [a]:
- (a) A Canadian pharmacy which is licensed by the Board and which has been recommended by the Board pursuant to subsection 4 of NRS 639.2328 for inclusion on the Internet website established and maintained pursuant to subsection 9 of NRS 223.560 from providing prescription drugs through mail order service to residents of Nevada in the manner set forth in NRS 639.2328 to 639.23286, inclusive [...]; or
- (b) A registered pharmacist or practitioner from collaborating in the implementation, monitoring and modification of drug therapy pursuant to guidelines and protocols approved by the Board.

Sec. 10.3. NRS 453.026 is hereby amended to read as follows:

453.026 "Agent" means a pharmacist [] who cares for a patient of a prescribing practitioner in a medical facility or in a setting that is affiliated with a medical facility where the patient is receiving care in accordance with written guidelines and protocols developed and approved pursuant to section 1 of this act, a licensed practical nurse or registered nurse who cares for a patient of a prescribing practitioner in a medical facility or an authorized person who acts on behalf of or at the direction of and is employed by a manufacturer, distributor, dispenser or prescribing practitioner. The term does not include a common or contract carrier, public warehouseman or employee of the carrier or warehouseman.

Sec. 10.7. NRS 454.213 is hereby amended to read as follows:

454.213 A drug or medicine referred to in NRS 454.181 to 454.371, inclusive, may be possessed and administered by:

- 1. A practitioner.
- 2. A physician assistant licensed pursuant to chapter 630 or 633 of NRS, at the direction of his or her supervising physician or a licensed dental hygienist acting in the office of and under the supervision of a dentist.
- 3. Except as otherwise provided in subsection 4, a registered nurse licensed to practice professional nursing or licensed practical nurse, at the direction of a prescribing physician, physician assistant licensed pursuant to chapter 630 or 633 of NRS, dentist, podiatric physician or advanced practitioner of nursing, or pursuant to a chart order, for administration to a patient at another location.
- 4. In accordance with applicable regulations of the Board, a registered nurse licensed to practice professional nursing or licensed practical nurse who is:
- (a) Employed by a health care agency or health care facility that is authorized to provide emergency care, or to respond to the immediate needs of a patient, in the residence of the patient; and
- (b) Acting under the direction of the medical director of that agency or facility who works in this State.
- 5. Except as otherwise provided in subsection 6, an intermediate emergency medical technician or an advanced emergency medical technician, as authorized by regulation of the State Board of Pharmacy and in accordance with any applicable regulations of:
- (a) The State Board of Health in a county whose population is less than 100,000:
- (b) A county board of health in a county whose population is 100,000 or more; or
- (c) A district board of health created pursuant to NRS 439.362 or 439.370 in any county.
- 6. An intermediate emergency medical technician or an advanced emergency medical technician who holds an endorsement issued pursuant to NRS 450B.1975, under the direct supervision of a local health officer or a designee of the local health officer pursuant to that section.
- 7. A respiratory therapist employed in a health care facility. The therapist may possess and administer respiratory products only at the direction of a physician.
- 8. A dialysis technician, under the direction or supervision of a physician or registered nurse only if the drug or medicine is used for the process of renal dialysis.
- 9. A medical student or student nurse in the course of his or her studies at an approved college of medicine or school of professional or practical nursing, at the direction of a physician and:
 - (a) In the presence of a physician or a registered nurse; or

- (b) Under the supervision of a physician or a registered nurse if the student is authorized by the college or school to administer the drug or medicine outside the presence of a physician or nurse.
- → A medical student or student nurse may administer a dangerous drug in the presence or under the supervision of a registered nurse alone only if the circumstances are such that the registered nurse would be authorized to administer it personally.
 - 10. Any person designated by the head of a correctional institution.
- 11. An ultimate user or any person designated by the ultimate user pursuant to a written agreement.
- 12. A nuclear medicine technologist, at the direction of a physician and in accordance with any conditions established by regulation of the Board.
- 13. A radiologic technologist, at the direction of a physician and in accordance with any conditions established by regulation of the Board.
- 14. A chiropractic physician, but only if the drug or medicine is a topical drug used for cooling and stretching external tissue during therapeutic treatments.
- 15. A physical therapist, but only if the drug or medicine is a topical drug which is:
- (a) Used for cooling and stretching external tissue during therapeutic treatments; and
 - (b) Prescribed by a licensed physician for:
 - (1) Iontophoresis; or
 - (2) The transmission of drugs through the skin using ultrasound.
- 16. In accordance with applicable regulations of the State Board of Health, an employee of a residential facility for groups, as defined in NRS 449.017, pursuant to a written agreement entered into by the ultimate user.
- 17. A veterinary technician at the direction of his or her supervising veterinarian.
- 18. In accordance with applicable regulations of the Board, a registered pharmacist who:
- (a) Is trained in and certified to carry out standards and practices for immunization programs;
- (b) Is authorized to administer immunizations pursuant to written protocols from a physician; and
- (c) Administers immunizations in compliance with the "Standards of Immunization Practices" recommended and approved by the United States Public Health Service Advisory Committee on Immunization Practices.
- 19. <u>A registered pharmacist pursuant to written guidelines and protocols developed and approved pursuant to section 1 of this act.</u>

<u>20.</u> A person who is enrolled in a training program to become a physician assistant licensed pursuant to chapter 630 or 633 of NRS, dental hygienist, intermediate emergency medical technician, advanced emergency medical technician, respiratory therapist, dialysis technician, nuclear medicine technologist, radiologic technologist, physical therapist or veterinary technician if the person possesses and administers the drug or medicine in the same manner and under the same conditions that apply, respectively, to a physician assistant licensed pursuant to chapter 630 or 633 of NRS, dental hygienist, intermediate emergency medical technician, advanced emergency medical technician, respiratory therapist, dialysis technician, nuclear medicine technologist, radiologic technologist, physical therapist or veterinary technician who may possess and administer the drug or medicine, and under the direct supervision of a person licensed or registered to perform the respective medical art or a supervisor of such a person.

Sec. 11. This act becomes effective upon passage and approval for the purpose of adopting regulations and on October 1, 2011, for all other purposes.

Assemblyman Atkinson moved that the Assembly do not concur in the Senate Amendment No. 658 to Assembly Bill No. 199.

Motion carried.

Bill ordered transmitted to the Senate.

Assembly Bill No. 225.

The following Senate amendment was read:

Amendment No. 860.

AN ACT relating to educational personnel; requiring certain teachers and administrators who receive unsatisfactory evaluations to serve an additional probationary period; authorizing certain employees to request an expedited hearing under certain circumstances; and providing other matters properly relating thereto.

Legislative Counsel's Digest:

Existing law provides that each teacher and administrator who is employed by a school district in this State must serve a 2-year probationary period, unless the second year of the probationary period is waived by the superintendent of schools of the school district or the superintendent's designee. A probationary employee who completes his or her probationary period and receives a notice of reemployment from the school district becomes a postprobationary employee in the ensuing year of employment. (NRS 391.3197) Existing law also provides that a postprobationary teacher or administrator must be evaluated at least once each year. (NRS 391.3125, 391.3127) **Section 1** of this bill provides that a postprobationary teacher or

administrator who receives an unsatisfactory evaluation , or any other equivalent evaluation which designates his or her overall performance as below average, for 2 consecutive school years shall be deemed to be a probationary employee and must serve an additional probationary period. Section 4 of this bill provides that the provisions of section 1 [do not apply iff] are not superseded by the terms of a collective bargaining agreement. Section 5 of this bill authorizes a teacher or administrator who is deemed to be a probationary employee pursuant to section 1 and who receives notice that he or she will be dismissed before the completion of the current school year to request an expedited hearing pursuant to the expedited hearing procedures established by the American Arbitration Association.

THE PEOPLE OF THE STATE OF NEVADA, REPRESENTED IN SENATE AND ASSEMBLY, DO ENACT AS FOLLOWS:

Section 1. Chapter 391 of NRS is hereby amended by adding thereto a new section to read as follows:

A postprobationary employee who receives an unsatisfactory evaluation pursuant to NRS 391.3125 or 391.3127, as applicable, or any other equivalent evaluation designating his or her overall performance as below average, for 2 consecutive school years shall be deemed to be a probationary employee for the purposes of NRS 391.311 to 391.3197, inclusive, and must serve an additional probationary period in accordance with the provisions of NRS 391.3197.

- **Sec. 2.** NRS 391.311 is hereby amended to read as follows:
- 391.311 As used in NRS 391.311 to 391.3197, inclusive, *and section 1 of this act*, unless the context otherwise requires:
- 1. "Administrator" means any employee who holds a license as an administrator and who is employed in that capacity by a school district.
- 2. "Board" means the board of trustees of the school district in which a licensed employee affected by NRS 391.311 to 391.3197, inclusive, *and section 1 of this act* is employed.
- 3. "Demotion" means demotion of an administrator to a position of lesser rank, responsibility or pay and does not include transfer or reassignment for purposes of an administrative reorganization.
 - 4. "Immorality" means:
- (a) An act forbidden by NRS 200.366, 200.368, 200.400, 200.508, 201.180, 201.190, 201.210, 201.220, 201.230, 201.265, 201.540, 201.560, 207.260, 453.316 to 453.336, inclusive, 453.337, 453.338, 453.3385 to 453.3405, inclusive, 453.560 or 453.562; or
- (b) An act forbidden by NRS 201.540 or any other sexual conduct or attempted sexual conduct with a pupil enrolled in an elementary or secondary

school. As used in this paragraph, "sexual conduct" has the meaning ascribed to it in NRS 201.520.

- 5. "Postprobationary employee" means an administrator or a teacher who has completed the probationary period as provided in NRS 391.3197 and has been given notice of reemployment. The term does not include a person who is deemed to be a probationary employee pursuant to section 1 of this act.
 - 6. "Probationary employee" means [an]:
- (a) An administrator or a teacher who is employed for the period set forth in NRS 391.3197 [-]; and
- (b) A person who is deemed to be a probationary employee pursuant to section 1 of this act.
- 7. "Superintendent" means the superintendent of a school district or a person designated by the board or superintendent to act as superintendent during the absence of the superintendent.
- 8. "Teacher" means a licensed employee the majority of whose working time is devoted to the rendering of direct educational service to pupils of a school district.
 - **Sec. 3.** NRS 391.3115 is hereby amended to read as follows:
- 391.3115 1. The demotion, suspension, dismissal and nonreemployment provisions of NRS 391.311 to 391.3197, inclusive, *and section 1 of this act* do not apply to:
 - (a) Substitute teachers; or
 - (b) Adult education teachers.
- 2. The provisions of NRS 391.311 to 391.3194, inclusive, do not apply to a teacher whose employment is suspended or terminated pursuant to subsection 3 of NRS 391.120 or NRS 391.3015 for failure to maintain a license in force.
- 3. A licensed employee who is employed in a position fully funded by a federal or private categorical grant or to replace another licensed employee during that employee's leave of absence is employed only for the duration of the grant or leave. Such a licensed employee and licensed employees who are employed on temporary contracts for 90 school days or less, or its equivalent in a school district operating under an alternative schedule authorized pursuant to NRS 388.090, to replace licensed employees whose employment has terminated after the beginning of the school year are entitled to credit for that time in fulfilling any period of probation and during that time the provisions of NRS 391.311 to 391.3197, inclusive, *and section 1 of this act* for demotion, suspension or dismissal apply to them.
 - **Sec. 4.** NRS 391.3116 is hereby amended to read as follows:
- 391.3116 [The] Excluding the provisions of section 1 of this act, the provisions of NRS 391.311 to 391.3197, inclusive, fand section 1 of this act]

do not apply to a teacher, administrator, or other licensed employee who has entered into a contract with the board negotiated pursuant to chapter 288 of NRS if the contract contains separate provisions relating to the board's right to dismiss or refuse to reemploy the employee or demote an administrator.

- **Sec. 5.** NRS 391.317 is hereby amended to read as follows:
- 391.317 1. At least 15 days before recommending to a board that it demote, dismiss or not reemploy a postprobationary employee, or dismiss or demote a probationary employee, the superintendent shall give written notice to the employee, by registered or certified mail, of the superintendent's intention to make the recommendation.
 - 2. The notice must:
 - (a) Inform the licensed employee of the grounds for the recommendation.
- (b) Inform the employee that, if a written request therefor is directed to the superintendent within 10 days after receipt of the notice, the employee is entitled to a hearing before a hearing officer [.] pursuant to NRS 391.315 to 391.3194, inclusive, or if the employee is deemed to be a probationary employee pursuant to section 1 of this act and dismissal of the employee will occur before the completion of the current school year, the employee may request an expedited hearing pursuant to subsection 3.
 - (c) Refer to chapter 391 of NRS.
- 3. If an employee who is deemed to be a probationary employee pursuant to section 1 of this act receives notice pursuant to subsection 1 that he or she will be dismissed before the completion of the current school year, the employee may request an expedited hearing pursuant to the Expedited Labor Arbitration Procedures established by the American Arbitration Association or its successor organization. If the employee elects to proceed under the expedited procedures, the provisions of NRS 391.3161, 391.3192 and 391.3193 do not apply.
 - **Sec. 6.** This act becomes effective on July 1, 2011.

Assemblyman Bobzien moved that the Assembly concur in the Senate Amendment No. 860 to Assembly Bill No. 225.

Remarks by Assemblyman Bobzien.

Motion carried by a constitutional majority.

Bill ordered enrolled.

Assembly Bill No. 229.

The following Senate amendment was read:

Amendment No. 861.

AN ACT relating to education; revising the annual reports of accountability information for public schools; requiring the board of trustees of each school district to establish and implement a program of performance pay and enhanced compensation for the recruitment and retention of licensed teachers

and administrators; removing probationary teachers and probationary administrators from the applicability of certain provisions governing certain disciplinary measures by school districts; revising provisions governing the demotion, suspension, dismissal and nonreemployment of certain teachers and administrators; expanding the grounds for immediate dismissal and refusal to reemploy; revising the designations of the overall performance of teachers and administrators required by the policies for evaluations of each school district; authorizing a **probationary or** postprobationary employee to request an expedited hearing under certain circumstances; revising provisions governing the probationary periods of teachers and administrators and the evaluations of probationary teachers and probationary administrators; **revising provisions governing the reduction in the workforce of a school district;** and providing other matters properly relating thereto.

Legislative Counsel's Digest:

Existing law requires the State Board of Education and the board of trustees of each school district to prepare an annual report of accountability information for public schools. (NRS 385.3469, 385.347) **Sections 1 and 2** of this bill expand the requirements of the annual reports of accountability to include a reporting of the number and percentages of administrators, teachers and **[support]** other staff for each elementary school, middle school or junior high school, and high school and for each school district in the State.

Section 8 of this bill requires the board of trustees of each school district to: (1) establish a program of performance pay and enhanced compensation for the recruitment and retention of licensed teachers and administrators; and (2) implement the program commencing with the 2014-2015 school year.

Existing law requires that the board of trustees of each school district develop a policy for the evaluation of teachers and administrators pursuant to which an individual teacher or administrator is designated as "satisfactory" or "unsatisfactory." (NRS 391.3125, 391.3127) Effective July 1, 2013, sections 14 and 16 of this bill revise the policies for evaluations to require the designation of an individual teacher or administrator as "highly effective," "effective," "minimally effective" or "ineffective [:-] " and provide that the policies must require that certain information on pupil achievement which is maintained by the automated system of accountability information for Nevada account for at least 50 percent of the evaluations.

Section 9 of this bill provides that if a written evaluation of a probationary teacher or probationary administrator states that the overall performance of the teacher or administrator has been designated as "unsatisfactory": (1) the evaluation must include a written statement that the contract of the person so evaluated may not be renewed for the next school year and that the employee may request reasonable assistance in correcting the deficiencies identified in

the evaluation; and (2) the person must acknowledge in writing that he or she has received and understands the written statement. **Section 20** of this bill amends **section 9**, effective July 1, 2013, when the four types of designations for the evaluations of teachers and administrators will take effect.

Existing law sets forth certain rights and responsibilities relating to disciplinary measures taken by school districts with respect to probationary and postprobationary teachers and administrators. (NRS 391.311-391.3197) **Section 11** of this bill removes probationary teachers and new employees hired as probationary administrators from the applicability of the provisions governing admonition, demotion, suspension, dismissal and nonreemployment.

Section 12 of this bill revises the grounds for which a teacher may be suspended, dismissed or not reemployed or for which an administrator may be demoted, suspended, dismissed or not reemployed to include gross misconduct.

Section 13 of this bill provides that a postprobationary teacher who receives an unsatisfactory evaluation must be evaluated three times in the immediately succeeding school year. Effective July 1, 2013, section 14 of this bill provides that a postprobationary teacher who receives an evaluation of "minimally effective" or "ineffective" must be evaluated three times in the immediately succeeding school year.

Section 17 of this bill expands the grounds for which a licensed employee is subject to immediate dismissal or a refusal to reemploy without first receiving a written admonition to include gross misconduct.

Section 18 of this bill authorizes a postprobationary employee who receives notice that he or she will be dismissed before completion of the current school year to request an expedited hearing pursuant to the expedited procedures established by the American Arbitration Association.

Under existing law, a probationary teacher and a probationary administrator serve two 1-year periods as a probationary employee. If the employee receives satisfactory evaluations in the first probationary year, the second probationary year must be waived and the person is entitled to postprobationary employment with the school district. (NRS 391.3197) Section 19 of this bill revises the probationary period from two 1-year periods to three 1-year periods, without a waiver of any of the probationary years. Section 19 also provides that a probationary employee who receives notice that he or she will be dismissed before the completion of the current school year may request an expedited hearing pursuant to the procedures established by the American Arbitration Association or its successor organization.

Section 19.6 provides that a board of trustees of a school district that determines a reduction in the existing workforce of the licensed

educational personnel in the school district is necessary must not base the decision to lay off a teacher or an administrator solely on the seniority of the teacher or administrator and may consider certain other factors.

Section 1 of Assembly Bill No. 225 of this session provides that if a postprobationary employee receives an unsatisfactory evaluation on an evaluation conducted pursuant to NRS 391.3125 or 391.3127, as applicable, or any other equivalent evaluation system which designates his or her overall performance as below average for 2 consecutive school vears, the employee shall be deemed a probationary employee and serve an additional probationary period. Section 20.5 of this bill amends section 1 of Assembly Bill No. 225 to provide that a postprobationary employee must serve an additional probationary period if he or she receives an evaluation for 2 consecutive school years as: (1) minimally effective; (2) ineffective; (3) minimally effective during 1 year of the 2vear consecutive period and ineffective during the other year of the period; or (4) if evaluated pursuant to any other system of evaluation, any designation which indicates that the overall performance of the employee is below average. Section 23 of this bill provides that section 20.5 becomes effective if, and only if, Assembly Bill No. 225 of this session is enacted by the Legislature and becomes effective.

THE PEOPLE OF THE STATE OF NEVADA, REPRESENTED IN SENATE AND ASSEMBLY, DO ENACT AS FOLLOWS:

Section 1. NRS 385.3469 is hereby amended to read as follows:

385.3469 1. The State Board shall prepare an annual report of accountability that includes, without limitation:

- (a) Information on the achievement of all pupils based upon the results of the examinations administered pursuant to NRS 389.015 and 389.550, reported for each school district, including, without limitation, each charter school in the district, and for this State as a whole.
- (b) Except as otherwise provided in subsection 2, pupil achievement, reported separately by gender and reported separately for the following groups of pupils:
- (1) Pupils who are economically disadvantaged, as defined by the State Board:
- (2) Pupils from major racial and ethnic groups, as defined by the State Board;
 - (3) Pupils with disabilities;
 - (4) Pupils who are limited English proficient; and
 - (5) Pupils who are migratory children, as defined by the State Board.

- (c) A comparison of the achievement of pupils in each group identified in paragraph (b) of subsection 1 of NRS 385.361 with the annual measurable objectives of the State Board.
- (d) The percentage of all pupils who were not tested, reported for each school district, including, without limitation, each charter school in the district, and for this State as a whole.
- (e) Except as otherwise provided in subsection 2, the percentage of pupils who were not tested, reported separately by gender and reported separately for the groups identified in paragraph (b).
- (f) The most recent 3-year trend in the achievement of pupils in each subject area tested and each grade level tested pursuant to NRS 389.015 and 389.550, reported for each school district, including, without limitation, each charter school in the district, and for this State as a whole, which may include information regarding the trend in the achievement of pupils for more than 3 years, if such information is available.
- (g) Information on whether each school district has made adequate yearly progress, including, without limitation, the name of each school district, if any, designated as demonstrating need for improvement pursuant to NRS 385.377 and the number of consecutive years that the school district has carried that designation.
- (h) Information on whether each public school, including, without limitation, each charter school, has made:
- (1) Adequate yearly progress, including, without limitation, the name of each public school, if any, designated as demonstrating need for improvement pursuant to NRS 385.3623 and the number of consecutive years that the school has carried that designation.
- (2) Progress based upon the model adopted by the Department pursuant to NRS 385.3595, if applicable for the grade level of pupils enrolled at the school.
- (i) Information on the results of pupils who participated in the examinations of the National Assessment of Educational Progress required pursuant to NRS 389.012.
- (j) The ratio of pupils to teachers in kindergarten and at each grade level for all elementary schools, reported for each school district, including, without limitation, each charter school in the district, and for this State as a whole, and the average class size for each core academic subject, as set forth in NRS 389.018, for each secondary school, reported for each school district and for this State as a whole.
- (k) The total number of persons employed by each school district in this State, including without limitation, each charter school in the district. Each such person must be reported as either an administrator, a teacher or [support] other staff [...] and must not be reported in more than one

<u>category</u>. In addition to the total number of persons employed by each school district in each category, the report must include the number of employees in each of the three categories expressed as a percentage of the total number of persons employed by the school district. As used in this paragraph:

- (1) "Administrator" means a person [employed primarily to supervise support] who spends at least 50 percent of his or her work year supervising <u>other_staff</u> or licensed personnel, or both, [in_addition_to_providing administrative services.] and who is not classified by the board of trustees of a school district as a professional-technical employee.
- (2) <u>["Support]</u> <u>"Other staff"</u> means all persons who are not reported as administrators or teachers [...], including, without limitation:
- (I) School counselors, school nurses and other employees who spend at least 50 percent of their work year providing emotional support, noninstructional guidance or medical support to pupils;
- (II) Noninstructional support staff, including, without limitation, janitors, school police officers and maintenance staff; and
- (III) Persons classified by the board of trustees of a school district as professional-technical employees, including, without limitation, technical employees and employees on the professional-technical pay scale.
- (3) "Teacher" means a person licensed pursuant to chapter 391 of NRS [primarily to provide] who is classified by the board of trustees of a school district:
- (I) As a teacher and who spends at least 50 percent of his or her work year providing instruction or discipline to pupils \(\frac{1-1}{2}\); or
- (II) As instructional support staff, who does not hold a supervisory position and who spends not more than 50 percent of his or her work year providing instruction to pupils. Such instructional support staff includes, without limitation, librarians and persons who provide instructional support.
- (1) For each school district, including, without limitation, each charter school in the district, and for this State as a whole, information on the professional qualifications of teachers employed by the school districts and charter schools, including, without limitation:
 - (1) The percentage of teachers who are:
 - (I) Providing instruction pursuant to NRS 391.125;
- (II) Providing instruction pursuant to a waiver of the requirements for licensure for the grade level or subject area in which the teachers are employed; or
- (III) Otherwise providing instruction without an endorsement for the subject area in which the teachers are employed;

- (2) The percentage of classes in the core academic subjects, as set forth in NRS 389.018, in this State that are not taught by highly qualified teachers;
- (3) The percentage of classes in the core academic subjects, as set forth in NRS 389.018, in this State that are not taught by highly qualified teachers, in the aggregate and disaggregated by high-poverty compared to low-poverty schools, which for the purposes of this subparagraph means schools in the top quartile of poverty and the bottom quartile of poverty in this State;
 - (4) For each middle school, junior high school and high school:
- (I) On and after July 1, 2005, the number of persons employed as substitute teachers for 20 consecutive days or more in the same classroom or assignment, designated as long-term substitute teachers, including the total number of days long-term substitute teachers were employed at each school, identified by grade level and subject area; and
- (II) On and after July 1, 2006, the number of persons employed as substitute teachers for less than 20 consecutive days, designated as short-term substitute teachers, including the total number of days short-term substitute teachers were employed at each school, identified by grade level and subject area; and
 - (5) For each elementary school:
- (I) On and after July 1, 2005, the number of persons employed as substitute teachers for 20 consecutive days or more in the same classroom or assignment, designated as long-term substitute teachers, including the total number of days long-term substitute teachers were employed at each school, identified by grade level; and
- (II) On and after July 1, 2006, the number of persons employed as substitute teachers for less than 20 consecutive days, designated as short-term substitute teachers, including the total number of days short-term substitute teachers were employed at each school, identified by grade level.
- [(1)] (m) The total expenditure per pupil for each school district in this State, including, without limitation, each charter school in the district. If this State has a financial analysis program that is designed to track educational expenditures and revenues to individual schools, the State Board shall use that statewide program in complying with this paragraph. If a statewide program is not available, the State Board shall use the Department's own financial analysis program in complying with this paragraph.
- [(m)] (n) The total statewide expenditure per pupil. If this State has a financial analysis program that is designed to track educational expenditures and revenues to individual schools, the State Board shall use that statewide program in complying with this paragraph. If a statewide program is not available, the State Board shall use the Department's own financial analysis program in complying with this paragraph.

- [(n)] (o) For all elementary schools, junior high schools and middle schools, the rate of attendance, reported for each school district, including, without limitation, each charter school in the district, and for this State as a whole.
- [(o)] (p) The annual rate of pupils who drop out of school in grade 8 and a separate reporting of the annual rate of pupils who drop out of school in grades 9 to 12, inclusive, reported for each school district, including, without limitation, each charter school in the district, and for this State as a whole. The reporting for pupils in grades 9 to 12, inclusive, excludes pupils who:
- (1) Provide proof to the school district of successful completion of the examinations of general educational development.
- (2) Are enrolled in courses that are approved by the Department as meeting the requirements for an adult standard diploma.
 - (3) Withdraw from school to attend another school.
- $\{(p)\}\$ (q) The attendance of teachers who provide instruction, reported for each school district, including, without limitation, each charter school in the district, and for this State as a whole.
- $\{(q)\}$ (r) Incidents involving weapons or violence, reported for each school district, including, without limitation, each charter school in the district, and for this State as a whole.
- $\{(r)\}$ (s) Incidents involving the use or possession of alcoholic beverages or controlled substances, reported for each school district, including, without limitation, each charter school in the district, and for this State as a whole.
- [(s)] (t) The suspension and expulsion of pupils required or authorized pursuant to NRS 392.466 and 392.467, reported for each school district, including, without limitation, each charter school in the district, and for this State as a whole.
- [(t)] (u) The number of pupils who are deemed habitual disciplinary problems pursuant to NRS 392.4655, reported for each school district, including, without limitation, each charter school in the district, and for this State as a whole.
- [(u)] (v) The number of pupils in each grade who are retained in the same grade pursuant to NRS 392.033 or 392.125, reported for each school district, including, without limitation, each charter school in the district, and for this State as a whole.
- [(v)] (w) The transiency rate of pupils, reported for each school district, including, without limitation, each charter school in the district, and for this State as a whole. For the purposes of this paragraph, a pupil is not a transient if the pupil is transferred to a different school within the school district as a result of a change in the zone of attendance by the board of trustees of the school district pursuant to NRS 388.040.

- $\frac{\{(w)\}}{\{(x)\}}$ Each source of funding for this State to be used for the system of public education.
- $\{(x)\}$ (y) A compilation of the programs of remedial study purchased in whole or in part with money received from this State that are used in each school district, including, without limitation, each charter school in the district. The compilation must include:
- (1) The amount and sources of money received for programs of remedial study.
- (2) An identification of each program of remedial study, listed by subject area.
- [(y)] (z) The percentage of pupils who graduated from a high school or charter school in the immediately preceding year and enrolled in remedial courses in reading, writing or mathematics at a university, state college or community college within the Nevada System of Higher Education, reported for each school district, including, without limitation, each charter school in the district, and for this State as a whole.
- [(z)] (aa) The technological facilities and equipment available for educational purposes, reported for each school district, including, without limitation, each charter school in the district, and for this State as a whole.
- [(aa)] (bb) For each school district, including, without limitation, each charter school in the district, and for this State as a whole, the number and percentage of pupils who received:
- (1) A standard high school diploma, reported separately for pupils who received the diploma pursuant to:
 - (I) Paragraph (a) of subsection 1 of NRS 389.805; and
 - (II) Paragraph (b) of subsection 1 of NRS 389.805.
 - (2) An adjusted diploma.
 - (3) A certificate of attendance.
- [(bb)] (cc) For each school district, including, without limitation, each charter school in the district, and for this State as a whole, the number and percentage of pupils who failed to pass the high school proficiency examination.
- [(ee)] (dd) The number of habitual truants who are reported to a school police officer or local law enforcement agency pursuant to paragraph (a) of subsection 2 of NRS 392.144 and the number of habitual truants who are referred to an advisory board to review school attendance pursuant to paragraph (b) of subsection 2 of NRS 392.144, reported for each school district, including, without limitation, each charter school in the district, and for this State as a whole.
- [(dd)] (ee) Information on the paraprofessionals employed at public schools in this State, including, without limitation, the charter schools in this State. The information must include:

- (1) The number of paraprofessionals employed, reported for each school district, including, without limitation, each charter school in the district, and for this State as a whole; and
- (2) For each school district, including, without limitation, each charter school in the district, and for this State as a whole, the number and percentage of all paraprofessionals who do not satisfy the qualifications set forth in 20 U.S.C. § 6319(c). The reporting requirements of this subparagraph apply to paraprofessionals who are employed in programs supported with Title I money and to paraprofessionals who are not employed in programs supported with Title I money.
- [(ee)] (ff) An identification of appropriations made by the Legislature to improve the academic achievement of pupils and programs approved by the Legislature to improve the academic achievement of pupils.
- [(ff)] (gg) A compilation of the special programs available for pupils at individual schools, listed by school and by school district, including, without limitation, each charter school in the district.
- [(gg)] (hh) For each school district, including, without limitation, each charter school in the district and for this State as a whole, information on pupils enrolled in career and technical education, including, without limitation:
- (1) The number of pupils enrolled in a course of career and technical education;
- (2) The number of pupils who completed a course of career and technical education;
- (3) The average daily attendance of pupils who are enrolled in a program of career and technical education;
- (4) The annual rate of pupils who dropped out of school and were enrolled in a program of career and technical education before dropping out;
- (5) The number and percentage of pupils who completed a program of career and technical education and who received a standard high school diploma, an adjusted diploma or a certificate of attendance; and
- (6) The number and percentage of pupils who completed a program of career and technical education and who did not receive a high school diploma because the pupils failed to pass the high school proficiency examination.
- 2. A separate reporting for a group of pupils must not be made pursuant to this section if the number of pupils in that group is insufficient to yield statistically reliable information or the results would reveal personally identifiable information about an individual pupil. The State Board shall prescribe a mechanism for determining the minimum number of pupils that must be in a group for that group to yield statistically reliable information.
 - 3. The annual report of accountability must:

- (a) Comply with 20 U.S.C. § 6311(h)(1) and the regulations adopted pursuant thereto;
 - (b) Be prepared in a concise manner; and
- (c) Be presented in an understandable and uniform format and, to the extent practicable, provided in a language that parents can understand.
 - 4. On or before September 1 of each year, the State Board shall:
- (a) Provide for public dissemination of the annual report of accountability by posting a copy of the report on the Internet website maintained by the Department; and
- (b) Provide written notice that the report is available on the Internet website maintained by the Department. The written notice must be provided to the:
 - (1) Governor;
 - (2) Committee;
 - (3) Bureau;
 - (4) Board of Regents of the University of Nevada;
 - (5) Board of trustees of each school district; and
 - (6) Governing body of each charter school.
- 5. Upon the request of the Governor, an entity described in paragraph (b) of subsection 4 or a member of the general public, the State Board shall provide a portion or portions of the annual report of accountability.
 - 6. As used in this section:
- (a) "Highly qualified" has the meaning ascribed to it in 20 U.S.C. \S 7801(23).
 - (b) "Paraprofessional" has the meaning ascribed to it in NRS 391.008.
 - **Sec. 2.** NRS 385.347 is hereby amended to read as follows:
- 385.347 1. The board of trustees of each school district in this State, in cooperation with associations recognized by the State Board as representing licensed educational personnel in the district, shall adopt a program providing for the accountability of the school district to the residents of the district and to the State Board for the quality of the schools and the educational achievement of the pupils in the district, including, without limitation, pupils enrolled in charter schools in the school district. The board of trustees of each school district shall report the information required by subsection 2 for each charter school that is located within the school district, regardless of the sponsor of the charter school. The information for charter schools must be reported separately and must denote the charter schools sponsored by the school district, the charter schools sponsored by the State Board and the charter schools sponsored by a college or university within the Nevada System of Higher Education.
- 2. The board of trustees of each school district shall, on or before August 15 of each year, prepare an annual report of accountability concerning:

- (a) The educational goals and objectives of the school district.
- (b) Pupil achievement for each school in the district and the district as a whole, including, without limitation, each charter school in the district. The board of trustees of the district shall base its report on the results of the examinations administered pursuant to NRS 389.015 and 389.550 and shall compare the results of those examinations for the current school year with those of previous school years. The report must include, for each school in the district, including, without limitation, each charter school in the district, and each grade in which the examinations were administered:
 - (1) The number of pupils who took the examinations.
- (2) A record of attendance for the period in which the examinations were administered, including an explanation of any difference in the number of pupils who took the examinations and the number of pupils who are enrolled in the school.
- (3) Except as otherwise provided in this paragraph, pupil achievement, reported separately by gender and reported separately for the following groups of pupils:
- (I) Pupils who are economically disadvantaged, as defined by the State Board;
- (II) Pupils from major racial and ethnic groups, as defined by the State Board:
 - (III) Pupils with disabilities;
 - (IV) Pupils who are limited English proficient; and
 - (V) Pupils who are migratory children, as defined by the State Board.
- (4) A comparison of the achievement of pupils in each group identified in paragraph (b) of subsection 1 of NRS 385.361 with the annual measurable objectives of the State Board.
 - (5) The percentage of pupils who were not tested.
- (6) Except as otherwise provided in this paragraph, the percentage of pupils who were not tested, reported separately by gender and reported separately for the groups identified in subparagraph (3).
- (7) The most recent 3-year trend in pupil achievement in each subject area tested and each grade level tested pursuant to NRS 389.015 and 389.550, which may include information regarding the trend in the achievement of pupils for more than 3 years, if such information is available.
- (8) Information that compares the results of pupils in the school district, including, without limitation, pupils enrolled in charter schools in the district, with the results of pupils throughout this State. The information required by this subparagraph must be provided in consultation with the Department to ensure the accuracy of the comparison.
- (9) For each school in the district, including, without limitation, each charter school in the district, information that compares the results of pupils

in the school with the results of pupils throughout the school district and throughout this State. The information required by this subparagraph must be provided in consultation with the Department to ensure the accuracy of the comparison.

- (10) Information on whether each school in the district, including, without limitation, each charter school in the district, has made progress based upon the model adopted by the Department pursuant to NRS 385.3595.
- → A separate reporting for a group of pupils must not be made pursuant to this paragraph if the number of pupils in that group is insufficient to yield statistically reliable information or the results would reveal personally identifiable information about an individual pupil. The State Board shall prescribe the mechanism for determining the minimum number of pupils that must be in a group for that group to yield statistically reliable information.
- (c) The ratio of pupils to teachers in kindergarten and at each grade level for each elementary school in the district and the district as a whole, including, without limitation, each charter school in the district, and the average class size for each core academic subject, as set forth in NRS 389.018, for each secondary school in the district and the district as a whole, including, without limitation, each charter school in the district.
- (d) The total number of persons employed for each elementary school, middle school or junior high school, and high school in the district, including, without limitation, each charter school in the district. Each such person must be reported as either an administrator, a teacher or [support] other staff [+] and must not be reported in more than one category. In addition to the total number of persons employed by each school in each category, the report must include the number of employees in each of the three categories for each school expressed as a percentage of the total number of persons employed by the school. As used in this paragraph:
- (1) "Administrator" means a person [employed primarily to supervise support] who spends at least 50 percent of his or her work year supervising other staff or licensed personnel, or both, [in addition to providing administrative services.] and who is not classified by the board of trustees of the school district as a professional-technical employee.
- (2) ["Support] "Other staff" means all persons who are not reported as administrators or teachers [...], including, without limitation:
- (I) School counselors, school nurses and other employees who spend at least 50 percent of their work year providing emotional support, noninstructional guidance or medical support to pupils;
- (II) Noninstructional support staff, including, without limitation, janitors, school police officers and maintenance staff; and

- (III) Persons classified by the board of trustees of the school district as professional-technical employees, including, without limitation, technical employees and employees on the professional-technical pay scale.
- (3) "Teacher" means a person licensed pursuant to chapter 391 of NRS who is [employed primarily to provide] classified by the board of trustees of the school district:
- (I) As a teacher and who spends at least 50 percent of his or her work year providing instruction or discipline to pupils [--]; or
- (II) As instructional support staff, who does not hold a supervisory position and who spends not more than 50 percent of his or her work year providing instruction to pupils. Such instructional support staff includes, without limitation, librarians and persons who provide instructional support.
- (e) The total number of persons employed by the school district, including without limitation, each charter school in the district. Each such person must be reported as either an administrator, a teacher or [support] other staff [++] and must not be reported in more than one category. In addition to the total number of persons employed by the school district in each category, the report must include the number of employees in each of the three categories expressed as a percentage of the total number of persons employed by the school district. As used in this paragraph [++]
- (1)="Administrator" means a person employed primarily to supervise support staff or licensed personnel, or both, in addition to providing administrative services.
- (2)—"Support staff" means all persons who are not reported as
- (3)-"Teacher" means a person licensed pursuant to chapter 391 of NRS who is employed primarily to provide instruction to pupils.], "administrator," "other staff" and "teacher" have the meanings ascribed to them in paragraph (d).
- (f) Information on the professional qualifications of teachers employed by each school in the district and the district as a whole, including, without limitation, each charter school in the district. The information must include, without limitation:
 - (1) The percentage of teachers who are:
 - (I) Providing instruction pursuant to NRS 391.125;
- (II) Providing instruction pursuant to a waiver of the requirements for licensure for the grade level or subject area in which the teachers are employed; or
- (III) Otherwise providing instruction without an endorsement for the subject area in which the teachers are employed;

- (2) The percentage of classes in the core academic subjects, as set forth in NRS 389.018, that are not taught by highly qualified teachers;
- (3) The percentage of classes in the core academic subjects, as set forth in NRS 389.018, that are not taught by highly qualified teachers, in the aggregate and disaggregated by high-poverty compared to low-poverty schools, which for the purposes of this subparagraph means schools in the top quartile of poverty and the bottom quartile of poverty in this State;
 - (4) For each middle school, junior high school and high school:
- (I) On and after July 1, 2005, the number of persons employed as substitute teachers for 20 consecutive days or more in the same classroom or assignment, designated as long-term substitute teachers, including the total number of days long-term substitute teachers were employed at each school, identified by grade level and subject area; and
- (II) On and after July 1, 2006, the number of persons employed as substitute teachers for less than 20 consecutive days, designated as short-term substitute teachers, including the total number of days short-term substitute teachers were employed at each school, identified by grade level and subject area; and
 - (5) For each elementary school:
- (I) On and after July 1, 2005, the number of persons employed as substitute teachers for 20 consecutive days or more in the same classroom or assignment, designated as long-term substitute teachers, including the total number of days long-term substitute teachers were employed at each school, identified by grade level; and
- (II) On and after July 1, 2006, the number of persons employed as substitute teachers for less than 20 consecutive days, designated as short-term substitute teachers, including the total number of days short-term substitute teachers were employed at each school, identified by grade level.
- [(e)] (g) The total expenditure per pupil for each school in the district and the district as a whole, including, without limitation, each charter school in the district. If this State has a financial analysis program that is designed to track educational expenditures and revenues to individual schools, each school district shall use that statewide program in complying with this paragraph. If a statewide program is not available, each school district shall use its own financial analysis program in complying with this paragraph.
 - [(f)] (h) The curriculum used by the school district, including:
 - (1) Any special programs for pupils at an individual school; and
 - (2) The curriculum used by each charter school in the district.
- $\{(g)\}$ (i) Records of the attendance and truancy of pupils in all grades, including, without limitation:

- (1) The average daily attendance of pupils, for each school in the district and the district as a whole, including, without limitation, each charter school in the district.
- (2) For each elementary school, middle school and junior high school in the district, including, without limitation, each charter school in the district that provides instruction to pupils enrolled in a grade level other than high school, information that compares the attendance of the pupils enrolled in the school with the attendance of pupils throughout the district and throughout this State. The information required by this subparagraph must be provided in consultation with the Department to ensure the accuracy of the comparison.
- [(h)] (j) The annual rate of pupils who drop out of school in grade 8 and a separate reporting of the annual rate of pupils who drop out of school in grades 9 to 12, inclusive, for each such grade, for each school in the district and for the district as a whole. The reporting for pupils in grades 9 to 12, inclusive, excludes pupils who:
- (1) Provide proof to the school district of successful completion of the examinations of general educational development.
- (2) Are enrolled in courses that are approved by the Department as meeting the requirements for an adult standard diploma.
 - (3) Withdraw from school to attend another school.
- $\{(i)\}$ (k) Records of attendance of teachers who provide instruction, for each school in the district and the district as a whole, including, without limitation, each charter school in the district.
- [(j)] (1) Efforts made by the school district and by each school in the district, including, without limitation, each charter school in the district, to increase:
 - (1) Communication with the parents of pupils in the district; and
- (2) The participation of parents in the educational process and activities relating to the school district and each school, including, without limitation, the existence of parent organizations and school advisory committees.
- [(k)] (m) Records of incidents involving weapons or violence for each school in the district, including, without limitation, each charter school in the district.
- [(1)] (n) Records of incidents involving the use or possession of alcoholic beverages or controlled substances for each school in the district, including, without limitation, each charter school in the district.
- [(m)] (o) Records of the suspension and expulsion of pupils required or authorized pursuant to NRS 392.466 and 392.467.
- [(n)] (p) The number of pupils who are deemed habitual disciplinary problems pursuant to NRS 392.4655, for each school in the district and the district as a whole, including, without limitation, each charter school in the district.

- [(o)] (q) The number of pupils in each grade who are retained in the same grade pursuant to NRS 392.033 or 392.125, for each school in the district and the district as a whole, including, without limitation, each charter school in the district.
- [(p)] (r) The transiency rate of pupils for each school in the district and the district as a whole, including, without limitation, each charter school in the district. For the purposes of this paragraph, a pupil is not transient if the pupil is transferred to a different school within the school district as a result of a change in the zone of attendance by the board of trustees of the school district pursuant to NRS 388.040.
 - $\frac{(a)}{(a)}$ (s) Each source of funding for the school district.
- [(r)] (t) A compilation of the programs of remedial study that are purchased in whole or in part with money received from this State, for each school in the district and the district as a whole, including, without limitation, each charter school sponsored by the district. The compilation must include:
- (1) The amount and sources of money received for programs of remedial study for each school in the district and the district as a whole, including, without limitation, each charter school in the district.
- (2) An identification of each program of remedial study, listed by subject area.
- [(s)] (u) For each high school in the district, including, without limitation, each charter school in the district, the percentage of pupils who graduated from that high school or charter school in the immediately preceding year and enrolled in remedial courses in reading, writing or mathematics at a university, state college or community college within the Nevada System of Higher Education.
- $\{(t)\}$ (ν) The technological facilities and equipment available at each school, including, without limitation, each charter school, and the district's plan to incorporate educational technology at each school.
- $\frac{\{(u)\}}{\{(w)\}}$ (w) For each school in the district and the district as a whole, including, without limitation, each charter school in the district, the number and percentage of pupils who received:
- (1) A standard high school diploma, reported separately for pupils who received the diploma pursuant to:
 - (I) Paragraph (a) of subsection 1 of NRS 389.805; and
 - (II) Paragraph (b) of subsection 1 of NRS 389.805.
 - (2) An adjusted diploma.
 - (3) A certificate of attendance.
- $\{(v)\}$ (x) For each school in the district and the district as a whole, including, without limitation, each charter school in the district, the number and percentage of pupils who failed to pass the high school proficiency examination.

- [(w)] (y) The number of habitual truants who are reported to a school police officer or law enforcement agency pursuant to paragraph (a) of subsection 2 of NRS 392.144 and the number of habitual truants who are referred to an advisory board to review school attendance pursuant to paragraph (b) of subsection 2 of NRS 392.144, for each school in the district and for the district as a whole.
- $\{(x)\}$ (z) The amount and sources of money received for the training and professional development of teachers and other educational personnel for each school in the district and for the district as a whole, including, without limitation, each charter school in the district.
- [(y)] (aa) Whether the school district has made adequate yearly progress. If the school district has been designated as demonstrating need for improvement pursuant to NRS 385.377, the report must include a statement indicating the number of consecutive years the school district has carried that designation.
- [(z)] (bb) Information on whether each public school in the district, including, without limitation, each charter school in the district, has made adequate yearly progress, including, without limitation:
- (1) The number and percentage of schools in the district, if any, that have been designated as needing improvement pursuant to NRS 385.3623; and
- (2) The name of each school, if any, in the district that has been designated as needing improvement pursuant to NRS 385.3623 and the number of consecutive years that the school has carried that designation.
- [(aa)] (cc) Information on the paraprofessionals employed by each public school in the district, including, without limitation, each charter school in the district. The information must include:
 - (1) The number of paraprofessionals employed at the school; and
- (2) The number and percentage of all paraprofessionals who do not satisfy the qualifications set forth in 20 U.S.C. § 6319(c). The reporting requirements of this subparagraph apply to paraprofessionals who are employed in positions supported with Title I money and to paraprofessionals who are not employed in positions supported with Title I money.
- [(bb)] (dd) For each high school in the district, including, without limitation, each charter school that operates as a high school, information that provides a comparison of the rate of graduation of pupils enrolled in the high school with the rate of graduation of pupils throughout the district and throughout this State. The information required by this paragraph must be provided in consultation with the Department to ensure the accuracy of the comparison.
- [(ce)] (ee) An identification of the appropriations made by the Legislature that are available to the school district or the schools within the district and

programs approved by the Legislature to improve the academic achievement of pupils.

- [(dd)] (ff) For each school in the district and the district as a whole, including, without limitation, each charter school in the district, information on pupils enrolled in career and technical education, including, without limitation:
- (1) The number of pupils enrolled in a course of career and technical education:
- (2) The number of pupils who completed a course of career and technical education;
- (3) The average daily attendance of pupils who are enrolled in a program of career and technical education;
- (4) The annual rate of pupils who dropped out of school and were enrolled in a program of career and technical education before dropping out;
- (5) The number and percentage of pupils who completed a program of career and technical education and who received a standard high school diploma, an adjusted diploma or a certificate of attendance; and
- (6) The number and percentage of pupils who completed a program of career and technical education and who did not receive a high school diploma because the pupils failed to pass the high school proficiency examination.

[(ee)] (gg) Such other information as is directed by the Superintendent of Public Instruction.

- 3. The records of attendance maintained by a school for purposes of paragraph $\frac{\{(i)\}}{k}$ of subsection 2 must include the number of teachers who are in attendance at school and the number of teachers who are absent from school. A teacher shall be deemed in attendance if the teacher is excused from being present in the classroom by the school in which the teacher is employed for one of the following reasons:
- (a) Acquisition of knowledge or skills relating to the professional development of the teacher; or
- (b) Assignment of the teacher to perform duties for cocurricular or extracurricular activities of pupils.
- 4. The annual report of accountability prepared pursuant to subsection 2 must:
- (a) Comply with 20 U.S.C. § 6311(h)(2) and the regulations adopted pursuant thereto; and
- (b) Be presented in an understandable and uniform format and, to the extent practicable, provided in a language that parents can understand.
 - 5. The Superintendent of Public Instruction shall:
- (a) Prescribe forms for the reports required pursuant to subsection 2 and provide the forms to the respective school districts.

- (b) Provide statistical information and technical assistance to the school districts to ensure that the reports provide comparable information with respect to each school in each district and among the districts throughout this State.
 - (c) Consult with a representative of the:
 - (1) Nevada State Education Association;
 - (2) Nevada Association of School Boards;
 - (3) Nevada Association of School Administrators;
 - (4) Nevada Parent Teacher Association;
 - (5) Budget Division of the Department of Administration; and
 - (6) Legislative Counsel Bureau,
- → concerning the program and consider any advice or recommendations submitted by the representatives with respect to the program.
- 6. The Superintendent of Public Instruction may consult with representatives of parent groups other than the Nevada Parent Teacher Association concerning the program and consider any advice or recommendations submitted by the representatives with respect to the program.
- 7. On or before August 15 of each year, the board of trustees of each school district shall submit to each advisory board to review school attendance created in the county pursuant to NRS 392.126 the information required in paragraph $\frac{\{(g)\}}{\{i\}}$ (i) of subsection 2.
- 8. On or before August 15 of each year, the board of trustees of each school district shall:
- (a) Provide written notice that the report required pursuant to subsection 2 is available on the Internet website maintained by the school district, if any, or otherwise provide written notice of the availability of the report. The written notice must be provided to the:
 - (1) Governor;
 - (2) State Board;
 - (3) Department;
 - (4) Committee: and
 - (5) Bureau.
- (b) Provide for public dissemination of the annual report of accountability prepared pursuant to subsection 2 in the manner set forth in 20 U.S.C. § 6311(h)(2)(E) by posting a copy of the report on the Internet website maintained by the school district, if any. If a school district does not maintain a website, the district shall otherwise provide for public dissemination of the annual report by providing a copy of the report to the schools in the school district, including, without limitation, each charter school in the district, the residents of the district, and the parents and guardians of pupils enrolled in

schools in the district, including, without limitation, each charter school in the district.

- 9. Upon the request of the Governor, an entity described in paragraph (a) of subsection 8 or a member of the general public, the board of trustees of a school district shall provide a portion or portions of the report required pursuant to subsection 2.
 - 10. As used in this section:
- (a) "Highly qualified" has the meaning ascribed to it in 20 U.S.C. § 7801(23).
 - (b) "Paraprofessional" has the meaning ascribed to it in NRS 391.008.
 - **Sec. 3.** NRS 385.36129 is hereby amended to read as follows:
- 385.36129 1. In addition to the duties prescribed in NRS 385.36127, a support team established for a school shall prepare an annual written report that includes:
- (a) Information concerning the most recent plan to improve the achievement of the school's pupils, the turnaround plan for the school or the plan for restructuring the school, whichever is applicable for the school, including, without limitation, an evaluation of:
 - (1) The appropriateness of the plan for the school; and
- (2) Whether the school has achieved the goals and objectives set forth in the plan;
- (b) The written revisions to the plan to improve the achievement of the school's pupils or written recommendations for revisions to the turnaround plan for the school or the plan for restructuring the school, whichever is applicable for the school, submitted by the support team pursuant to NRS 385.36127:
- (c) A summary of each program for remediation, if any, purchased for the school with money that is available from the Federal Government, this state and the school district in which the school is located, including, without limitation:
 - (1) The name of the program;
- (2) The date on which the program was purchased and the date on which the program was carried out by the school;
- (3) The percentage of personnel at the school who were trained regarding the use of the program;
 - (4) The satisfaction of the personnel at the school with the program; and
- (5) An evaluation of whether the program has improved the academic achievement of the pupils enrolled in the school who participated in the program;
- (d) An analysis of the problems and factors at the school which contributed to the designation of the school as demonstrating need for improvement, including, without limitation, issues relating to:

- (1) The financial resources of the school;
- (2) The administrative and educational personnel of the school;
- (3) The curriculum of the school;
- (4) The facilities available at the school, including the availability and accessibility of educational technology; and
- (5) Any other factors that the support team believes contributed to the designation of the school as demonstrating need for improvement; and
 - (e) Other information concerning the school, including, without limitation:
- (1) The results of the pupils who are enrolled in the school on the examinations that are administered pursuant to NRS 389.550 or the high school proficiency examination, as applicable;
- (2) Records of the attendance and truancy of pupils who are enrolled in the school;
 - (3) The transiency rate of pupils who are enrolled in the school;
- (4) A description of the number of years that each teacher has provided instruction at the school and the rate of turnover of teachers and other educational personnel employed at the school;
- (5) A description of the participation of parents and legal guardians in the educational process and other activities relating to the school;
- (6) A description of each source of money for the remediation of pupils who are enrolled in the school; and
- (7) A description of the disciplinary problems of the pupils who are enrolled in the school, including, without limitation, the information contained in paragraphs $\frac{\{(k) \text{ to } (n), \}}{\{(m) \text{ to } (p), \text{ inclusive, of subsection 2 of NRS 385.347.}$
- 2. On or before November 1, the support team shall submit a copy of the final written report to the:
 - (a) Principal of the school;
 - (b) Board of trustees of the school district in which the school is located;
- (c) Superintendent of schools of the school district in which the school is located;
 - (d) Department; and
 - (e) Bureau.
- → The support team shall make the written report available, upon request, to each parent or legal guardian of a pupil who is enrolled in the school.
 - **Sec. 4.** NRS 385.620 is hereby amended to read as follows:
 - 385.620 The Advisory Council shall:
- 1. Review the policy of parental involvement adopted by the State Board and the policy of parental involvement adopted by the board of trustees of each school district pursuant to NRS 392.457;
- 2. Review the information relating to communication with and participation of parents that is included in the annual report of accountability

for each school district pursuant to paragraph [(i)] (*l*) of subsection 2 of NRS 385.347:

- 3. Review any effective practices carried out in individual school districts to increase parental involvement and determine the feasibility of carrying out those practices on a statewide basis;
- 4. Review any effective practices carried out in other states to increase parental involvement and determine the feasibility of carrying out those practices in this State;
- 5. Identify methods to communicate effectively and provide outreach to parents and legal guardians of pupils who have limited time to become involved in the education of their children for various reasons, including, without limitation, work schedules, single-parent homes and other family obligations;
- 6. Identify the manner in which the level of parental involvement affects the performance, attendance and discipline of pupils;
- 7. Identify methods to communicate effectively with and provide outreach to parents and legal guardians of pupils who are limited English proficient;
- 8. Determine the necessity for the appointment of a statewide parental involvement coordinator or a parental involvement coordinator in each school district, or both;
- 9. On or before July 1 of each year, submit a report to the Legislative Committee on Education describing the activities of the Advisory Council and any recommendations for legislation; and
- 10. On or before February 1 of each odd-numbered year, submit a report to the Director of the Legislative Counsel Bureau for transmission to the next regular session of the Legislature describing the activities of the Advisory Council and any recommendations for legislation.
 - **Sec. 5.** NRS 386.520 is hereby amended to read as follows:
- 386.520 1. A committee to form a charter school must consist of at least three teachers, as defined in subsection 4. In addition to the teachers who serve, the committee may consist of:
 - (a) Members of the general public;
 - (b) Representatives of nonprofit organizations and businesses; or
- (c) Representatives of a college or university within the Nevada System of Higher Education.
- A majority of the persons described in paragraphs (a), (b) and (c) who serve on the committee must be residents of this State at the time that the application to form the charter school is submitted to the Department.
- 2. Before a committee to form a charter school may submit an application to the board of trustees of a school district, the Subcommittee on Charter Schools, the State Board or a college or university within the Nevada

System of Higher Education, it must submit the application to the Department. The application must include all information prescribed by the Department by regulation and:

- (a) A written description of how the charter school will carry out the provisions of NRS 386.500 to 386.610, inclusive.
- (b) A written description of the mission and goals for the charter school. A charter school must have as its stated purpose at least one of the following goals:
 - (1) Improving the opportunities for pupils to learn;
 - (2) Encouraging the use of effective methods of teaching;
- (3) Providing an accurate measurement of the educational achievement of pupils;
 - (4) Establishing accountability of public schools;
- (5) Providing a method for public schools to measure achievement based upon the performance of the schools; or
 - (6) Creating new professional opportunities for teachers.
 - (c) The projected enrollment of pupils in the charter school.
 - (d) The proposed dates of enrollment for the charter school.
- (e) The proposed system of governance for the charter school, including, without limitation, the number of persons who will govern, the method of selecting the persons who will govern and the term of office for each person.
- (f) The method by which disputes will be resolved between the governing body of the charter school and the sponsor of the charter school.
- (g) The proposed curriculum for the charter school and, if applicable to the grade level of pupils who are enrolled in the charter school, the requirements for the pupils to receive a high school diploma, including, without limitation, whether those pupils will satisfy the requirements of the school district in which the charter school is located for receipt of a high school diploma.
 - (h) The textbooks that will be used at the charter school.
- (i) The qualifications of the persons who will provide instruction at the charter school.
- (j) Except as otherwise required by NRS 386.595, the process by which the governing body of the charter school will negotiate employment contracts with the employees of the charter school.
- (k) A financial plan for the operation of the charter school. The plan must include, without limitation, procedures for the audit of the programs and finances of the charter school and guidelines for determining the financial liability if the charter school is unsuccessful.
- (l) A statement of whether the charter school will provide for the transportation of pupils to and from the charter school. If the charter school will provide transportation, the application must include the proposed plan

for the transportation of pupils. If the charter school will not provide transportation, the application must include a statement that the charter school will work with the parents and guardians of pupils enrolled in the charter school to develop a plan for transportation to ensure that pupils have access to transportation to and from the charter school.

- (m) The procedure for the evaluation of teachers of the charter school, if different from the procedure prescribed in NRS 391.3125 [...] and section 9 of this act. If the procedure is different from the procedure prescribed in NRS 391.3125 [...] and section 9 of this act, the procedure for the evaluation of teachers of the charter school must provide the same level of protection and otherwise comply with the standards for evaluation set forth in NRS 391.3125 [...] and section 9 of this act.
- (n) The time by which certain academic or educational results will be achieved.
- (o) The kind of school, as defined in subsections 1 to 4, inclusive, of NRS 388.020, for which the charter school intends to operate.
- (p) A statement of whether the charter school will enroll pupils who are in a particular category of at-risk pupils before enrolling other children who are eligible to attend the charter school pursuant to NRS 386.580 and the method for determining eligibility for enrollment in each such category of at-risk pupils served by the charter school.
- 3. The Department shall review an application to form a charter school to determine whether it is complete. If an application proposes to convert an existing public school, homeschool or other program of home study into a charter school, the Department shall deny the application. The Department shall provide written notice to the applicant of its approval or denial of the application. If the Department denies an application, the Department shall include in the written notice the reason for the denial and the deficiencies in the application. The applicant must be granted 30 days after receipt of the written notice to correct any deficiencies identified in the written notice and resubmit the application.
 - 4. As used in subsection 1, "teacher" means a person who:
- (a) Holds a current license to teach issued pursuant to chapter 391 of NRS; and
 - (b) Has at least 2 years of experience as an employed teacher.
- → The term does not include a person who is employed as a substitute teacher.
 - **Sec. 6.** NRS 388.795 is hereby amended to read as follows:
- 388.795 1. The Commission shall establish a plan for the use of educational technology in the public schools of this State. In preparing the plan, the Commission shall consider:

- (a) Plans that have been adopted by the Department and the school districts in this State:
 - (b) Plans that have been adopted in other states;
- (c) The information reported pursuant to paragraph $\frac{\{(t)\}}{\{(v)\}}$ of subsection 2 of NRS 385.347;
- (d) The results of the assessment of needs conducted pursuant to subsection 6; and
- (e) Any other information that the Commission or the Committee deems relevant to the preparation of the plan.
- 2. The plan established by the Commission must include recommendations for methods to:
- (a) Incorporate educational technology into the public schools of this State:
- (b) Increase the number of pupils in the public schools of this State who have access to educational technology;
- (c) Increase the availability of educational technology to assist licensed teachers and other educational personnel in complying with the requirements of continuing education, including, without limitation, the receipt of credit for college courses completed through the use of educational technology;
- (d) Facilitate the exchange of ideas to improve the achievement of pupils who are enrolled in the public schools of this State; and
- (e) Address the needs of teachers in incorporating the use of educational technology in the classroom, including, without limitation, the completion of training that is sufficient to enable the teachers to instruct pupils in the use of educational technology.
 - 3. The Department shall provide:
 - (a) Administrative support;
 - (b) Equipment; and
 - (c) Office space,
- → as is necessary for the Commission to carry out the provisions of this section.
- 4. The following entities shall cooperate with the Commission in carrying out the provisions of this section:
 - (a) The State Board.
 - (b) The board of trustees of each school district.
 - (c) The superintendent of schools of each school district.
 - (d) The Department.
 - 5. The Commission shall:
- (a) Develop technical standards for educational technology and any electrical or structural appurtenances necessary thereto, including, without limitation, uniform specifications for computer hardware and wiring, to

ensure that such technology is compatible, uniform and can be interconnected throughout the public schools of this State.

- (b) Allocate money to the school districts from the Trust Fund for Educational Technology created pursuant to NRS 388.800 and any money appropriated by the Legislature for educational technology, subject to any priorities for such allocation established by the Legislature.
- (c) Establish criteria for the board of trustees of a school district that receives an allocation of money from the Commission to:
 - (1) Repair, replace and maintain computer systems.
- (2) Upgrade and improve computer hardware and software and other educational technology.
- (3) Provide training, installation and technical support related to the use of educational technology within the district.
- (d) Submit to the Governor, the Committee and the Department its plan for the use of educational technology in the public schools of this State and any recommendations for legislation.
- (e) Review the plan annually and make revisions as it deems necessary or as directed by the Committee or the Department.
- (f) In addition to the recommendations set forth in the plan pursuant to subsection 2, make further recommendations to the Committee and the Department as the Commission deems necessary.
- 6. During the spring semester of each even-numbered school year, the Commission shall conduct an assessment of the needs of each school district relating to educational technology. In conducting the assessment, the Commission shall consider:
 - (a) The recommendations set forth in the plan pursuant to subsection 2;
- (b) The plan for educational technology of each school district, if applicable;
- (c) Evaluations of educational technology conducted for the State or for a school district, if applicable; and
 - (d) Any other information deemed relevant by the Commission.
- → The Commission shall submit a final written report of the assessment to the Superintendent of Public Instruction on or before April 1 of each even-numbered year.
- 7. The Superintendent of Public Instruction shall prepare a written compilation of the results of the assessment conducted by the Commission and transmit the written compilation on or before June 1 of each even-numbered year to the Legislative Committee on Education and to the Director of the Legislative Counsel Bureau for transmission to the next regular session of the Legislature.
- 8. The Commission may appoint an advisory committee composed of members of the Commission or other qualified persons to provide

recommendations to the Commission regarding standards for the establishment, coordination and use of a telecommunications network in the public schools throughout the various school districts in this State. The advisory committee serves at the pleasure of the Commission and without compensation unless an appropriation or other money for that purpose is provided by the Legislature.

- 9. As used in this section, "public school" includes the Caliente Youth Center, the Nevada Youth Training Center and any other state facility for the detention of children that is operated pursuant to title 5 of NRS.
- **Sec. 7.** Chapter 391 of NRS is hereby amended by adding thereto the provisions set forth as sections 8 and 9 of this act.
 - Sec. 8. 1. The board of trustees of each school district shall:
- (a) Establish a program of performance pay and enhanced compensation for the recruitment and retention of licensed teachers and administrators which must be negotiated pursuant to chapter 288 of NRS; and
- (b) Commencing with the 2014-2015 school year, implement the program established pursuant to paragraph (a).
- 2. The program of performance pay and enhanced compensation established by a school district pursuant to subsection 1 must have as its primary focus the improvement in the academic achievement of pupils and must give appropriate consideration to implementation in at-risk schools. In addition, the program may include, without limitation, the following components:
- (a) Career leadership advancement options to maximize the retention of teachers in the classroom and the retention of administrators;
 - (b) Professional development;
 - (c) Group incentives; and
- (d) Multiple assessments of individual teachers and administrators, with primary emphasis on individual pupil improvement and growth in academic achievement, including, without limitation, portfolios of instruction, leadership and professional growth, and other appropriate measures of teacher and administrator performance which must be considered.
- Sec. 9. 1. If a written evaluation of a probationary teacher or probationary administrator designates the overall performance of the teacher or administrator as "unsatisfactory":
- (a) The written evaluation must include the following statement: "Please be advised that, pursuant to Nevada law, your contract may not be renewed for the next school year. If you receive two evaluations for this school year which designate your performance as 'unsatisfactory,' and if you have another evaluation remaining this school year, you may request that the

evaluation be conducted by another administrator. You may also request, to the administrator who conducted the evaluation, reasonable assistance in correcting the deficiencies reported in the evaluation for which you request assistance, and upon such request, a reasonable effort will be made to assist you in correcting those deficiencies."

- (b) The probationary teacher or probationary administrator, as applicable, must acknowledge in writing that he or she has received and understands the statement described in paragraph (a).
- 2. If a probationary teacher or probationary administrator requests that his or her next evaluation be conducted by another administrator in accordance with the notice required by subsection 1, the administrator conducting the evaluation must be:
- (a) Employed by the school district or, if the school district has five or fewer administrators, employed by another school district in this State; and
- (b) Selected by the [superintendent and the] probationary teacher or probationary administrator, as applicable [-], from a list of three candidates submitted by the superintendent.
- 3. If a probationary teacher or probationary administrator requests assistance in correcting deficiencies reported in his or her evaluation, the administrator who conducted the evaluation shall ensure that a reasonable effort is made to assist the probationary teacher or <u>probationary</u> administrator in correcting those deficiencies.
 - **Sec. 10.** NRS 391.311 is hereby amended to read as follows:
- 391.311 As used in NRS 391.311 to 391.3197, inclusive, *and section 9 of this act*, unless the context otherwise requires:
- 1. "Administrator" means any employee who holds a license as an administrator and who is employed in that capacity by a school district.
- 2. "Board" means the board of trustees of the school district in which a licensed employee affected by NRS 391.311 to 391.3197, inclusive, *and section 9 of this act* is employed.
- 3. "Demotion" means demotion of an administrator to a position of lesser rank, responsibility or pay and does not include transfer or reassignment for purposes of an administrative reorganization.
 - 4. "Immorality" means:
- (a) An act forbidden by NRS 200.366, 200.368, 200.400, 200.508, 201.180, 201.190, 201.210, 201.220, 201.230, 201.265, 201.540, 201.560, 207.260, 453.316 to 453.336, inclusive, 453.337, 453.338, 453.3385 to 453.3405, inclusive, 453.560 or 453.562; or
- (b) An act forbidden by NRS 201.540 or any other sexual conduct or attempted sexual conduct with a pupil enrolled in an elementary or secondary school. As used in this paragraph, "sexual conduct" has the meaning ascribed to it in NRS 201.520.

- 5. "Postprobationary employee" means an administrator or a teacher who has completed the probationary period as provided in NRS 391.3197 and has been given notice of reemployment.
- 6. "Probationary employee" means an administrator or a teacher who is employed for the period set forth in NRS 391.3197.
- 7. "Superintendent" means the superintendent of a school district or a person designated by the board or superintendent to act as superintendent during the absence of the superintendent.
- 8. "Teacher" means a licensed employee the majority of whose working time is devoted to the rendering of direct educational service to pupils of a school district.
 - **Sec. 11.** NRS 391.3115 is hereby amended to read as follows:
- 391.3115 1. The demotion, suspension, dismissal and nonreemployment provisions of NRS 391.311 to 391.3197, inclusive, <u>and section 9 of this act</u> do not apply to:
 - (a) Substitute teachers; or
 - (b) Adult education teachers.
- 2. The admonition, demotion, suspension, dismissal and nonreemployment provisions of NRS 391.311 to 391.3194, inclusive, do not apply to:
- (a) A probationary teacher. The policy for evaluations prescribed in NRS 391.3125 and section 9 of this act applies to a probationary teacher.
- (b) A new employee who is employed as a probationary administrator. The policy for evaluations prescribed in NRS 391.3127 and section 9 of this act applies to a probationary administrator.
- 3. The admonition, demotion and suspension provisions of NRS 391.311 to 391.3194, inclusive, do not apply to a postprobationary teacher who is employed as a probationary administrator with respect to his or her employment in the administrative position. The policy for evaluations prescribed in NRS 391.3127 and section 9 of this act applies to a probationary administrator.
- 4. The provisions of NRS 391.311 to 391.3194, inclusive, and section 9 of this act do not apply to a teacher whose employment is suspended or terminated pursuant to subsection 3 of NRS 391.120 or NRS 391.3015 for failure to maintain a license in force.
- [3.] 5. A licensed employee who is employed in a position fully funded by a federal or private categorical grant or to replace another licensed employee during that employee's leave of absence is employed only for the duration of the grant or leave. Such a licensed employee and licensed employees who are employed on temporary contracts for 90 school days or less, or its equivalent in a school district operating under an alternative schedule authorized pursuant to NRS 388.090, to replace licensed employees

whose employment has terminated after the beginning of the school year are entitled to credit for that time in fulfilling any period of probation and during that time the provisions of NRS 391.311 to 391.3197, inclusive, *and section 9 of this act* for demotion, suspension or dismissal apply to them.

- **Sec. 12.** NRS 391.312 is hereby amended to read as follows:
- 391.312 1. A teacher may be suspended, dismissed or not reemployed and an administrator may be demoted, suspended, dismissed or not reemployed for the following reasons:
 - (a) Inefficiency;
 - (b) Immorality;
 - (c) Unprofessional conduct;
 - (d) Insubordination;
 - (e) Neglect of duty;
 - (f) Physical or mental incapacity;
- (g) A justifiable decrease in the number of positions due to decreased enrollment or district reorganization;
 - (h) Conviction of a felony or of a crime involving moral turpitude;
 - (i) Inadequate performance;
 - (j) Evident unfitness for service;
- (k) Failure to comply with such reasonable requirements as a board may prescribe;
- (1) Failure to show normal improvement and evidence of professional training and growth;
- (m) Advocating overthrow of the Government of the United States or of the State of Nevada by force, violence or other unlawful means, or the advocating or teaching of communism with the intent to indoctrinate pupils to subscribe to communistic philosophy;
- (n) Any cause which constitutes grounds for the revocation of a teacher's license;
- (o) Willful neglect or failure to observe and carry out the requirements of this title;
 - (p) Dishonesty;
- (q) Breaches in the security or confidentiality of the questions and answers of the achievement and proficiency examinations that are administered pursuant to NRS 389.015;
- (r) Intentional failure to observe and carry out the requirements of a plan to ensure the security of examinations adopted pursuant to NRS 389.616 or 389.620; [or]
 - (s) An intentional violation of NRS 388.5265 or 388.527 [.]; or
 - (t) Gross misconduct.
- 2. In determining whether the professional performance of a licensed employee is inadequate, consideration must be given to the regular and

special evaluation reports prepared in accordance with the policy of the employing school district and to any written standards of performance which may have been adopted by the board.

- 3. As used in this section, "gross misconduct" includes any act or omission that is in wanton, willful, reckless or deliberate disregard of the interests of a school or school district or a pupil thereof.
 - **Sec. 13.** NRS 391.3125 is hereby amended to read as follows:
- 391.3125 1. It is the intent of the Legislature that a uniform system be developed for objective evaluation of teachers and other licensed personnel in each school district.
- 2. Each board, following consultation with and involvement of elected representatives of the teachers or their designees, shall develop a policy for objective evaluations in narrative form. The policy must set forth a means according to which an employee's overall performance may be determined to be satisfactory or unsatisfactory. The policy may include an evaluation by the teacher, pupils, administrators or other teachers or any combination thereof. In a similar manner, counselors, librarians and other licensed personnel must be evaluated on forms developed specifically for their respective specialties. A copy of the policy adopted by the board must be filed with the Department. The primary purpose of an evaluation is to provide a format for constructive assistance. Evaluations, while not the sole criterion, must be used in the dismissal process.
- 3. A conference and a written evaluation for a probationary employee must be concluded not later than:
 - (a) December 1;
 - (b) February 1; and
 - (c) April 1,
- → of each school year of the probationary period, except that a probationary employee assigned to a school that operates all year must be evaluated at least three times during each 12 months of employment on a schedule determined by the board. An administrator charged with the evaluation of a probationary teacher shall personally observe the performance of the teacher in the classroom for not less than a cumulative total of 60 minutes during each evaluation period, with at least one observation during that 60-minute evaluation period consisting of at least 45 consecutive minutes.
- 4. [Whenever an administrator charged with the evaluation of a probationary employee believes the employee will not be reemployed for the second year of the probationary period or the school year following the probationary period, the administrator shall bring the matter to the employee's attention in a written document which is separate from the evaluation not later than March 1 of the current school year. The notice must include the reasons for the potential decision not to reemploy or refer to the

evaluation in which the reasons are stated. Such a notice is not required if the probationary employee has received a letter of admonition during the current school year.

- 5.] Each Except as otherwise provided in this subsection, each postprobationary teacher must be evaluated at least once each year. If a postprobationary teacher receives an unsatisfactory evaluation, the postprobationary teacher must be evaluated three times in the immediately succeeding school year. An administrator charged with the evaluation of a postprobationary teacher shall personally observe the performance of the teacher in the classroom for not less than a cumulative total of 60 minutes during each evaluation period, with at least one observation during that 60-minute evaluation period consisting of at least 30 consecutive minutes.
- [6.] If a postprobationary teacher is evaluated three times in a school year and he or she receives an unsatisfactory evaluation on the first or second evaluation, or both evaluations, the postprobationary teacher may request that the third evaluation be conducted by another administrator. If a postprobationary teacher requests that his or her third evaluation be conducted by another administrator, that administrator must be:
- (a) Employed by the school district or, if the school district has five or fewer administrators, employed by another school district in this State; and
- (b) Selected by the postprobationary teacher from a list of three candidates submitted by the superintendent.
- 5. The evaluation of a probationary teacher or a postprobationary teacher must include, without limitation:
 - (a) An evaluation of the classroom management skills of the teacher;
- (b) A review of the lesson plans and the work log or grade book of pupils prepared by the teacher;
- (c) An evaluation of whether the curriculum taught by the teacher is aligned with the standards of content and performance established pursuant to NRS 389.520, as applicable for the grade level taught by the teacher;
- (d) An evaluation of whether the teacher is appropriately addressing the needs of the pupils in the classroom, including, without limitation, special educational needs, cultural and ethnic diversity, the needs of pupils enrolled in advanced courses of study and the needs of pupils who are limited English proficient;
- (e) If necessary, recommendations for improvements in the performance of the teacher;
- (f) A description of the action that will be taken to assist the teacher in correcting any deficiencies reported in the evaluation; and
- (g) A statement by the administrator who evaluated the teacher indicating the amount of time that the administrator personally observed the performance of the teacher in the classroom.

- [7.] 6. The teacher must receive a copy of each evaluation not later than 15 days after the evaluation. A copy of the evaluation and the teacher's response must be permanently attached to the teacher's personnel file. Upon the request of a teacher, a reasonable effort must be made to assist the teacher to correct those deficiencies reported in the evaluation of the teacher for which the teacher requests assistance.
 - **Sec. 14.** NRS 391.3125 is hereby amended to read as follows:
- 391.3125 1. It is the intent of the Legislature that a uniform system be developed for objective evaluation of teachers and other licensed personnel in each school district.
- 2. Each board, following consultation with and involvement of elected representatives of the teachers or their designees, shall develop a policy for objective evaluations in narrative form. The policy must set forth a means according to which an employee's overall performance [may be] is determined to be [satisfactory or unsatisfactory.] highly effective, effective, minimally effective or ineffective. The policy must require that the information maintained pursuant to paragraphs (c), (d) and (e) of subsection 1 of NRS 386.650 account for at least 50 percent of the evaluation. The policy may include an evaluation by the teacher, pupils, administrators or other teachers or any combination thereof. In a similar manner, counselors, librarians and other licensed personnel must be evaluated on forms developed specifically for their respective specialties. A copy of the policy adopted by the board must be filed with the Department. The primary purpose of an evaluation is to provide a format for constructive assistance. Evaluations, while not the sole criterion, must be used in the dismissal process.
- 3. A conference and a written evaluation for a probationary employee must be concluded not later than:
 - (a) December 1;
 - (b) February 1; and
 - (c) April 1,
- → of each school year of the probationary period, except that a probationary employee assigned to a school that operates all year must be evaluated at least three times during each 12 months of employment on a schedule determined by the board. An administrator charged with the evaluation of a probationary teacher shall personally observe the performance of the teacher in the classroom for not less than a cumulative total of 60 minutes during each evaluation period, with at least one observation during that 60-minute evaluation period consisting of at least 45 consecutive minutes.
- 4. Except as otherwise provided in this subsection, each postprobationary teacher must be evaluated at least once each year. If a postprobationary teacher receives an **[unsatisfactory]** evaluation **[]** designating his or her

overall performance as minimally effective or ineffective, the postprobationary teacher must be evaluated three times in the immediately succeeding school year. An administrator charged with the evaluation of a postprobationary teacher shall personally observe the performance of the teacher in the classroom for not less than a cumulative total of 60 minutes during each evaluation period, with at least one observation during that 60-minute evaluation period consisting of at least 30 consecutive minutes. If a postprobationary teacher is evaluated three times in a school year and he or she receives an [unsatisfactory] evaluation designating his or her overall performance as minimally effective or ineffective on the first or second evaluation, or both evaluations, the postprobationary teacher may request that the third evaluation be conducted by another administrator. If a postprobationary teacher requests that his or her third evaluation be conducted by another administrator must be:

- (a) Employed by the school district or, if the school district has five or fewer administrators, employed by another school district in this State; and
- (b) Selected by the postprobationary teacher from a list of three candidates submitted by the superintendent.
- 5. The evaluation of a probationary teacher or a postprobationary teacher must include, without limitation:
 - (a) An evaluation of the classroom management skills of the teacher;
- (b) A review of the lesson plans and the work log or grade book of pupils prepared by the teacher;
- (c) An evaluation of whether the curriculum taught by the teacher is aligned with the standards of content and performance established pursuant to NRS 389.520, as applicable for the grade level taught by the teacher;
- (d) An evaluation of whether the teacher is appropriately addressing the needs of the pupils in the classroom, including, without limitation, special educational needs, cultural and ethnic diversity, the needs of pupils enrolled in advanced courses of study and the needs of pupils who are limited English proficient;
- (e) If necessary, recommendations for improvements in the performance of the teacher;
- (f) A description of the action that will be taken to assist the teacher in correcting any deficiencies reported in the evaluation; and
- (g) A statement by the administrator who evaluated the teacher indicating the amount of time that the administrator personally observed the performance of the teacher in the classroom.
- 6. The teacher must receive a copy of each evaluation not later than 15 days after the evaluation. A copy of the evaluation and the teacher's response must be permanently attached to the teacher's personnel file. Upon the request of a teacher, a reasonable effort must be made to assist the teacher to

correct those deficiencies reported in the evaluation of the teacher for which the teacher requests assistance.

- **Sec. 15.** NRS 391.3127 is hereby amended to read as follows:
- 391.3127 1. Each board, following consultation with and involvement of elected representatives of administrative personnel or their designated representatives, shall develop an objective policy for the objective evaluation of administrators in narrative form. The policy must set forth a means according to which an administrator's overall performance may be determined to be satisfactory or unsatisfactory. The policy may include an evaluation by the administrator, superintendent, pupils or other administrators or any combination thereof. A copy of the policy adopted by the board must be filed with the Department and made available to the Commission.
 - 2. Each administrator must be evaluated in writing at least once a year.
- 3. Each probationary administrator is subject to the provisions of NRS 391.3197 and section 9 of this act.
- 4. Before a superintendent transfers or assigns an administrator to another administrative position as part of an administrative reorganization, if the transfer or reassignment is to a position of lower rank, responsibility or pay, the superintendent shall give written notice of the proposed transfer or assignment to the administrator at least 30 days before the date on which it is to be effective. The administrator may appeal the decision of the superintendent to the board by requesting a hearing in writing to the president of the board within 5 days after receiving the notice from the superintendent. The board shall hear the matter within 10 days after the president receives the request, and shall render its decision within 5 days after the hearing. The decision of the board is final.
 - **Sec. 16.** NRS 391.3127 is hereby amended to read as follows:
- 391.3127 1. Each board, following consultation with and involvement of elected representatives of administrative personnel or their designated representatives, shall develop an objective policy for the objective evaluation of administrators in narrative form. The policy must set forth a means according to which an administrator's overall performance [may be] is determined to be [satisfactory or unsatisfactory.] highly effective, effective, minimally effective or ineffective. The policy must require that the information maintained pursuant to paragraphs (c), (d) and (e) of subsection 1 of NRS 386.650 account for at least 50 percent of the evaluation. The policy may include an evaluation by the administrator, superintendent, pupils or other administrators or any combination thereof. A copy of the policy adopted by the board must be filed with the Department and made available to the Commission.
 - 2. Each administrator must be evaluated in writing at least once a year.

- 3. Each probationary administrator is subject to the provisions of NRS 391.3197 and section 9 of this act.
- 4. Before a superintendent transfers or assigns an administrator to another administrative position as part of an administrative reorganization, if the transfer or reassignment is to a position of lower rank, responsibility or pay, the superintendent shall give written notice of the proposed transfer or assignment to the administrator at least 30 days before the date on which it is to be effective. The administrator may appeal the decision of the superintendent to the board by requesting a hearing in writing to the president of the board within 5 days after receiving the notice from the superintendent. The board shall hear the matter within 10 days after the president receives the request, and shall render its decision within 5 days after the hearing. The decision of the board is final.
 - **Sec. 17.** NRS 391.313 is hereby amended to read as follows:
- 391.313 1. Whenever an administrator charged with supervision of a licensed employee believes it is necessary to admonish the employee for a reason that the administrator believes may lead to demotion or dismissal or may cause the employee not to be reemployed under the provisions of NRS 391.312, the administrator shall:
- (a) Except as otherwise provided in subsection 3, bring the matter to the attention of the employee involved, in writing, stating the reasons for the admonition and that it may lead to the employee's demotion, dismissal or a refusal to reemploy him or her, and make a reasonable effort to assist the employee to correct whatever appears to be the cause for the employee's potential demotion, dismissal or a potential recommendation not to reemploy him or her; and
- (b) Except as otherwise provided in NRS 391.314, allow reasonable time for improvement, which must not exceed 3 months for the first admonition.
- The admonition must include a description of the deficiencies of the teacher and the action that is necessary to correct those deficiencies.
- 2. An admonition issued to a licensed employee who, within the time granted for improvement, has met the standards set for the employee by the administrator who issued the admonition must be removed from the records of the employee together with all notations and indications of its having been issued. The admonition must be removed from the records of the employee not later than 3 years after it is issued.
- 3. An administrator need not admonish an employee pursuant to paragraph (a) of subsection 1 if his or her employment will be terminated pursuant to NRS 391.3197. [If by March 1 of the first or second year of the employee's probationary period a probationary employee does not receive a written notice pursuant to subsection 4 of NRS 391.3125 of a potential

decision not to reemploy him or her, the employee must receive an admonition before any such decision is made.]

- 4. A licensed employee is subject to immediate dismissal or a refusal to reemploy according to the procedures provided in NRS 391.311 to 391.3197, inclusive, *and section 9 of this act* without the admonition required by this section, on grounds contained in paragraphs (b), (f), (g), (h), [and] (p) *and* (t) of subsection 1 of NRS 391.312.
 - **Sec. 18.** NRS 391.317 is hereby amended to read as follows:
- 391.317 1. At least 15 days before recommending to a board that it demote, dismiss or not reemploy a postprobationary employee, [or dismiss or demote a probationary employee,] the superintendent shall give written notice to the employee, by registered or certified mail, of the superintendent's intention to make the recommendation.
 - 2. The notice must:
 - (a) Inform the licensed employee of the grounds for the recommendation.
- (b) Inform the employee that, if a written request therefor is directed to the superintendent within 10 days after receipt of the notice, the employee is entitled to a hearing before a hearing officer [-] pursuant to NRS 391.315 to 391.3194, inclusive, or if a dismissal of the employee will occur before the completion of the current school year, the employee may request an expedited hearing pursuant to subsection 3.
 - (c) Refer to chapter 391 of NRS.
- 3. If a postprobationary employee receives notice pursuant to subsection 1 that he or she will be dismissed before the completion of the current school year, the employee may request an expedited hearing pursuant to the Expedited Labor Arbitration Procedures established by the American Arbitration Association or its successor organization. If the employee elects to proceed under the expedited procedures, the provisions of NRS 391.3161, 391.3192 and 391.3193 do not apply.
 - **Sec. 19.** NRS 391.3197 is hereby amended to read as follows:
- 391.3197 1. A probationary employee is employed on a contract basis for [two] three 1-year periods and has no right to employment after [either] any of the [two] three probationary contract years.
- 2. The board shall notify each probationary employee in writing on or before May 1 of the first, [and] second and third school years of the employee's probationary period, as appropriate, whether the employee is to be reemployed for the second or third year of the probationary period or for the [next] fourth school year as a postprobationary employee. Failure of the board to notify the probationary employee in writing on or before May 1 in the first or second year of the probationary period does not entitle the employee to postprobationary status. The employee must advise the board in writing on or before May 10 of the first, [or] second or third year of the

employee's probationary period, as appropriate, of the employee's acceptance of reemployment. If a probationary employee is assigned to a school that operates all year, the board shall notify the employee in writing, in [both] the first , [and] second and third years of the employee's probationary period, no later than 45 days before his or her last day of work for the year under his or her contract whether the employee is to be reemployed for the second or third year of the probationary period or for the [next] fourth school year as a postprobationary employee. Failure of the board to notify a probationary employee in writing within the prescribed period in the first or second year of the probationary period does not entitle the employee to postprobationary status. The employee must advise the board in writing within 10 days after the date of notification of his or her acceptance or rejection of reemployment for another year. Failure to advise the board of the employee's acceptance of reemployment pursuant to this subsection constitutes rejection of the contract.

- 3. A probationary employee who [completes] :
- (a) Completes a [2 year] 3-year probationary period [and receives];
- (b) Receives a designation of "satisfactory" on each of his or her performance evaluations for 2 consecutive school years; and
- (c) <u>Receives</u> a notice of reemployment from the school district in the <u>[second]</u> third year of the employee's probationary period.
- is entitled to be a postprobationary employee in the ensuing year of employment.
- 4. [If a probationary employee receives notice pursuant to subsection 4 of NRS 391.3125 not later than March 1 of a potential decision not to reemploy him or her, the employee may request a supplemental evaluation by another administrator in the school district selected by the employee and the superintendent. If a school district has five or fewer administrators, the supplemental evaluator may be an administrator from another school district in this State. If a probationary employee has received during the first school year of the employee's probationary period three evaluations which state that the employee's overall performance has been satisfactory, the superintendent of schools of the school district or the superintendent's designee shall waive the second year of the employee's probationary period by expressly providing in writing on the final evaluation of the employee for the first probationary year that the second year of the employee's probationary period is waived. Such an employee is entitled to be a postprobationary employee in the ensuing year of employment.
- 5.] If a probationary employee is notified that the employee will not be reemployed for the [second year of the employee's probationary period or the ensuing] school year [.] following the 3-year probationary period, his or her employment ends on the last day of the current school year. The notice that

the employee will not be reemployed must include a statement of the reasons for that decision.

- [6.] 5. A new employee who is employed as an administrator or a postprobationary teacher who is employed as an administrator shall be deemed to be a probationary employee for the purposes of this section and must serve a [2-year] 3-year probationary period as an administrator in accordance with the provisions of this section. [If the administrator does not receive an unsatisfactory evaluation during the first year of probation, the superintendent or the superintendent's designee shall waive the second year of the administrator's probationary period. Such an administrator is entitled to be a postprobationary employee in the ensuing year of employment.] If:
- (a) A postprobationary teacher who is an administrator is not reemployed as an administrator after [either] any year of his or her probationary period; and
- (b) There is a position as a teacher available for the ensuing school year in the school district in which the person is employed,
- → the board of trustees of the school district shall, on or before May 1, offer the person a contract as a teacher for the ensuing school year. The person may accept the contract in writing on or before May 10. If the person fails to accept the contract as a teacher, the person shall be deemed to have rejected the offer of a contract as a teacher.
- [7.] 6. An administrator who has completed his or her probationary period pursuant to subsection [6] 5 and is thereafter promoted to the position of principal must serve an additional probationary period of 1 year in the position of principal. If an administrator is promoted to the position of principal before completion of his or her probationary period pursuant to subsection 5, the administrator must serve the remainder of his or her probationary period pursuant to subsection 5 or an additional probationary period of 1 year in the position of principal, whichever is longer. If the administrator serving the additional probationary period is not reemployed as a principal after the expiration of the *probationary period or* additional probationary period, as applicable, the board of trustees of the school district in which the person is employed shall, on or before May 1, offer the person a contract for the ensuing school year for the administrative position in which the person attained postprobationary status. The person may accept the contract in writing on or before May 10. If the person fails to accept such a contract, the person shall be deemed to have rejected the offer of employment.
- [8: Before dismissal, the probationary employee is entitled to a hearing before a hearing officer which affords due process as set out in NRS 391.311 to 391.3196, inclusive.]

7. If a probationary employee receives notice that he or she will be dismissed before the completion of the current school year, the probationary employee may request an expedited hearing pursuant to the Expedited Labor Arbitration Procedures established by the American Arbitration Association or its successor organization.

Sec. 19.5. NRS 391.3197 is hereby amended to read as follows:

- 391.3197 1. A probationary employee is employed on a contract basis for three 1-year periods and has no right to employment after any of the three probationary contract years.
- 2. The board shall notify each probationary employee in writing on or before May 1 of the first, second and third school years of the employee's probationary period, as appropriate, whether the employee is to be reemployed for the second or third year of the probationary period or for the fourth school year as a postprobationary employee. Failure of the board to notify the probationary employee in writing on or before May 1 in the first or second year of the probationary period does not entitle the employee to postprobationary status. The employee must advise the board in writing on or before May 10 of the first, second or third year of the employee's probationary period, as appropriate, of the employee's acceptance of reemployment. If a probationary employee is assigned to a school that operates all year, the board shall notify the employee in writing, in the first, second and third years of the employee's probationary period, no later than 45 days before his or her last day of work for the year under his or her contract whether the employee is to be reemployed for the second or third year of the probationary period or for the fourth school year as a postprobationary employee. Failure of the board to notify a probationary employee in writing within the prescribed period in the first or second year of the probationary period does not entitle the employee to postprobationary status. The employee must advise the board in writing within 10 days after the date of notification of his or her acceptance or rejection of reemployment for another year. Failure to advise the board of the employee's acceptance of reemployment pursuant to this subsection constitutes rejection of the contract.
 - 3. A probationary employee who:
 - (a) Completes a 3-year probationary period;
- (b) Receives a designation of ["satisfactory"] "highly effective" or "effective" on each of his or her performance evaluations for 2 consecutive school years; and
- (c) Receives a notice of reemployment from the school district in the third year of the employee's probationary period,
- is entitled to be a postprobationary employee in the ensuing year of employment.

- 4. If a probationary employee is notified that the employee will not be reemployed for the school year following the 3-year probationary period, his or her employment ends on the last day of the current school year. The notice that the employee will not be reemployed must include a statement of the reasons for that decision.
- 5. A new employee who is employed as an administrator or a postprobationary teacher who is employed as an administrator shall be deemed to be a probationary employee for the purposes of this section and must serve a 3-year probationary period as an administrator in accordance with the provisions of this section. If:
- (a) A postprobationary teacher who is an administrator is not reemployed as an administrator after any year of his or her probationary period; and
- (b) There is a position as a teacher available for the ensuing school year in the school district in which the person is employed,
- the board of trustees of the school district shall, on or before May 1, offer the person a contract as a teacher for the ensuing school year. The person may accept the contract in writing on or before May 10. If the person fails to accept the contract as a teacher, the person shall be deemed to have rejected the offer of a contract as a teacher.
- 6. An administrator who has completed his or her probationary period pursuant to subsection 5 and is thereafter promoted to the position of principal must serve an additional probationary period of 1 year in the position of principal. If an administrator is promoted to the position of principal before the completion of his or her probationary period pursuant to subsection 5, the administrator must serve the remainder of his or her probationary period pursuant to subsection 5 or an additional probationary period of 1 year in the position of principal, whichever is longer. If the administrator serving the additional probationary period is not reemployed as a principal after the expiration of the probationary period or additional probationary period, as applicable, the board of trustees of the school district in which the person is employed shall, on or before May 1, offer the person a contract for the ensuing school year for the administrative position in which the person attained postprobationary status. The person may accept the contract in writing on or before May 10. If the person fails to accept such a contract, the person shall be deemed to have rejected the offer of employment.
- 7. If a probationary employee receives notice that he or she will be dismissed before the completion of the current school year, the employee may request an expedited hearing pursuant to the Expedited Labor Arbitration Procedures established by the American Arbitration Association or its successor organization.

Sec. 19.6. Chapter 288 of NRS is hereby amended by adding thereto a new section to read as follows:

If the board of trustees of a school district determines that a reduction in the existing workforce of the licensed educational personnel in the school district is necessary, the decision to lay off a teacher or an administrator must not be based solely on the seniority of the teacher or administrator and may include, without limitation, a consideration of the following factors:

- 1. Whether the teacher or administrator is employed in a position which is hard to fill:
- 2. Whether the teacher or administrator has received a national board certification;
 - 3. The performance evaluations of the teacher or administrator;
- 4. The disciplinary record of the teacher or administrator within the school district;
 - 5. The criminal record of the teacher or administrator, if any;
 - 6. The type of licensure held by the teacher or administrator; and
- 7. The type of degree attained by the teacher or administrator and whether the degree is in a subject area that is related to his or her position.

Sec. 19.7. NRS 288.150 is hereby amended to read as follows:

- 288.150 1. Except as provided in subsection 4, every local government employer shall negotiate in good faith through one or more representatives of its own choosing concerning the mandatory subjects of bargaining set forth in subsection 2 with the designated representatives of the recognized employee organization, if any, for each appropriate bargaining unit among its employees. If either party so requests, agreements reached must be reduced to writing.
 - 2. The scope of mandatory bargaining is limited to:
 - (a) Salary or wage rates or other forms of direct monetary compensation.
 - (b) Sick leave.
 - (c) Vacation leave.
 - (d) Holidays.
 - (e) Other paid or nonpaid leaves of absence.
 - (f) Insurance benefits.
- (g) Total hours of work required of an employee on each workday or workweek.
 - (h) Total number of days' work required of an employee in a work year.
 - (i) Discharge and disciplinary procedures.
 - (j) Recognition clause.
 - (k) The method used to classify employees in the bargaining unit.
 - (1) Deduction of dues for the recognized employee organization.

- (m) Protection of employees in the bargaining unit from discrimination because of participation in recognized employee organizations consistent with the provisions of this chapter.
 - (n) No-strike provisions consistent with the provisions of this chapter.
- (o) Grievance and arbitration procedures for resolution of disputes relating to interpretation or application of collective bargaining agreements.
 - (p) General savings clauses.
 - (q) Duration of collective bargaining agreements.
 - (r) Safety of the employee.
 - (s) Teacher preparation time.
 - (t) Materials and supplies for classrooms.
 - (u) The policies for the transfer and reassignment of teachers.
- (v) Procedures for reduction in workforce [...] consistent with the provisions of this chapter.
- 3. Those subject matters which are not within the scope of mandatory bargaining and which are reserved to the local government employer without negotiation include:
- (a) Except as otherwise provided in paragraph (u) of subsection 2, the right to hire, direct, assign or transfer an employee, but excluding the right to assign or transfer an employee as a form of discipline.
- (b) The right to reduce in force or lay off any employee because of lack of work or lack of money, subject to paragraph (v) of subsection 2.
 - (c) The right to determine:
- (1) Appropriate staffing levels and work performance standards, except for safety considerations;
- (2) The content of the workday, including without limitation workload factors, except for safety considerations;
 - (3) The quality and quantity of services to be offered to the public; and
 - (4) The means and methods of offering those services.
 - (d) Safety of the public.
- 4. Notwithstanding the provisions of any collective bargaining agreement negotiated pursuant to this chapter, a local government employer is entitled to take whatever actions may be necessary to carry out its responsibilities in situations of emergency such as a riot, military action, natural disaster or civil disorder. Those actions may include the suspension of any collective bargaining agreement for the duration of the emergency. Any action taken under the provisions of this subsection must not be construed as a failure to negotiate in good faith.
- 5. The provisions of this chapter, including without limitation the provisions of this section, recognize and declare the ultimate right and responsibility of the local government employer to manage its operation in

the most efficient manner consistent with the best interests of all its citizens, its taxpayers and its employees.

- 6. This section does not preclude, but this chapter does not require the local government employer to negotiate subject matters enumerated in subsection 3 which are outside the scope of mandatory bargaining. The local government employer shall discuss subject matters outside the scope of mandatory bargaining but it is not required to negotiate those matters.
- 7. Contract provisions presently existing in signed and ratified agreements as of May 15, 1975, at 12 p.m. remain negotiable.

Sec. 19.8. NRS 288.195 is hereby amended to read as follows:

- 288.195 Whenever an employee organization enters into negotiations with a local government employer, pursuant to NRS 288.140 to 288.220, inclusive, *and section 19.6 of this act*, such employee organization may be represented by an attorney licensed to practice law in the State of Nevada.
 - **Sec. 20.** Section 9 of this act is hereby amended to read as follows:
- Sec. 9. 1. If a written evaluation of a probationary teacher or probationary administrator designates the overall performance of the teacher or administrator as ["unsatisfactory":] "minimally effective" or "ineffective":
- (a) The written evaluation must include the following statement: "Please be advised that, pursuant to Nevada law, your contract may not be renewed for the next school year. If you receive two evaluations for this school year which designate your performance as ['unsatisfactory,'] 'minimally effective' or 'ineffective,' and if you have another evaluation remaining this school year, you may request that the evaluation be conducted by another administrator. You may also request, to the administrator who conducted the evaluation, reasonable assistance in correcting the deficiencies reported in the evaluation for which you request assistance, and upon such request, a reasonable effort will be made to assist you in correcting those deficiencies."
- (b) The probationary teacher or probationary administrator, as applicable, must acknowledge in writing that he or she has received and understands the statement described in paragraph (a).
- 2. If a probationary teacher or probationary administrator requests that his or her next evaluation be conducted by another administrator in accordance with the notice required by subsection 1, the administrator conducting the evaluation must be:
- (a) Employed by the school district or, if the school district has five or fewer administrators, employed by another school district in this State; and
- (b) Selected by the probationary teacher or probationary administrator, as applicable, from a list of three candidates submitted by the superintendent.
- 3. If a probationary teacher or probationary administrator requests assistance in correcting deficiencies reported in his or her evaluation, the

administrator who conducted the evaluation shall ensure that a reasonable effort is made to assist the probationary teacher or probationary administrator in correcting those deficiencies.

Sec. 20.5. Section 1 of Assembly Bill No. 225 of this session is hereby amended to read as follows:

Section 1. Chapter 391 of NRS is hereby amended by adding thereto a new section to read as follows:

A postprobationary employee who receives an [unsatisfactory evaluation pursuant to NRS 391.3125 or 391.3127, as applicable, or any other equivalent] evaluation designating his or her overall performance as [below average,]:

- 1. If evaluated pursuant to NRS 391.3125 or 391.3127, as applicable:
- (a) Minimally effective;
- (b) Ineffective; or
- (c) Minimally effective during 1 year of the 2-year consecutive period and ineffective during the other year of the period; or
- 2. If evaluated pursuant to any other system of evaluation, any designation which indicates that the overall performance of the employee is below average,
- for 2 consecutive school years shall be deemed to be a probationary employee for the purposes of NRS 391.311 to 391.3197, inclusive, <u>and section 9 of Assembly Bill No. 229 of this session</u>, and must serve an additional probationary period in accordance with the provisions of NRS 391.3197.
- **Sec. 21.** The provisions of section 9 of this act, NRS 391.311 to 391.3125, inclusive, as amended by sections 10 to 13, inclusive, of this act, NRS 391.3127, as amended by section 15 of this act, NRS 391.313, as amended by section 17 of this act, NRS 391.317, as amended by section 18 of this act, and NRS 391.3197, as amended by section 19 of this act, apply to all:
- 1. Teachers who are initially employed by a school district on or after July 1, 2011.
- 2. A new employee who is hired by a school district as an administrator on or after July 1, 2011.
- 3. A postprobationary teacher who is employed as an administrator on or after July 1, 2011.
 - **Sec. 22.** The board of trustees of each school district shall:
- 1. Commencing with the 2013-2104 school year, implement and carry out the policies for evaluations of teachers and administrators required by NRS 391.3125, as amended by section 14 of this act, NRS 391.3127, as amended by section 16 of this act, NRS 391.3197, as amended by section 19.5 of this act, and section 20 of this act.

- 2. <u>Commencing with the 2013-2014 school year, implement and carry out section 20.5 of this act if, and only if, Assembly Bill No. 225 of this session is enacted by the Legislature and becomes effective.</u>
- 3. Commencing with the 2014-2015 school year, implement and carry out the program of performance pay and enhanced compensation established by the board of trustees pursuant to section 8 of this act.
- **Sec. 23.** 1. This section and sections 1 to 7, inclusive, 9 to 13, inclusive, 15, 17, 18, 19, <u>19.6</u>, <u>19.7</u>, <u>19.8</u>, 21 and 22 of this act become effective on July 1, 2011.
- 2. Sections 8, 14, 16, 19.5 and 20 of this act become effective on July 1, 2013.
- 3. Section 20.5 of this act becomes effective on July 1, 2013, if, and only if, Assembly Bill No. 225 of this session is enacted by the Legislature and becomes effective.

Assemblyman Bobzien moved that the Assembly concur in the Senate Amendment No. 861 to Assembly Bill No. 229.

Remarks by Assemblyman Bobzien.

Motion carried by a constitutional majority.

Bill ordered enrolled.

GENERAL FILE AND THIRD READING

Assembly Bill No. 380.

Bill read third time.

The following amendment was proposed by the Committee on Ways and Means:

Amendment No. 862.

AN ACT relating to energy; revising the prospective expiration of the Wind Energy Systems Demonstration Program and the Waterpower Energy Systems Demonstration Program; providing for the expiration of the Solar Energy Systems Incentive Program; frevising provisions governing the Solar Program, the Wind Program and the Waterpower Program; revising provisions governing the amount of incentives which a utility must pay to participants in the Solar Program, the Wind Program and the Waterpower Program; revising provisions relating to net metering; requiring the Public Utilities Commission of Nevada to adopt regulations;] and providing other matters properly relating thereto.

Legislative Counsel's Digest:

Existing law provides that the Wind Energy Systems Demonstration Program and the Waterpower Energy Systems Demonstration Program will expire on June 30, 2011. (NRS 701B.400-701B.890; chapter 509, Statutes of Nevada 2007, p. 2999) [Sections 49 51 and 54 of this] This bill [revise] revises the prospective expiration dates of these programs and [provide]

<u>provides</u> that the Wind Program, the Waterpower Program and the Solar Energy Systems Incentive Program will expire on December 31, 2021.

E Section 13 of this bill provides that, for the period beginning July 1, 2010, and ending December 31, 2012, a utility is not required to award an incentive under the Solar Program if the payment of the incentive would cause the total amount of incentives paid by all utilities for the installation of solar energy systems and distributed generation systems would exceed \$140,000,000. Section 26 of this bill prohibits the Public Utilities Commission of Nevada from authorizing the payment of an incentive for the installation of a wind energy system under the Wind Program if, for the same period, the amount of the incentive would cause the total amount of incentives paid by all utilities for the installation of wind energy systems and distributed generation systems to exceed \$30,000,000.

Sections 1, 3-12, 14-25, 27-43 and 47 of this bill become effective on January 1, 2013, and revise provisions governing the Solar Program, the Wind Program and the Waterpower Program. Section 5 provides limits on the total amount of an incentive which may be awarded to a participant in the Wind Program. Section 5 additionally provides that a utility is not required to award an incentive if such an award would cause the total amount of incentives awarded pursuant to the Wind Program, the Solar Program and the Waterpower Program in a program year by all utilities in this State to exceed one half of 1 percent of the total revenues received by all utilities in this State from retail customers during the immediately preceding program year. Section 9 imposes the same requirements for the payment of incentives pursuant to the Solar Program. Section 37 provides the same cap for the award of incentives pursuant to the Waterpower Program. Sections 14 and 29 provide that to be eligible for an incentive under the Solar Program or Wind Program, a renewable energy system must meet the requirements for net metering. Sections 11, 22 and 35 expand the Solar Program, Wind Program and Waterpower Program to include Indian tribes and tribal organizations.

Existing law requires a utility to offer net metering until the cumulative capacity of all net metering systems within the service area of the utility is equal to 1 percent of the utility's peak capacity. (NRS 704.773) Section 45 of this bill requires a utility to offer net metering until the cumulative capacity of all net metering systems in this State is equal to 3 percent of the total peak capacity of all utilities in this State.]

THE PEOPLE OF THE STATE OF NEVADA, REPRESENTED IN SENATE AND ASSEMBLY, DO ENACT AS FOLLOWS:

Section 1. (Deleted by amendment.)

Sec. 2. NRS 701.180 is hereby amended to read as follows:

701.180 The Director shall:

- 1. Acquire and analyze information relating to energy and to the supply, demand and conservation of its sources, including, without limitation [:
- (a) Information relating to the Solar Energy Systems Incentive Program created pursuant to NRS 701B.240 and the Wind Energy Systems Incentive Program created pursuant to 701B.580, including, without limitation, information relating to:
- (1) The development of distributed generation systems in this State pursuant to participation in the Solar Energy Systems Incentive Program;
- (2) The use of carbon-based energy in residential and commercial applications due to participation in the Programs; and
- (3) The average cost of generation on a kilowatt hour basis for residential and commercial applications due to participation in the Programs; and
- (b) Information], *information* relating to any money distributed pursuant to NRS 702.270.
- 2. Review and evaluate information which identifies trends and permits forecasting of the energy available to the State. Such forecasts must include estimates on:
- (a) The level of demand for energy in the State for 5-, 10- and 20-year periods;
 - (b) The amount of energy available to meet each level of demand;
- (c) The probable implications of the forecast on the demand and supply of energy; and
- (d) The sources of renewable energy and other alternative sources of energy which are available and their possible effects.
- 3. Study means of reducing wasteful, inefficient, unnecessary or uneconomical uses of energy and encourage the maximum utilization of existing sources of energy in the State.
- 4. Solicit and serve as the point of contact for grants and other money from the Federal Government, including, without limitation, any grants and other money available pursuant to any program administered by the United States Department of Energy, and other sources to cooperate with the Commissioner and the Authority:
- (a) To promote energy projects that enhance the economic development of the State;
 - (b) To promote the use of renewable energy in this State;
- (c) To promote the use of measures which conserve or reduce the demand for energy or which result in more efficient use of energy;
- (d) To develop a comprehensive program for retrofitting public buildings in this State with energy efficiency measures; and
- (e) If the Commissioner determines that it is feasible and cost-effective, to enter into contracts with researchers from the Nevada System of Higher

Education for the design of energy efficiency and retrofit projects to carry out the comprehensive program for retrofitting public buildings in this State developed pursuant to paragraph (d).

- 5. Coordinate the activities and programs of the Office of Energy with the activities and programs of the Authority, the Consumer's Advocate and the Public Utilities Commission of Nevada, and with other federal, state and local officers and agencies that promote, fund, administer or operate activities and programs related to the use of renewable energy and the use of measures which conserve or reduce the demand for energy or which result in more efficient use of energy.
- 6. If requested to make a determination pursuant to NRS 111.239 or 278.0208, make the determination within 30 days after receiving the request. If the Director needs additional information to make the determination, the Director may request the information from the person making the request for a determination. Within 15 days after receiving the additional information, the Director shall make a determination on the request.
- 7. Carry out all other directives concerning energy that are prescribed by the Governor.
 - Sec. 3. (Deleted by amendment.)
 - Sec. 4. (Deleted by amendment.)
 - Sec. 5. (Deleted by amendment.)
 - Sec. 6. (Deleted by amendment.)
 - Sec. 7. (Deleted by amendment.)
 - Sec. 8. (Deleted by amendment.)
 - Sec. 9. (Deleted by amendment.)
 - Sec. 10. (Deleted by amendment.)
 - Sec. 11. (Deleted by amendment.)
 - Sec. 12. (Deleted by amendment.)
 - Sec. 13. (Deleted by amendment.)
 - Sec. 14. (Deleted by amendment.)
 - Sec. 15. (Deleted by amendment.)
 - Sec. 16. (Deleted by amendment.)
 - Sec. 17. (Deleted by amendment.)
 - Sec. 18. (Deleted by amendment.)
 - Sec. 19. (Deleted by amendment.)
 - Sec. 20. (Deleted by amendment.)
 - Sec. 21. (Deleted by amendment.)
 - Sec. 22. (Deleted by amendment.)
 - Sec. 23. (Deleted by amendment.)
 - Sec. 24. (Deleted by amendment.)
 - Sec. 25. (Deleted by amendment.)
 - Sec. 26. (Deleted by amendment.)

- Sec. 27. (Deleted by amendment.)
- Sec. 27.5. (Deleted by amendment.)
- Sec. 28. (Deleted by amendment.)
- Sec. 29. (Deleted by amendment.)
- Sec. 30. (Deleted by amendment.)
- Sec. 31. (Deleted by amendment.)
- Sec. 32. (Deleted by amendment.)
- Sec. 33. (Deleted by amendment.)
- Sec. 34. (Deleted by amendment.)
- Sec. 35. (Deleted by amendment.)
- Sec. 36. (Deleted by amendment.)
- Sec. 37. (Deleted by amendment.)
- Sec. 38. (Deleted by amendment.)
- Sec. 39. (Deleted by amendment.)
- Sec. 40. (Deleted by amendment.)
- Sec. 41. (Deleted by amendment.)
- Sec. 42. (Deleted by amendment.)
- **Sec. 43.** NRS 701B.924 is hereby amended to read as follows:
- 701B.924 1. The State Public Works Board shall, within 90 days after June 9, 2009, determine the specific projects to weatherize and retrofit public buildings, facilities and structures, including, without limitation, traffic-control systems, and to otherwise use sources of renewable energy to serve those buildings, facilities and structures pursuant to the provisions of this section and NRS 701B.921. The projects must be prioritized and selected on the basis of the following criteria:
 - (a) The length of time necessary to commence the project.
 - (b) The number of workers estimated to be employed on the project.
 - (c) The effectiveness of the project in reducing energy consumption.
 - (d) The estimated cost of the project.
- (e) Whether the project is able to be powered by or to otherwise use sources of renewable energy.
- (f) Whether the project has qualified for participation in [one or more of the following programs:
- (1) The Solar Energy Systems Incentive Program created by NRS 701B.240:

(2)] :

- (1) The Renewable Energy School Pilot Program created by NRS 701B.350:
- [(3) The Wind Energy Systems] [Demonstration] [Hincentive Program ereated by NRS 701B.580:
- (4) The Waterpower Energy Systems] [Demonstration] [Incentive Program created by NRS 701B.820; or

(5)] or

- (2) An energy efficiency or energy conservation program offered by a public utility, as defined in NRS 704.020, pursuant to a plan approved by the Public Utilities Commission of Nevada pursuant to NRS 704.741.
- 2. The board of trustees of each school district shall, within 90 days after June 9, 2009, determine the specific projects to weatherize and retrofit public buildings, facilities and structures, including, without limitation, traffic-control systems, and to otherwise use sources of renewable energy to serve those buildings, facilities and structures pursuant to the provisions of this section and NRS 701B.921. The projects must be prioritized and selected on the basis of the following criteria:
 - (a) The length of time necessary to commence the project.
 - (b) The number of workers estimated to be employed on the project.
 - (c) The effectiveness of the project in reducing energy consumption.
 - (d) The estimated cost of the project.
- (e) Whether the project is able to be powered by or to otherwise use sources of renewable energy.
- (f) Whether the project has qualified for participation in $\frac{\text{Fone or more of the following programs:}}{\text{the following programs:}}$
- (1) The Solar Energy Systems Incentive Program created by NRS 701B.240:

 $\frac{(2)}{(2)}$:

- (1) The Renewable Energy School Pilot Program created by NRS 701B.350;
- [(3) The Wind Energy Systems] [Demonstration] [Hincentive Program created by NRS 701B.580;
- (4) The Waterpower Energy Systems] [Demonstration] [Incentive Program created by NRS 701B.820; or

(5)] or

- (2) An energy efficiency or energy conservation program offered by a public utility, as defined in NRS 704.020, pursuant to a plan approved by the Public Utilities Commission of Nevada pursuant to NRS 704.741.
- 3. The Board of Regents of the University of Nevada shall, within 90 days after June 9, 2009, determine the specific projects to weatherize and retrofit public buildings, facilities and structures, including, without limitation, traffic-control systems, and to otherwise use sources of renewable energy to serve those buildings, facilities and structures pursuant to the provisions of this section and NRS 701B.921. The projects must be prioritized and selected on the basis of the following criteria:
 - (a) The length of time necessary to commence the project.
 - (b) The number of workers estimated to be employed on the project.
 - (c) The effectiveness of the project in reducing energy consumption.

- (d) The estimated cost of the project.
- (e) Whether the project is able to be powered by or to otherwise use sources of renewable energy.
- (f) Whether the project has qualified for participation in [one or more of the following programs:
- (1) The Solar Energy Systems Incentive Program ereated by NRS 701B.240:

 $\frac{(2)}{(2)}$:

- (1) The Renewable Energy School Pilot Program created by NRS 701B.350;
- [(3) The Wind Energy Systems] [Demonstration] [Hincentive Program ereated by NRS 701B.580:
- (4) The Waterpower Energy Systems] [Demonstration] [Incentive Program created by NRS 701B.820; or

(5)] <u>or</u>

- (2) An energy efficiency or energy conservation program offered by a public utility, as defined in NRS 704.020, pursuant to a plan approved by the Public Utilities Commission of Nevada pursuant to NRS 704.741.
- 4. As soon as practicable after an entity described in subsections 1, 2 and 3 selects a project, the entity shall proceed to enter into a contract with one or more contractors to perform the work on the project. The request for proposals and all contracts for each project must include, without limitation:
- (a) Provisions stipulating that all employees of the contractors and subcontractors who work on the project must be paid prevailing wages pursuant to the requirements of chapter 338 of NRS;
- (b) Provisions requiring that each contractor and subcontractor employed on each such project:
- (1) Employ a number of persons trained as described in paragraph (b) of subsection 3 of NRS 701B.921 that is equal to or greater than 50 percent of the total workforce the contractor or subcontractor employs on the project; or
- (2) If the Director of the Department determines in writing, pursuant to a request submitted by the contractor or subcontractor, that the contractor or subcontractor cannot reasonably comply with the provisions of subparagraph (1) because there are not available a sufficient number of such trained persons, employ a number of persons trained as described in paragraph (b) of subsection 3 of NRS 701B.921 or trained through any apprenticeship program that is registered and approved by the State Apprenticeship Council pursuant to chapter 610 of NRS that is equal to or greater than 50 percent of the total workforce the contractor or subcontractor employs on the project;
- (c) A component pursuant to which persons trained as described in paragraph (b) of subsection 3 of NRS 701B.921 must be classified and paid

prevailing wages depending upon the classification of the skill in which they are trained; and

- (d) A component that requires each contractor or subcontractor to offer to employees working on the project, and to their dependents, health care in the same manner as a policy of insurance pursuant to chapters 689A and 689B of NRS or the Employee Retirement Income Security Act of 1974.
- 5. The State Public Works Board, each of the school districts and the Board of Regents of the University of Nevada shall each provide a report to the Interim Finance Committee which describes the projects selected pursuant to this section and a report of the dates on which those projects are scheduled to be completed.
 - Sec. 44. (Deleted by amendment.)
 - Sec. 45. (Deleted by amendment.)
 - **Sec. 46.** (Deleted by amendment.)
 - **Sec. 47.** NRS 338.1908 is hereby amended to read as follows:
- 338.1908 1. The governing body of each local government shall, by July 28, 2009, develop a plan to retrofit public buildings, facilities and structures, including, without limitation, traffic-control systems, and to otherwise use sources of renewable energy to serve those buildings, facilities and structures. Such a plan must:
- (a) Be developed with input from one or more energy retrofit coordinators designated pursuant to NRS 338.1907, if any.
- (b) Include a list of specific projects. The projects must be prioritized and selected on the basis of the following criteria:
 - (1) The length of time necessary to commence the project.
 - (2) The number of workers estimated to be employed on the project.
 - (3) The effectiveness of the project in reducing energy consumption.
 - (4) The estimated cost of the project.
- (5) Whether the project is able to be powered by or otherwise use sources of renewable energy.
- (6) Whether the project has qualified for participation in [one or more of the following programs:
- (I) The Solar Energy Systems Incentive Program created by NRS 701B-240:
- (H) The] the Renewable Energy School Pilot Program created by NRS 701B.350. [+;
- (III) The Wind Energy Systems] [Demonstration] [Incentive Program created by NRS 701B.580; or
- (IV) The Waterpower Energy Systems] [Demonstration] [Incentive Program created by NRS 701B.820.]
- (c) Include a list of potential funding sources for use in implementing the projects, including, without limitation, money available through the Energy

Efficiency and Conservation Block Grant Program as set forth in 42 U.S.C. § 17152 and grants, gifts, donations or other sources of money from public and private sources.

- 2. The governing body of each local government shall transmit the plan developed pursuant to subsection 1 to the Nevada Energy Commissioner and to any other entity designated for that purpose by the Legislature.
 - 3. As used in this section:
- (a) "Local government" means each city or county that meets the definition of "eligible unit of local government" as set forth in 42 U.S.C. § 17151 and each unit of local government, as defined in subsection 11 of NRS 338.010, that does not meet the definition of "eligible entity" as set forth in 42 U.S.C. § 17151.
- (b) "Renewable energy" means a source of energy that occurs naturally or is regenerated naturally, including, without limitation:
 - (1) Biomass;
 - (2) Fuel cells;
 - (3) Geothermal energy;
 - (4) Solar energy;
 - (5) Waterpower; and
 - (6) Wind.
- → The term does not include coal, natural gas, oil, propane or any other fossil fuel, or nuclear energy.
- (c) "Retrofit" means to alter, improve, modify, remodel or renovate a building, facility or structure to make that building, facility or structure more energy-efficient.

Sec. 48. (Deleted by amendment.)

Sec. 49. Section 113 of chapter 509, Statutes of Nevada 2007, at page 2999, is hereby amended to read as follows:

Sec. 113. 1. This act becomes effective:

- (a) Upon passage and approval for the purposes of adopting regulations and taking such other actions as are necessary to carry out the provisions of this act; and
 - (b) For all other purposes besides those described in paragraph (a):
- (1) For this section and sections 1, 30, 32, 36 to 46, inclusive, 49, 51 to 61, inclusive, 107, 109, 110 and 111 of this act, upon passage and approval.
- (2) For sections 1.5 to 29, inclusive, 43.5, 47, 51.3, 51.7, 108, 112 and 112.5 of this act, on July 1, 2007.
 - (3) For sections 62 to 106, inclusive, of this act, on October 1, 2007.
- (4) For sections 31, 32.3, 32.5, 32.7, 33, 34 and 35 of this act, on January 1, 2009.
 - (5) For section 48 of this act, on January 1, 2010.
 - (6) For section 50 of this act, on January 1, 2011.

- 2. Sections <u>62 to [106,]</u> [63 to] 75, inclusive, 77 to 82, inclusive, 85_f, 86, 88] to 94, inclusive, and 95 to 105, inclusive, of this act expire by limitation on [June 30, 2011.] December 31, 2021.
- **Sec. 50.** Section 13 of chapter 246, Statutes of Nevada 2009, at page 1002, is hereby amended to read as follows:
 - Sec. 13. 1. This act becomes effective on July 1, 2009.
- 2. Sections 2 and 3 of this act expire by limitation on [June 30, 2011.] **December 31, 2021.**
- **Sec. 51.** Section 21 of chapter 321, Statutes of Nevada 2009, at page 1410, is hereby amended to read as follows:
- Sec. 21. 1. This section and sections 1 to 1.51, inclusive, 1.55 to 19.7, inclusive, and 19.9 to 20.9, inclusive, of this act become effective upon passage and approval.
- 2. Sections 1.51, 1.85, 1.87, 1.92, <u>1.93</u>, 1.95, 4.3 *fto 7, inclusive*, *7.31* to 9, inclusive, and 19.4 of this act expire by limitation on [June 30, 2011.
- 3. Sections 1.53 and 19.8 of this act become effective on July 1, 2011.] December 31, 2021.
- **Sec. 52.** 1. NRS 701B.010, 701B.020, 701B.030, 701B.040, 701B.050, 701B.055, 701B.060, 701B.070, 701B.080, 701B.090, 701B.100, 701B.110, 701B.120, 701B.130, 701B.140, 701B.150, 701B.160, 701B.170, 701B.180, 701B.200, 701B.210, 701B.220, 701B.230, 701B.240, 701B.250, 701B.255, **701B.260**, 701B.265, 701B.280 and 701B.290 are hereby repealed.
 - 2. [NRS 701B.260 is hereby repealed.
 - 3. NRS 701B.400, 701B.615 and 701B.700 are hereby repealed.
- 4.] Sections 1.53 and 19.8 of chapter 321, Statutes of Nevada 2009, at pages 1372 and 1408, respectively, are hereby repealed.
 - Sec. 53. (Deleted by amendment.)
- Sec. 54. 1. This section [49, 50 and 51 and] 1, 3 to 42, inclusive, 44, 45, 46, 48 to 51, inclusive, subsection [4] 2 of section 52 and section 53 of this act become effective upon passage and approval.
- 2. [Sections 13, 26, 45, and 46 of this act become effective on July 1, 2011.
- 3. Sections 1, 3 to 12, inclusive, 14 to 25, inclusive, 27 to 43, inclusive, 47, subsection 3 of section 52 and section 53 of this act become effective upon passage and approval for the purpose of adopting regulations and on January 1, 2013, for all other purposes.
- 4. Subsection 2 of section 52 of this act becomes effective on December 31, 2012.
- 5. Sections 3 to 6, inclusive, 15 to 25, inclusive, 27 to 43, inclusive, and 47 of this act expire by limitation on December 31, 2021.

6.] Sections 2, [44, 48] 43, 47 and subsection 1 of section 52 of this act become effective on January 1, 2022.

DEADLINES OF REPEALED SECTIONS OF NRS AND TEXT OF REPEALED SECTIONS OF STATUTES OF NEVADA

- 701B.010 Applicability.
- 701B.020 Definitions.
- 701B.030 "Applicant" defined.
- 701B.040 "Category" defined.
- 701B.050 "Commission" defined.
- 701B.055 "Distributed generation system" defined.
- 701B.060 "Institution of higher education" defined.
- 701B.070 "Owned, leased or occupied" defined.
- 701B.080 "Participant" defined.
- 701B.090 "Person" defined.
- 701B.100 "Program year" defined.
- 701B.110 "Public and other property" defined.
- 701B.120 "Public entity" defined.
- 701B.130 "School property" defined.
- 701B.140 "Small business" defined.
- 701B.150 "Solar energy system" defined.
- 701B.160 "Solar Program" defined.
- 701B.170 "Task Force" defined.
- 701B.180 "Utility" defined.
- 701B.200 Regulations: Establishment of incentives and requirements for utility's annual plan; exceptions; recovery of costs by utility.
- 701B.210 Regulations: Establishment of qualifications and requirements for participation; form and content of utility's master application.
 - 701B.220 Regulations: Establishment of incentives for participation.
- 701B.230 Duty of utility to file annual plan; review and approval of annual plan by Commission; recovery of costs by utility.
- 701B.240 Creation of Solar Program; categories of participation; eligibility requirements.
 - 701B.250 Application to participate; review of application by utility.
- 701B.255 Procedure for selection and notification of participants; authorization to install and energize solar energy system; submission of incentive claim form; determination of amount of incentive; withdrawal of participant; forfeiture of incentive.
- 701B.260 Capacity allocated to each category; reallocation of capacity; limitations on incentives.

701B.265 Installation of solar energy system deemed public work under certain circumstances.

701B.280 Participation in net metering.

701B.290 Issuance of portfolio energy credits.

[701B.400 Short title.

701B.615 Procedure for selection and notification of participants; authorization to install and energize wind energy system; submission of incentive claim form; determination of amount of incentive; withdrawal of participant; forfeiture of incentive.

701B.700 Short title.

Section 1.53 of chapter 321, Statutes of Nevada 2009, at pages 1372-73.

Sec. 1.53. NRS 701.180 is hereby amended to read as follows:

701.180 The Director shall:

- 1. Acquire and analyze information relating to energy and to the supply, demand and conservation of its sources, including, without limitation:
- (a) Information relating to the Solar Energy Systems Incentive Program created pursuant to NRS 701B.240 [and the Wind Energy Systems Demonstration Program created pursuant to 701B.580,] including, without limitation, information relating to:
- (1) The development of distributed generation systems in this State pursuant to participation in the Solar Energy Systems Incentive Program;
- (2) The use of carbon-based energy in residential and commercial applications due to participation in the [Programs;] Program; and
- (3) The average cost of generation on a kilowatt-hour basis for residential and commercial applications due to participation in the [Programs;] Program; and
- (b) Information relating to any money distributed pursuant to NRS 702.270.
- 2. Review and evaluate information which identifies trends and permits forecasting of the energy available to the State. Such forecasts must include estimates on:
- (a) The level of demand for energy in the State for 5-, 10- and 20-year periods;
 - (b) The amount of energy available to meet each level of demand;
- (c) The probable implications of the forecast on the demand and supply of energy; and
- (d) The sources of renewable energy and other alternative sources of energy which are available and their possible effects.
- 3. Study means of reducing wasteful, inefficient, unnecessary or uneconomical uses of energy and encourage the maximum utilization of existing sources of energy in the State.

- 4. Solicit and serve as the point of contact for grants and other money from the Federal Government, including, without limitation, any grants and other money available pursuant to any program administered by the United States Department of Energy, and other sources to cooperate with the Commissioner and the Authority:
- (a) To promote energy projects that enhance the economic development of the State:
 - (b) To promote the use of renewable energy in this State;
- (c) To promote the use of measures which conserve or reduce the demand for energy or which result in more efficient use of energy;
- (d) To develop a comprehensive program for retrofitting public buildings in this State with energy efficiency measures; and
- (e) If the Commissioner determines that it is feasible and cost-effective, to enter into contracts with researchers from the Nevada System of Higher Education for the design of energy efficiency and retrofit projects to carry out the comprehensive program for retrofitting public buildings in this State developed pursuant to paragraph (d).
- 5. Coordinate the activities and programs of the Office of Energy with the activities and programs of the Authority, the Consumer's Advocate and the Public Utilities Commission of Nevada, and with other federal, state and local officers and agencies that promote, fund, administer or operate activities and programs related to the use of renewable energy and the use of measures which conserve or reduce the demand for energy or which result in more efficient use of energy.
- 6. Carry out all other directives concerning energy that are prescribed by the Governor.

Section 19.8 of chapter 321, Statutes of Nevada 2009, at pages 1408-09.

- Sec. 19.8. Section 19.4 of this act is hereby amended to read as follows:
- Sec. 19.4. Chapter 338 of NRS is hereby amended by adding thereto a new section to read as follows:
- 1. The governing body of each local government shall, within 60 days after the effective date of this section, develop a plan to retrofit public buildings, facilities and structures, including, without limitation, traffic-control systems, and to otherwise use sources of renewable energy to serve those buildings, facilities and structures. Such a plan must:
- (a) Be developed with input from one or more energy retrofit coordinators designated pursuant to NRS 338.1907, if any.
- (b) Include a list of specific projects. The projects must be prioritized and selected on the basis of the following criteria:
 - (1) The length of time necessary to commence the project.
 - (2) The number of workers estimated to be employed on the project.

- (3) The effectiveness of the project in reducing energy consumption.
- (4) The estimated cost of the project.
- (5) Whether the project is able to be powered by or otherwise use sources of renewable energy.
- (6) Whether the project has qualified for participation in one or more of the following programs:
- (I) The Solar Energy Systems Incentive Program created by NRS 701B.240: or
- (II) The Renewable Energy School Pilot Program created by NRS 701B.350. $\mbox{\colored}$
- (III) The Wind Energy Systems Demonstration Program created by NRS 701B.580; or

(IV) The Waterpower Energy Systems Demonstration Program created by NRS 701B.820.]

- (c) Include a list of potential funding sources for use in implementing the projects, including, without limitation, money available through the Energy Efficiency and Conservation Block Grant Program as set forth in 42 U.S.C. § 17152 and grants, gifts, donations or other sources of money from public and private sources.
- 2. The governing body of each local government shall transmit the plan developed pursuant to subsection 1 to the Nevada Energy Commissioner and to any other entity designated for that purpose by the Legislature.
 - 3. As used in this section:
- (a) "Local government" means each city or county that meets the definition of "eligible unit of local government" as set forth in 42 U.S.C. § 17151 and each unit of local government, as defined in subsection 11 of NRS 338.010, that does not meet the definition of "eligible entity" as set forth in 42 U.S.C. § 17151.
- (b) "Renewable energy" means a source of energy that occurs naturally or is regenerated naturally, including, without limitation:
 - (1) Biomass;
 - (2) Fuel cells;
 - (3) Geothermal energy;
 - (4) Solar energy;
 - (5) Waterpower; and
 - (6) Wind.
- → The term does not include coal, natural gas, oil, propane or any other fossil fuel, or nuclear energy.
- (c) "Retrofit" means to alter, improve, modify, remodel or renovate a building, facility or structure to make that building, facility or structure more energy-efficient.

Assemblywoman Kirkpatrick moved the adoption of the amendment.

Amendment adopted.

Bill ordered reprinted, reengrossed, and to third reading.

Assembly Bill No. 503.

Bill read third time.

The following amendment was proposed by the Committee on Ways and Means:

Amendment No. 865.

AN ACT relating to wildlife; imposing certain conservation fees; [requiring] authorizing a person who accesses a wildlife management area and who is not the holder of an annual hunting, trapping, fishing or combined hunting and fishing license to pay an annual conservation fee.; [to access a wildlife management area;] revising certain provisions governing the use of money in the Wildlife Obligated Reserve Account; and providing other matters properly relating thereto.

Legislative Counsel's Digest:

Existing law provides that, in addition to any fee charged and collected for an annual hunting, trapping, fishing or combined hunting and fishing license, a \$3 habitat conservation fee must be paid. The proceeds from this fee must be deposited in the Wildlife Obligated Reserve Account and must be used for wildlife habitat rehabilitation and restoration. (NRS 502.242) Section 2 of this bill redesignates the habitat conservation fee as the conservation fee and sets the conservation fee at \$5 for residents and \$10 for nonresidents. In addition, section 2 [imposes] authorizes a person accessing a wildlife conservation area who is not the holder of a hunting, trapping, fishing or combined hunting and fishing license to pay an annual conservation fee of \$5 for residents and \$10 for nonresidents. For any person who wishes to access a wildlife management area but is not the holder of a hunting. trapping, fishing or combined hunting and fishing license.] Section 2 also provides that, each year, not more than 18 percent of the money credited to the Wildlife Obligated Reserve Account from any revenue received from those conservation fees may be used to monitor wildlife and its habitat for the purposes of wildlife habitat rehabilitation and restoration. Section 3 of this bill revises the authority of the Board of Wildlife Commissioners concerning the use of a wildlife management area by a person who pays the annual conservation fee.

THE PEOPLE OF THE STATE OF NEVADA, REPRESENTED IN SENATE AND ASSEMBLY, DO ENACT AS FOLLOWS:

Section 1. NRS 502.066 is hereby amended to read as follows:

- 502.066 1. The Department shall issue an apprentice hunting license to a person who:
 - (a) Is 12 years of age or older;
- (b) Has not previously been issued a hunting license by the Department, another state, an agency of a Canadian province or an agency of any other foreign country, including, without limitation, an apprentice hunting license; and
- (c) Except as otherwise provided in subsection 5, is otherwise qualified to obtain a hunting license in this State.
- 2. Except as otherwise provided in this subsection, the Department shall not impose a fee for the issuance of an apprentice hunting license. For each apprentice hunting license issued, the applicant or the mentor hunter for the applicant shall pay:
 - (a) Any service fee required by a license agent pursuant to NRS 502.040;
 - (b) The [habitat] conservation fee required by NRS 502.242; and
- (c) Any transaction fee that is set forth in a contract of this State with a third-party electronic services provider for each online transaction that is conducted with the Department.
- 3. An apprentice hunting license authorizes the apprentice hunter to hunt in this State as provided in this section.
- 4. It is unlawful for an apprentice hunter to hunt in this State unless a mentor hunter accompanies and directly supervises the apprentice hunter at all times during a hunt. During the hunt, the mentor hunter shall ensure that:
- (a) The apprentice hunter safely handles and operates the firearm or weapon used by the apprentice hunter; and
- (b) The apprentice hunter complies with all applicable laws and regulations concerning hunting and the use of firearms.
- 5. A person is not required to complete a course of instruction in the responsibilities of hunters as provided in NRS 502.340 to obtain an apprentice hunting license.
 - 6. The issuance of an apprentice hunting license does not:
 - (a) Authorize the apprentice hunter to obtain any other hunting license;
- (b) Authorize the apprentice hunter to hunt any animal for which a tag is required pursuant to NRS 502.130; or
 - (c) Exempt the apprentice hunter from any requirement of this title.
- 7. The Commission may adopt regulations to carry out the provisions of this section.
 - 8. As used in this section:
- (a) "Accompanies and directly supervises" means maintains close visual and verbal contact with, provides adequate direction to and maintains the ability readily to assume control of any firearm or weapon from an apprentice hunter.

- (b) "Apprentice hunter" means a person who obtains an apprentice hunting license pursuant to this section.
- (c) "Mentor hunter" means a person 18 years of age or older who holds a hunting license issued in this State and who accompanies and directly supervises an apprentice hunter. The term does not include a person who holds an apprentice hunting license pursuant to this section.
 - **Sec. 2.** NRS 502.242 is hereby amended to read as follows:
- 502.242 1. In addition to any fee charged and collected for an annual hunting, trapping, fishing or combined hunting and fishing license pursuant to NRS 502.240, a [habitat] conservation fee [of \$3] must be paid [...] in the amount of \$5 for a resident and \$10 for a nonresident.
- 2. [In order to access a wildlife management area, a] A person accessing a wildlife management area who is not the holder of an annual hunting, trapping, fishing or combined hunting and fishing license [must] may pay an annual conservation fee in the amount of \$5 for a resident and \$10 for a nonresident.
- 3. The Wildlife Obligated Reserve Account is hereby created in the State General Fund. Revenue from the [habitat] conservation fee must be accounted for separately, deposited with the State Treasurer for credit to the Wildlife Obligated Reserve Account and, except as otherwise provided in this subsection and NRS 502.294 and 502.310, used by the Department for the purposes of wildlife habitat rehabilitation and restoration. Each year, not more than 18 percent of the money credited to the Wildlife Obligated Reserve Account from any revenue received pursuant to subsections 1 and 2 may be used to monitor wildlife and its habitat for those purposes. The interest and income earned on the money in the Wildlife Obligated Reserve Account, after deducting any applicable charges, must be credited to the Account.
- [3.] 4. The money in the Wildlife Obligated Reserve Account remains in the Account and does not revert to the State General Fund at the end of any fiscal year.
 - **Sec. 3.** NRS 504.143 is hereby amended to read as follows:
- 504.143 1. To effectuate a coordinated and balanced program resulting in the maximum revival of wildlife in the State and in the maximum recreational advantages to the people of the State, the Commission has created and maintains state-owned wildlife management areas, and, in cooperation with the United States Fish and Wildlife Service, the Department of Interior and other federal agencies, has created and maintains other cooperative wildlife management areas.
- 2. <u>The [Except as otherwise provided in NRS 502.242, the]</u> Commission may permit hunting, fishing or trapping on or within, or access to, occupancy and use of, areas so created and maintained.

- 3. The Commission may by regulation:
- (a) Establish, extend, shorten or abolish open seasons and closed seasons within such areas.
- (b) Establish, change or abolish bag and creel limits and possession limits in such areas.
 - (c) Prescribe the manner and the means of taking wildlife in such areas.
- (d) Establish, change or abolish restrictions in such areas based upon sex, maturity or other physical distinctions.
- (e) Prescribe the manner of using such areas for a person who pays the annual fee [to access such areas] pursuant to NRS 502.242.
 - **Sec. 4.** This act becomes effective on July 1, 2011.

Assemblyman Hickey moved the adoption of the amendment.

Amendment adopted.

Bill ordered reprinted, reengrossed, and to third reading.

Senate Bill No. 60.

Bill read third time.

Roll call on Senate Bill No. 60:

YEAS—42.

NAYS-None.

Senate Bill No. 60 having received a constitutional majority, Mr. Speaker declared it passed.

Bill ordered transmitted to the Senate.

Senate Bill No. 443.

Bill read third time.

Roll call on Senate Bill No. 443:

YEAS—39.

NAYS—Ellison, Grady, McArthur—3.

Senate Bill No. 443 having received a constitutional majority, Mr. Speaker declared it passed.

Bill ordered transmitted to the Senate.

Senate Bill No. 446.

Bill read third time.

Remarks by Assemblywoman Smith.

Roll call on Senate Bill No. 446:

YEAS-41

NAYS-McArthur.

Senate Bill No. 446 having received a constitutional majority, Mr. Speaker declared it passed.

Bill ordered transmitted to the Senate.

UNFINISHED BUSINESS

CONSIDERATION OF SENATE AMENDMENTS

Assembly Bill No. 337.

The following Senate amendment was read:

Amendment No. 689.

AN ACT relating to campaign practices; requiring a notice of an alleged violation of provisions governing campaign practices to include certain information; requiring the Secretary of State to provide a copy of the notice and any accompanying information to the person alleged in the notice to have committed the violation; authorizing the person to respond to such a notice; authorizing the Secretary of State to conduct an investigation based on such a notice in certain circumstances; authorizing the Secretary of State or a designated officer or employee of the Secretary of State to subpoena witnesses and require the production of documents or records by subpoena when conducting an investigation based on such a notice in certain circumstances; and providing other matters properly relating thereto.

Legislative Counsel's Digest:

Existing law confers authority upon the Secretary of State to conduct investigations concerning alleged violations of chapter 294A of NRS governing campaign practices. Existing law also authorizes a person who believes that any provision of that chapter has been violated to notify the Secretary of State in writing. The notice must be signed by the person and include any information in support of the alleged violation. (NRS 294A.410) This bill specifies the information that must be included in the notice and requires the Secretary of State to provide a copy of the notice and any accompanying information to the person, if any, alleged in the notice to have committed the violation. If, based on such a notice, the Secretary of State determines that reasonable suspicion exists that a violation has occurred, the Secretary of State is authorized to investigate the allegation. This bill further provides that, if the notice is received within 180 days after the general election, general city election or special election for the office or ballot question to which the notice pertains, the Secretary of State is authorized, when conducting an investigation based on the notice, to subpoena witnesses and require the production by subpoena of any books, papers, correspondence, memoranda, agreements or other documents or records in the possession of [the] any person: (1) alleged in the notice to have committed the violation [that]; or (2) who the Secretary of State or a designated officer or employee of the Secretary of State has reasonable cause to believe produced or disseminated the materials that are the subject of the notice, if the Secretary of State or a designated officer or employee of the Secretary of State determines that the documents or <u>records</u> are relevant or material to the investigation. Finally, this bill authorizes the Secretary of State or a designated officer or employee of the Secretary of State to apply to a court for an order compelling compliance if a person fails to testify or produce the required documents or records.

THE PEOPLE OF THE STATE OF NEVADA, REPRESENTED IN SENATE AND ASSEMBLY, DO ENACT AS FOLLOWS:

Section 1. NRS 294A.410 is hereby amended to read as follows:

- 294A.410 1. If it appears that the provisions of this chapter have been violated, the Secretary of State may:
- (a) Conduct an investigation concerning the alleged violation and cause the appropriate proceedings to be instituted and prosecuted in the First Judicial District Court; or
- (b) Refer the alleged violation to the Attorney General. The Attorney General shall investigate the alleged violation and institute and prosecute the appropriate proceedings in the First Judicial District Court without delay.
- 2. A person who believes that any provision of this chapter has been violated may notify the Secretary of State, in writing, of the alleged violation. The notice must be signed by the person alleging the violation and include [any]:
 - (a) The full name and address of the person alleging the violation;
- (b) A clear and concise statement of facts sufficient to establish that the alleged violation occurred;
 - (c) Any evidence substantiating the alleged violation;
- (d) A certification by the person alleging the violation that the facts alleged in the notice are true to the best knowledge and belief of that person; and
 - (e) Any other information in support of the alleged violation.
- 3. As soon as practicable after receiving a notice of an alleged violation pursuant to subsection 2, the Secretary of State shall provide a copy of the notice and any accompanying information to the person <u>if any</u> alleged in the notice to have committed the violation. Any response submitted to the notice must be accompanied by a short statement of the grounds, if any, for objecting to the alleged violation and include any evidence substantiating the objection.
- 4. If the Secretary of State determines, based on a notice of an alleged violation received pursuant to subsection 2, that reasonable suspicion exists that a violation of this chapter has occurred, the Secretary of State may conduct an investigation of the alleged violation.
- 5. If a notice of an alleged violation is received pursuant to subsection 2 not later than 180 days after the general election, general city election or special election for the office or ballot question to which the notice

pertains, the Secretary of State, when conducting an investigation of the alleged violation pursuant to subsection 4, may subpoena witnesses and require the production by subpoena of any books, papers, correspondence, memoranda, agreements or other documents or records [in the possession of the person alleged in the notice to have committed the violation] that the Secretary of State or a designated officer or employee of the Secretary of State determines are relevant or material to the investigation [...] and are in the possession of:

- (a) Any person alleged in the notice to have committed the violation; or
- (b) If the notice does not include the name of a person alleged to have committed the violation, any person who the Secretary of State or a designated officer or employee of the Secretary of State has reasonable cause to believe produced or disseminated the materials that are the subject of the notice.
- 6. If a person fails to testify or produce any documents or records in accordance with a subpoena issued pursuant to subsection 5, the Secretary of State or designated officer or employee may apply to the court for an order compelling compliance. A request for an order of compliance may be addressed to:
- (a) The district court in and for the county where service may be obtained on the person refusing to testify or produce the documents or records, if the person is subject to service of process in this State; or
- (b) A court of another state having jurisdiction over the person refusing to testify or produce the documents or records, if the person is not subject to service of process in this State.

Assemblyman Segerblom moved that the Assembly concur in the Senate Amendment No. 689 to Assembly Bill No. 337.

Remarks by Assemblyman Segerblom.

Motion carried by a constitutional majority.

Bill ordered enrolled.

Assembly Bill No. 82.

The following Senate amendment was read:

Amendment No. 567.

AN ACT relating to elections; revising provisions governing registering to vote by computer; authorizing additional mailing precincts in certain circumstances; making various other changes relating to the administration and conduct of elections; **prohibiting foreign nationals from making campaign contributions; prohibiting certain persons from receiving such contributions;** authorizing the disposition of unspent campaign contributions to a governmental entity and for the use of legal expenses; requiring the annual registration of committees for political action; making various other

changes relating to campaign finance; providing penalties; and providing other matters properly relating thereto.

Legislative Counsel's Digest:

Section 7 of this bill allows a county clerk to establish mailing precincts or absent ballot mailing precincts if approved by the Secretary of State, in addition to circumstances authorized for the creation of mailing precincts in existing law. (NRS 293.213)

Sections 8 and 23 of this bill provide that if a county clerk establishes a system for using a computer for voter registration for that county, the system established must comply with any procedures and requirements prescribed by the Secretary of State. Existing law requires county clerks to verify the validity of the signatures of persons who sign petitions for initiative or referendum, petitions to recall public officers, petitions to qualify as a political party, petitions for filling ballot vacancies or petitions to place minor party or independent candidates on the ballot against the voter registration records, including applications to register to vote. (NRS 293.1277) Section 5 of this bill provides that if a computer is used for voter registration in a county, the county clerk may rely on such indicia as may be prescribed by the Secretary of State to complete the signature verification.

Sections 9, 26 and 36 of this bill revise the manner in which it is required to list on sample ballots and ballots the names of candidates who have the same names so that if two or more candidates in an election have the same given name and surname and one candidate is an incumbent, the word "Incumbent" must appear on the sample ballot and ballot next to the name of the candidate who is the incumbent.

Section 12 of this bill requires, in addition to other information posted at polling places on election day, the posting of information concerning the eligibility of a candidate, question or other matter to appear on the ballot as a result of judicial determination or by operation of law.

Sections 15 and 35 of this bill authorize voters to vote in mailing precincts if it appears to the satisfaction of the Secretary of State, in addition to the county clerk, that the circumstances authorizing the creation of a mailing precinct exist.

Existing law authorizes a city or county clerk to assess a charge, not to exceed the cost of printing the applications, against a political party or other entity that requests more than 50 applications to register to vote by mail in any 12-month period. (NRS 293.443) **Section 16** of this bill authorizes the Secretary of State to assess such a charge as well.

Section 19 of this bill changes the deadline for the Secretary of State to submit a report concerning primary and general elections to the Legislature from not later than 30 days before the start of a regular legislative session to

not sooner than 30 days before and not later than 30 days after the first day of each regular legislative session.

Section 20 of this bill requires recruitment offices of the Armed Forces of the United States to serve as voter registration agencies, in addition to other entities specified in existing law.

Section 21 of this bill prohibits a voter registration agency from knowingly employing a person whose duties will include the registration of voters if the person has been convicted of a felony involving theft or fraud.

Section 22 of this bill prohibits a county clerk from knowingly appointing as a field registrar any person who has been convicted of a felony involving theft or fraud.

Section 27 of this bill amends the deadlines for the county clerk to transmit the number of registered voters in the county to the Secretary of State for the primary and general elections.

Section 28 of this bill expands the crime of threatening a person in connection with an election or petition to include threatening a person in connection with the registration of voters and to include the use of or threatening to use intimidation. **Section 28** also increases the penalty for such a crime from a gross misdemeanor to a category E felony.

Section 29 of this bill increases the penalty for interfering with the conduct of an election from a gross misdemeanor to a category E felony.

Section 30 of this bill provides that polling information from a voter regarding whether the voter intends to vote for or against a particular political party, candidate or ballot question is not "electioneering."

Section 32 of this bill provides that if a person tampers or interferes with, or attempts to tamper or interfere with, a mechanical voting system, mechanical voting device or any computer program used to count ballots, such an act is punishable as a category B felony.

Section 33 of this bill makes the intentional failure to submit to the county clerk an elector's completed application to register to vote by a person who provided the application to the elector an unlawful act punishable as a category E felony.

The Federal Election Campaign Act prohibits: (1) a foreign national from directly or indirectly making a campaign contribution in connection with a state or local election; and (2) any person from knowingly soliciting, accepting or receiving any campaign contribution from a foreign national. (2 U.S.C. § 441e; 11 C.F.R. § 110.20) Section 40.5 of this bill provides a similar prohibition in state law against a foreign national making a contribution to certain persons or groups, including candidates, committees for political action, persons who make independent expenditures, political parties and legislative caucuses.

Section 40.5 also prohibits those persons and groups from receiving a contribution from a foreign national.

Existing law prohibits a person from making certain campaign contributions over \$5,000 during certain periods and prohibits candidates from accepting such contributions during those periods. (NRS 294A.100, 294A.287) **Sections 43 and 60** of this bill also prohibit a person from committing to make such a contribution. **Section 61** of this bill similarly adds the prohibition on committing to make such a contribution to the prohibition on soliciting and accepting any monetary contribution for any political purpose during a specified period which is applicable to Legislators, the Lieutenant Governor, the Lieutenant Governor-Elect, the Governor and the Governor-Elect.

Section 50 of this bill adds to the acceptable methods of disposing of unspent campaign contributions donating money to a governmental entity and allows the person disposing of the unspent contributions to specify how the governmental entity may use the money. **Section 50** also allows certain public officers to use unspent campaign contributions in a future election in certain circumstances.

Section 56 of this bill requires committees for political action to file with the Secretary of State an updated form of registration on or before January 15 of each year.

Section 59 of this bill sets forth the acceptable methods of and deadline for disposing of unspent money in a legal defense fund. **Sections 51 and 63** of this bill require a person who disposes of unspent money in a legal defense fund to report to the Secretary of State how the person disposed of such money.

Sections 48, 54 and 84 of this bill remove requirements that certain persons or groups who advocate the passage or defeat of a ballot question register. As a result, these persons and groups are subject to the same registration [and reporting] requirements as committees for political action. [and must] Under existing law, such persons and groups are not required to report [certain] contributions and expenditures [which exceed \$100, rather than \$10,000 as provided in existing law for] until they have received or expended money in excess of \$10,000 to advocate the passage or defeat of a ballot question. (NRS 294A.150, 294A.220) Sections 48 and 54 eliminate this threshold and therefore require these persons or groups [1] to report contributions received and expenditures made in excess of \$1,000 during any reporting period.

THE PEOPLE OF THE STATE OF NEVADA, REPRESENTED IN SENATE AND ASSEMBLY, DO ENACT AS FOLLOWS:

Section 1. (Deleted by amendment.)

- **Sec. 2.** (Deleted by amendment.)
- **Sec. 3.** (Deleted by amendment.)
- **Sec. 4.** (Deleted by amendment.)
- **Sec. 5.** NRS 293.1277 is hereby amended to read as follows:
- 293.1277 1. If the Secretary of State finds that the total number of signatures submitted to all the county clerks is 100 percent or more of the number of registered voters needed to declare the petition sufficient, the Secretary of State shall immediately so notify the county clerks. Within 9 days, excluding Saturdays, Sundays and holidays, after notification, each of the county clerks shall determine the number of registered voters who have signed the documents submitted in the county clerk's county and, in the case of a petition proposing a statute, an amendment to a statute or an amendment to the Constitution, shall tally the number of signatures for each petition district contained or fully contained within the county clerk's county.
- 2. If more than 500 names have been signed on the documents submitted to a county clerk, the county clerk shall examine the signatures by sampling them at random for verification. The random sample of signatures to be verified must be drawn in such a manner that every signature which has been submitted to the county clerk is given an equal opportunity to be included in the sample. The sample must include an examination of at least 500 or 5 percent of the signatures, whichever is greater.
- 3. In determining from the records of registration the number of registered voters who signed the documents, the county clerk may use the signatures contained in the file of applications to register to vote. If the county clerk uses that file, the county clerk shall ensure that every application in the file is examined, including any application in his or her possession which may not yet be entered into the county clerk's records. [The] Except as otherwise provided in this subsection, the county clerk shall rely only on the appearance of the signature and the address and date included with each signature in making his or her determination. If, pursuant to NRS 293.506, a county clerk establishes a system to allow persons to register to vote by computer, the county clerk may rely on such other indicia as prescribed by the Secretary of State in making his or her determination.
- 4. In the case of a petition proposing a statute, an amendment to a statute or an amendment to the Constitution, when the county clerk is determining the number of registered voters who signed the documents from each petition district contained fully or partially within the county clerk's county, he or she must use the statewide voter registration list available pursuant to NRS 293.675.
- 5. Except as otherwise provided in subsection 7, upon completing the examination, the county clerk shall immediately attach to the documents a

certificate properly dated, showing the result of the examination, including the tally of signatures by petition district, if required, and transmit the documents with the certificate to the Secretary of State. If a petition district comprises more than one county and the petition proposes a statute, an amendment to a statute or an amendment to the Constitution, the appropriate county clerks shall comply with the regulations adopted by the Secretary of State pursuant to this section to complete the certificate. A copy of this certificate must be filed in the clerk's office. When the county clerk transmits the certificate to the Secretary of State, the county clerk shall notify the Secretary of State of the number of requests to remove a name received by the county clerk pursuant to NRS 295.055 or 306.015.

- 6. A person who submits a petition to the county clerk which is required to be verified pursuant to NRS 293.128, 293.165, 293.172, 293.200, 295.056, 298.109, 306.035 or 306.110 must be allowed to witness the verification of the signatures. A public officer who is the subject of a recall petition must also be allowed to witness the verification of the signatures on the petition.
- 7. For any petition containing signatures which are required to be verified pursuant to the provisions of NRS 293.165, 293.200, 306.035 or 306.110 for any county, district or municipal office within one county, the county clerk shall not transmit to the Secretary of State the documents containing the signatures of the registered voters.
- 8. The Secretary of State shall by regulation establish further procedures for carrying out the provisions of this section.
 - **Sec. 6.** (Deleted by amendment.)
 - **Sec. 7.** NRS 293.213 is hereby amended to read as follows:
- 293.213 1. Whenever there were not more than 20 voters registered in a precinct for the last preceding general election, the county clerk may establish that precinct as a mailing precinct.
- 2. Except as otherwise provided in NRS 293.208, the county clerk in any county [where] in which an absent ballot central counting board is appointed may abolish two or more existing mailing precincts and combine those mailing precincts into absent ballot precincts. Those mailing precincts must be designated absent ballot mailing precincts.
- 3. In any county [where] in which an absent ballot central counting board is appointed, any established precinct which had less than 200 ballots cast at the last preceding general election, or any newly established precinct with less than 200 registered voters, may be designated an absent ballot mailing precinct.
- 4. A county clerk may establish a mailing precinct or an absent ballot mailing precinct that does not meet the requirements of subsection 1, 2 or 3 if the county clerk obtains prior approval from the Secretary of State.

- 5. The county clerk shall, at least 14 days before establishing or designating a precinct as a mailing precinct or absent ballot mailing precinct or before abolishing a mailing precinct pursuant to this section, cause notice of such action to be:
- (a) Posted in the manner prescribed for a regular meeting of the board of county commissioners; and
- (b) Mailed to each Assemblyman, Assemblywoman, State Senator, county commissioner and, if applicable, member of the governing body of a city who represents residents of a precinct affected by the action.
 - **Sec. 8.** NRS 293.250 is hereby amended to read as follows:
- 293.250 1. The Secretary of State shall, in a manner consistent with the election laws of this State, prescribe:
- (a) The form of all ballots, absent ballots, diagrams, sample ballots, certificates, notices, declarations, applications to register to vote, lists, applications, registers, rosters, statements and abstracts required by the election laws of this State.
- (b) The [procedure] procedures to be followed [when a computer is used] and the requirements of a system established pursuant to NRS 293.506 for using a computer to register voters and to keep records of registration.
- 2. The Secretary of State shall prescribe with respect to the matter to be printed on every kind of ballot:
- (a) The placement and listing of all offices, candidates and measures upon which voting is statewide, which must be uniform throughout the State.
- (b) The listing of all other candidates required to file with the Secretary of State, and the order of listing all offices, candidates and measures upon which voting is not statewide, from which each county or city clerk shall prepare appropriate ballot forms for use in any election in his or her county.
- 3. The Secretary of State shall place the condensation of each proposed constitutional amendment or statewide measure near the spaces or devices for indicating the voter's choice.
- 4. The fiscal note for, explanation of, arguments for and against, and rebuttals to such arguments of each proposed constitutional amendment or statewide measure must be included on all sample ballots.
- 5. The condensations and explanations for constitutional amendments and statewide measures proposed by initiative or referendum must be prepared by the Secretary of State, upon consultation with the Attorney General. The arguments and rebuttals for or against constitutional amendments and statewide measures proposed by initiative or referendum must be prepared in the manner set forth in NRS 293.252. The fiscal notes for constitutional amendments and statewide measures proposed by initiative or referendum must be prepared by the Secretary of State, upon consultation with the Fiscal Analysis Division of the Legislative Counsel Bureau. The

condensations, explanations, arguments, rebuttals and fiscal notes must be in easily understood language and of reasonable length, and whenever feasible must be completed by August 1 of the year in which the general election is to be held.

- 6. The names of candidates for township and legislative or special district offices must be printed only on the ballots furnished to voters of that township or district.
 - 7. A county clerk:
- (a) May divide paper ballots into two sheets in a manner which provides a clear understanding and grouping of all measures and candidates.
- (b) Shall prescribe the color or colors of the ballots and voting receipts used in any election which the clerk is required to conduct.
 - **Sec. 9.** NRS 293.2565 is hereby amended to read as follows:
- 293.2565 1. Except as otherwise provided in subsection 2, in any election regulated by this chapter, the name of a candidate printed on a ballot may be the given name and surname of the candidate or a contraction or familiar form of his or her given name followed by his or her surname. A nickname of not more than 10 letters may be incorporated into the name of a candidate. The nickname must be in quotation marks and appear immediately before the surname of the candidate. A nickname must not indicate any political, economic, social or religious view or affiliation and must not be the name of any person, living or dead, whose reputation is known on a statewide, nationwide or worldwide basis, or in any other manner deceive a voter regarding the person or principles for which he or she is voting.
- 2. [Except as otherwise provided in subsection 3, in] In any election regulated by this chapter, if two or more candidates have the same given name and surname [or surnames so similar as to be likely to cause confusion] and:
- (a) None of the candidates is an incumbent, the middle names or middle initials, if any, of the candidates must be included in the names of the candidates; [as printed on the ballot;] or
- (b) One of the candidates is an incumbent, the name of the incumbent must be listed first and [must be printed in bold type.
- 3. Where a system of voting other than by paper ballot is used and the provisions of paragraph (b) of subsection 2 are applicable, the Secretary of State may distinguish a candidate who is an incumbent in a manner other than printing the name of the incumbent in bold type provided that the manner used clearly emphasizes the name of the incumbent in a manner similar to printing his or her name in bold type.] the word "Incumbent" must appear next to the name of the candidate who is the incumbent.
 - **Sec. 10.** NRS 293.272 is hereby amended to read as follows:

- 293.272 1. Except as otherwise provided in subsection 2 and in NRS 293.2725 and 293.3083, a person who registered *by mail or computer* to vote [pursuant to the provisions of NRS 293.5235] shall, for the first election in which the person votes at which that registration is valid, vote in person unless he or she has previously voted in the county in which he or she is registered to vote.
 - 2. The provisions of subsection 1 do not apply to a person who:
- (a) Is entitled to vote in the manner prescribed in NRS 293.343 to 293.355, inclusive:
- (b) Is entitled to vote an absent ballot pursuant to federal law or NRS 293.316 or 293.3165;
 - (c) Is disabled;
- (d) Submits or has previously submitted a written request for an absent ballot that is signed by the registered voter before a notary public or other person authorized to administer an oath; or
 - (e) Requests an absent ballot in person at the office of the county clerk.
 - **Sec. 11.** NRS 293.2725 is hereby amended to read as follows:
- 293.2725 1. Except as otherwise provided in subsection 2, in NRS 293.3081 and 293.3083 and in federal law, a person who registers by mail *or computer* to vote in this State and who has not previously voted in an election for federal office in this State:
- (a) May vote at a polling place only if the person presents to the election board officer at the polling place:
 - (1) A current and valid photo identification of the person; or
- (2) A copy of a current utility bill, bank statement, paycheck, or document issued by a governmental entity, including a check which indicates the name and address of the person, but not including a voter registration card issued pursuant to NRS 293.517; and
- (b) May vote by mail only if the person provides to the county or city clerk:
 - (1) A copy of a current and valid photo identification of the person; or
- (2) A copy of a current utility bill, bank statement, paycheck, or document issued by a governmental entity, including a check which indicates the name and address of the person, but not including a voter registration card issued pursuant to NRS 293.517.
 - 2. The provisions of this section do not apply to a person who:
- (a) Registers to vote by mail and submits with an application to register to vote:
 - (1) A copy of a current and valid photo identification; or
- (2) A copy of a current utility bill, bank statement, paycheck, or document issued by a governmental entity, including a check which indicates

the name and address of the person, but not including a voter registration card issued pursuant to NRS 293.517;

- (b) Registers to vote by mail and submits with an application to register to vote a driver's license number or at least the last four digits of his or her social security number, if a state or local election official has matched that information with an existing identification record bearing the same number, name and date of birth as provided by the person in the application;
- (c) Is entitled to vote an absent ballot pursuant to the Uniformed and Overseas Citizens Absentee Voting Act, 42 U.S.C. §§ 1973ff et seq.;
- (d) Is provided the right to vote otherwise than in person under the Voting Accessibility for the Elderly and Handicapped Act, 42 U.S.C. §§ 1973ee et seq.; or
 - (e) Is entitled to vote otherwise than in person under any other federal law.
 - **Sec. 12.** NRS 293.3025 is hereby amended to read as follows:
- 293.3025 The Secretary of State and each county and city clerk shall ensure that a copy of each of the following is posted in a conspicuous place at each polling place on election day:
 - 1. A sample ballot;
- 2. Information concerning the date and hours of operation of the polling place;
- 3. Instructions for voting and casting a ballot, including a provisional ballot;
- 4. Instructions concerning the identification required for persons who registered by mail and are first-time voters for federal office in this State;
- 5. Information concerning the accessibility of polling places to persons with disabilities: [and]
- 6. General information concerning federal and state laws which prohibit acts of fraud and misrepresentation \Box ; and
- 7. Information concerning the eligibility of a candidate, a ballot question or any other matter appearing on the ballot as a result of a judicial determination or by operation of law, if any.
 - **Sec. 13.** NRS 293.3081 is hereby amended to read as follows:
- 293.3081 A person at a polling place may cast a provisional ballot in an election to vote for a candidate for federal office if the person complies with the applicable provisions of NRS 293.3082 and:
- 1. Declares that he or she has registered to vote and is eligible to vote at that election in that jurisdiction, but his or her name does not appear on a voter registration list as a voter eligible to vote in that election in that jurisdiction or an election official asserts that the person is not eligible to vote in that election in that jurisdiction;
- 2. Applies by mail [,] or computer, on or after January 1, 2003, to register to vote and has not previously voted in an election for federal office

in this State and fails to provide the identification required pursuant to paragraph (a) of subsection 1 of NRS 293.2725 to the election board officer at the polling place; or

- 3. Declares that he or she is entitled to vote after the polling place would normally close as a result of a court order or other order extending the time established for the closing of polls pursuant to a law of this State in effect 10 days before the date of the election.
 - **Sec. 14.** NRS 293.3083 is hereby amended to read as follows:
- 293.3083 A person may cast a ballot by mail to vote for a candidate for federal office, which must be treated as a provisional ballot by the county or city clerk if the person:
- 1. Applies by mail *or computer* to register to vote and has not previously voted in an election for federal office in this State;
- 2. Fails to provide the identification required pursuant to paragraph (b) of subsection 1 of NRS 293.2725 to the county or city clerk at the time that the person mails the ballot; and
- 3. Completes the written affirmation set forth in subsection 1 of NRS 293.3082.
 - **Sec. 15.** NRS 293.343 is hereby amended to read as follows:
- 293.343 1. A registered voter who resides in an election precinct in which there were not more than 200 voters registered for the last preceding general election, or in a precinct in which it appears to the satisfaction of the county clerk *and Secretary of State* that there are not more than 200 registered voters, may vote at any election regulated by this chapter in the manner provided in NRS 293.345 to 293.355, inclusive.
- 2. Whenever the county clerk has designated a precinct as a mailing precinct, registered voters residing in that precinct may vote at any election regulated by this chapter in the manner provided in NRS 293.345 to 293.355, inclusive.
- 3. In a county whose population is 100,000 or more, whenever a registered voter is entitled to vote in a mailing precinct or an absent ballot mailing precinct, the county clerk:
- (a) Shall designate at least one polling place in the county as the polling place where such a voter may vote in person, pursuant to paragraph (b) of subsection 2 of NRS 293.353 or subsection 3 of NRS 293.353, on election day; and
- (b) May designate certain polling places for early voting as the polling places where such a voter may vote in person, pursuant to paragraph (b) of subsection 2 of NRS 293.353 or subsection 3 of NRS 293.353, during the period for early voting, if it is impractical for the county clerk to provide at each polling place for early voting a ballot in every form required in the county.

- 4. In a county whose population is less than 100,000, whenever a registered voter is entitled to vote in a mailing precinct or an absent ballot mailing precinct, the county clerk:
- (a) May designate one or more polling places in the county as the polling place where such a voter may vote in person, pursuant to paragraph (b) of subsection 2 of NRS 293.353 or subsection 3 of NRS 293.353, on election day; and
- (b) May designate certain polling places for early voting as the polling places where such a voter may vote in person, pursuant to paragraph (b) of subsection 2 of NRS 293.353 or subsection 3 of NRS 293.353, during the period for early voting, if it is impractical for the county clerk to provide at each polling place for early voting a ballot in every form required in the county.
- 5. Polling places designated pursuant to subsection 3 or 4 may include, without limitation, polling places located as closely as practicable to the mailing precincts.
 - **Sec. 16.** NRS 293.443 is hereby amended to read as follows:
- 293.443 1. Except as otherwise provided in subsection 3, the expense of providing all ballots, forms and other supplies to be used at any election regulated by this chapter or chapter 293C of NRS and all expenses necessarily incurred in the preparation for, or the conduct of, any such election is a charge upon the municipality, county, district or State, as the case may be.
- 2. The county or city clerk may submit the printing of ballots for competitive bidding.
- 3. If a political party or other entity requests more than 50 applications to register to vote by mail in any 12-month period, the clerk *or the Secretary of State* may assess a charge, not to exceed the cost of printing the applications.
 - **Sec. 17.** NRS 293.4687 is hereby amended to read as follows:
- 293.4687 1. The Secretary of State shall maintain a website on the Internet for public information maintained, collected or compiled by the Secretary of State that relates to elections, which must include, without limitation:
- (a) The Voters' Bill of Rights required to be posted on the Secretary of State's Internet website pursuant to the provisions of NRS 293.2549;
- (b) The abstract of votes required to be posted on a website pursuant to the provisions of NRS 293.388;
- (c) A current list of the registered voters in this State that also indicates the petition district in which each registered voter resides;
 - (d) A map or maps indicating the boundaries of each petition district; and
- (e) All reports on campaign contributions and expenditures submitted to the Secretary of State pursuant to the provisions of NRS 294A.120,

- 294A.125, 294A.140, 294A.150, 294A.200, 294A.210, 294A.220, 294A.270, 294A.280, [294A.283,] 294A.360 and 294A.362 and all reports on contributions received by and expenditures made from a legal defense fund submitted to the Secretary of State pursuant to NRS 294A.286.
- 2. The abstract of votes required to be maintained on the website pursuant to paragraph (b) of subsection 1 must be maintained in such a format as to permit the searching of the abstract of votes for specific information.
- 3. If the information required to be maintained by the Secretary of State pursuant to subsection 1 may be obtained by the public from a website on the Internet maintained by a county clerk or city clerk, the Secretary of State may provide a hyperlink to that website to comply with the provisions of subsection 1 with regard to that information.
 - **Sec. 18.** (Deleted by amendment.)
 - **Sec. 19.** NRS 293.4695 is hereby amended to read as follows:
- 293.4695 1. Each county clerk shall collect the following information regarding each primary and general election, on a form provided by the Secretary of State and made available at each polling place in the county, each polling place for early voting in the county, the office of the county clerk and any other location deemed appropriate by the Secretary of State:
- (a) The number of ballots that have been discarded or for any reason not included in the final canvass of votes, along with an explanation for the exclusion of each such ballot from the final canvass of votes.
- (b) A report on each malfunction of any mechanical voting system, including, without limitation:
 - (1) Any known reason for the malfunction;
- (2) The length of time during which the mechanical voting system could not be used;
- (3) Any remedy for the malfunction which was used at the time of the malfunction; and
 - (4) Any effect the malfunction had on the election process.
- (c) A list of each polling place not open during the time prescribed pursuant to NRS 293.273 and an account explaining why each such polling place was not open during the time prescribed pursuant to NRS 293.273.
- (d) A description of each challenge made to the eligibility of a voter pursuant to NRS 293.303 and the result of each such challenge.
- (e) A description of each complaint regarding a ballot cast by mail or facsimile filed with the county clerk and the resolution, if any, of the complaint.
- (f) The results of any audit of election procedures and practices conducted pursuant to regulations adopted by the Secretary of State pursuant to this chapter.

- (g) The number of provisional ballots cast and the reason for the casting of each provisional ballot.
- 2. Each county clerk shall submit to the Secretary of State, on a form provided by the Secretary of State, the information collected pursuant to subsection 1 not more than 60 days after each primary and general election.
- 3. The Secretary of State may contact any political party and request information to assist in the investigation of any allegation of voter intimidation.
- 4. The Secretary of State shall establish and maintain an Internet website pursuant to which the Secretary of State shall solicit and collect voter comments regarding election processes.
- 5. The Secretary of State shall compile the information and comments collected pursuant to this section into a report and shall submit the report to the Director of the Legislative Counsel Bureau for transmission to the Legislature [not later than] not sooner than 30 days before and not later than 30 days after the [start] first day of each regular session of the Legislature.
- 6. The Secretary of State may make the report required pursuant to subsection 5 available on an Internet website established and maintained by the Secretary of State.
 - **Sec. 20.** NRS 293.504 is hereby amended to read as follows:
- 293.504 1. The following offices shall serve as voter registration agencies:
- (a) Such offices that provide public assistance as are designated by the Secretary of State;
- (b) Each office that receives money from the State of Nevada to provide services to persons with disabilities in this State;
 - (c) The offices of the Department of Motor Vehicles;
 - (d) The offices of the city and county clerks;
- (e) Such other county and municipal facilities as a county clerk or city clerk may designate pursuant to NRS 293.5035 or 293C.520, as applicable; [and]
 - (f) Recruitment offices of the United States Armed Forces; and
 - (g) Such other offices as the Secretary of State deems appropriate.
 - 2. Each voter registration agency shall:
- (a) Post in a conspicuous place, in at least 12-point type, instructions for registering to vote;
- (b) [Make] Except as otherwise provided in subsection 3, distribute applications to register to vote which may be returned by mail [available to each person who applies for or receives] with any application for services or assistance from the agency [;] or submitted for any other purpose and with each application for recertification, renewal or change of address

submitted to the agency that relates to such services, assistance or other purpose;

- (c) Provide the same amount of assistance to an applicant in completing an application to register to vote as the agency provides to a person completing any other forms for the agency; and
 - (d) Accept completed applications to register to vote.
- 3. A voter registration agency is not required to provide an application to register to vote pursuant to paragraph (b) of subsection 2 to a person who applies for or receives services or assistance from the agency or submits an application for any other purpose if the person declines to register to vote and submits to the agency a written form that meets the requirements of 42 U.S.C. § 1973gg-5(a)(6). No information related to the declination to register to vote may be used for any purpose other than voter registration.
- 4. Except as otherwise provided in this subsection and NRS 293.524, any application to register to vote accepted by a voter registration agency must be transmitted to the county clerk not later than 10 days after the application is accepted. The applications must be forwarded daily during the 2 weeks immediately preceding the fifth Sunday preceding an election. The county clerk shall accept any application to register to vote which is obtained from a voter registration agency pursuant to this section and completed by the fifth Sunday preceding an election if the county clerk receives the application not later than 5 days after that date.
- [4.] 5. The Secretary of State shall cooperate with the Secretary of Defense to develop and carry out procedures to enable persons in this State to apply to register to vote at recruitment offices of the United States Armed Forces.
 - **Sec. 21.** NRS 293.5045 is hereby amended to read as follows:
- 293.5045 1. A person who works in a voter registration agency shall not:
- (a) Seek to influence an applicant's political preference or party registration;
- (b) Display a political preference or party allegiance in a place where it can be seen by an applicant;
- (c) Make any statement or take any action to discourage an applicant from registering to vote; or
- (d) Make any statement or take any action which would lead the applicant to believe that a decision to register to vote has any effect on the availability of any services or benefits provided by the State or Federal Government.
- 2. A person who violates any of the provisions of $[this\ section]$ subsection I is guilty of a category E felony and shall be punished as provided in NRS 193.130.

- 3. A voter registration agency shall not knowingly employ a person whose duties will include the registration of voters if the person has been convicted of a felony involving theft or fraud. The Secretary of State may bring an action against a voter registration agency to collect a civil penalty of not more than \$5,000 for each person who is employed by the voter registration agency in violation of this subsection. Any civil penalty collected pursuant to this subsection must be deposited with the State Treasurer for credit to the State General Fund.
 - **Sec. 22.** NRS 293.505 is hereby amended to read as follows:
- 293.505 1. All justices of the peace, except those located in county seats, are ex officio field registrars to carry out the provisions of this chapter.
- 2. The county clerk shall appoint at least one registered voter to serve as a field registrar of voters who, except as otherwise provided in NRS 293.5055, shall register voters within the county for which the field registrar is appointed. Except as otherwise provided in subsection 1, a candidate for any office may not be appointed or serve as a field registrar. A field registrar serves at the pleasure of the county clerk and shall perform such duties as the county clerk may direct. The county clerk shall not knowingly appoint any person as a field registrar who has been convicted of a felony involving theft or fraud. The Secretary of State may bring an action against a county clerk to collect a civil penalty of not more than \$5,000 for each person who is appointed as a field registrar in violation of this subsection. Any civil penalty collected pursuant to this subsection must be deposited with the State Treasurer for credit to the State General Fund.
- 3. A field registrar shall demand of any person who applies for registration all information required by the application to register to vote and shall administer all oaths required by this chapter.
- 4. When a field registrar has in his or her possession five or more completed applications to register to vote, the field registrar shall forward them to the county clerk, but in no case may the field registrar hold any number of them for more than 10 days.
- 5. Each field registrar shall forward to the county clerk all completed applications in his or her possession immediately after the fifth Sunday preceding an election. Within 5 days after the fifth Sunday preceding any general election or general city election, a field registrar shall return all unused applications in his or her possession to the county clerk. If all of the unused applications are not returned to the county clerk, the field registrar shall account for the unreturned applications.
- 6. Each field registrar shall submit to the county clerk a list of the serial numbers of the completed applications to register to vote and the names of the electors on those applications. The serial numbers must be listed in numerical order.

- 7. Each field registrar shall post notices sent to him or her by the county clerk for posting in accordance with the election laws of this State.
- 8. A field registrar, employee of a voter registration agency or person assisting a voter pursuant to subsection 13 of NRS 293.5235 shall not:
 - (a) Delegate any of his or her duties to another person; or
- (b) Refuse to register a person on account of that person's political party affiliation.
- 9. A person shall not hold himself or herself out to be or attempt to exercise the duties of a field registrar unless the person has been so appointed.
- 10. A county clerk, field registrar, employee of a voter registration agency or person assisting a voter pursuant to subsection 13 of NRS 293.5235 shall not:
 - (a) Solicit a vote for or against a particular question or candidate;
- (b) Speak to a voter on the subject of marking his or her ballot for or against a particular question or candidate; or
- (c) Distribute any petition or other material concerning a candidate or question which will be on the ballot for the ensuing election,
- → while registering an elector.
- 11. When the county clerk receives applications to register to vote from a field registrar, the county clerk shall issue a receipt to the field registrar. The receipt must include:
 - (a) The number of persons registered; and
 - (b) The political party of the persons registered.
- 12. A county clerk, field registrar, employee of a voter registration agency or person assisting a voter pursuant to subsection 13 of NRS 293.5235 shall not:
- (a) Knowingly register a person who is not a qualified elector or a person who has filed a false or misleading application to register to vote; or
- (b) Register a person who fails to provide satisfactory proof of identification and the address at which the person actually resides.
- 13. A county clerk, field registrar, employee of a voter registration agency, person assisting a voter pursuant to subsection 13 of NRS 293.5235 or any other person providing a form for the application to register to vote to an elector for the purpose of registering to vote:
- (a) If the person who assists an elector with completing the form for the application to register to vote retains the form, shall enter his or her name on the duplicate copy or receipt retained by the voter upon completion of the form; and
- (b) Shall not alter, deface or destroy an application to register to vote that has been signed by an elector except to correct information contained in the

application after receiving notice from the elector that a change in or addition to the information is required.

- 14. If a field registrar violates any of the provisions of this section, the county clerk shall immediately suspend the field registrar and notify the district attorney of the county in which the violation occurred.
- 15. A person who violates any of the provisions of subsection 8, 9, 10, 12 or 13 is guilty of a category E felony and shall be punished as provided in NRS 193.130.
 - **Sec. 23.** NRS 293.506 is hereby amended to read as follows:
- 293.506 1. A county clerk may, with approval of the board of county commissioners, establish a system for using a computer to register voters and to keep records of registration. [The county clerk may, for that purpose, issue to a voter a card, bearing the signature of the voter, attesting to the voter's registration.]
- 2. A system established pursuant to subsection 1 must comply with any procedures and requirements prescribed by the Secretary of State pursuant to NRS 293.250.
 - **Sec. 24.** NRS 293.517 is hereby amended to read as follows:
- 293.517 1. Any elector residing within the county may register [:] to vote:
- (a) Except as otherwise provided in NRS 293.560 and 293C.527, by appearing before the county clerk, a field registrar or a voter registration agency, completing the application to register to vote, giving true and satisfactory answers to all questions relevant to his or her identity and right to vote, and providing proof of residence and identity;
- (b) By completing and mailing or personally delivering to the county clerk an application to register to vote pursuant to the provisions of NRS 293.5235;
 - (c) Pursuant to the provisions of NRS 293.501 or 293.524; [or]
- (d) At his or her residence with the assistance of a field registrar pursuant to NRS 293.5237 [-]; or
- (e) By submitting an application to register to vote by computer, if the county clerk has established a system pursuant to NRS 293.506 for using a computer to register voters.
- → The county clerk shall require a person to submit official identification as proof of residence and identity, such as a driver's license or other official document, before registering the person. If the applicant registers to vote pursuant to this subsection and fails to provide proof of residence and identity, the applicant must provide proof of residence and identity before casting a ballot in person or by mail or after casting a provisional ballot pursuant to NRS 293.3081 or 293.3083. For the purposes of this subsection, a voter registration card issued pursuant to subsection 6 does not provide proof of the residence or identity of a person.

- 2. The application to register to vote must be signed and verified under penalty of perjury by the elector registering.
- 3. Each elector who is or has been married must be registered under his or her own given or first name, and not under the given or first name or initials of his or her spouse.
- 4. An elector who is registered and changes his or her name must complete a new application to register to vote. The elector may obtain a new application:
 - (a) At the office of the county clerk or field registrar;
- (b) By submitting an application to register to vote pursuant to the provisions of NRS 293.5235;
- (c) By submitting a written statement to the county clerk requesting the county clerk to mail an application to register to vote; [or]
 - (d) At any voter registration agency [.]; or
- (e) By submitting an application to register to vote by computer, if the county clerk has established a system pursuant to NRS 293.506 for using a computer to register voters.
- → If the elector fails to register under his or her new name, the elector may be challenged pursuant to the provisions of NRS 293.303 or 293C.292 and may be required to furnish proof of identity and subsequent change of name.
- 5. Except as otherwise provided in subsection 7, an elector who registers to vote pursuant to paragraph (a) of subsection 1 shall be deemed to be registered upon the completion of an application to register to vote.
- 6. After the county clerk determines that the application to register to vote of a person is complete and that the person is eligible to vote pursuant to NRS 293.485, the county clerk shall issue a voter registration card to the voter which contains:
- (a) The name, address, political affiliation and precinct number of the voter;
 - (b) The date of issuance; and
 - (c) The signature of the county clerk.
- 7. If an elector submits an application to register to vote or an affidavit described in paragraph (c) of subsection 1 of NRS 293.507 that contains any handwritten additions, erasures or interlineations, the county clerk may object to the application to register to vote if the county clerk believes that because of such handwritten additions, erasures or interlineations, the application to register to vote of the elector is incomplete or that the elector is not eligible to vote pursuant to NRS 293.485. If the county clerk objects pursuant to this subsection, he shall immediately notify the elector and the district attorney of the county. Not later than 5 business days after the district attorney receives such notification, the district attorney shall advise the county clerk as to whether:

- (a) The application to register to vote of the elector is complete and the elector is eligible to vote pursuant to NRS 293.485; and
- (b) The county clerk should proceed to process the application to register to vote.
- → If the District Attorney advises the county clerk to process the application to register to vote, the county clerk shall immediately issue a voter registration card to the applicant pursuant to subsection 6.
 - **Sec. 25.** NRS 293.5235 is hereby amended to read as follows:
- 293.5235 1. Except as otherwise provided in NRS 293.502, a person may register to vote by mailing an application to register to vote to the county clerk of the county in which the person resides [.] or may register to vote by computer, if the county clerk has established a system pursuant to NRS 293.506 for using a computer to register to vote. The county clerk shall, upon request, mail an application to register to vote to an applicant. The county clerk shall make the applications available at various public places in the county. An application to register to vote may be used to correct information in the registrar of voters' register.
- 2. An application to register to vote which is mailed to an applicant by the county clerk or made available to the public at various locations or voter registration agencies in the county may be returned to the county clerk by mail or in person. For the purposes of this section, an application which is personally delivered to the county clerk shall be deemed to have been returned by mail.
- 3. The applicant must complete the application, including, without limitation, checking the boxes described in paragraphs (b) and (c) of subsection 10 and signing the application.
- 4. The county clerk shall, upon receipt of an application, determine whether the application is complete.
- 5. If the county clerk determines that the application is complete, he or she shall, within 10 days after receiving the application, mail to the applicant:
- (a) A notice that the applicant is registered to vote and a voter registration card as required by subsection 6 of NRS 293.517; or
- (b) A notice that the registrar of voters' register has been corrected to reflect any changes indicated on the application.
- 6. Except as otherwise provided in subsection 5 of NRS 293.518, if the county clerk determines that the application is not complete, the county clerk shall, as soon as possible, mail a notice to the applicant that additional information is required to complete the application. If the applicant provides the information requested by the county clerk within 15 days after the county clerk mails the notice, the county clerk shall, within 10 days after receiving the information, mail to the applicant:

- (a) A notice that the applicant is registered to vote and a voter registration card as required by subsection 6 of NRS 293.517; or
- (b) A notice that the registrar of voters' register has been corrected to reflect any changes indicated on the application.
- → If the applicant does not provide the additional information within the prescribed period, the application is void.
- 7. The applicant shall be deemed to be registered or to have corrected the information in the register on the date the application is postmarked or received by the county clerk, whichever is earlier.
- 8. If the applicant fails to check the box described in paragraph (b) of subsection 10, the application shall not be considered invalid and the county clerk shall provide a means for the applicant to correct the omission at the time the applicant appears to vote in person at the assigned polling place.
- 9. The Secretary of State shall prescribe the form for an application to register to vote by [mail]:
 - (a) Mail, which must be used to register to vote by mail in this State.
- (b) Computer, which must be used to register to vote in a county if the county clerk has established a system pursuant to NRS 293.506 for using a computer to register to vote.
 - 10. The application to register to vote by mail must include:
 - (a) A notice in at least 10-point type which states:
 - NOTICE: You are urged to return your application to register to vote to the County Clerk in person or by mail. If you choose to give your completed application to another person to return to the County Clerk on your behalf, and the person fails to deliver the application to the County Clerk, you will not be registered to vote. Please retain the duplicate copy or receipt from your application to register to vote.
- (b) The question, "Are you a citizen of the United States?" and boxes for the applicant to check to indicate whether or not the applicant is a citizen of the United States.
- (c) The question, "Will you be at least 18 years of age on or before election day?" and boxes for the applicant to check to indicate whether or not the applicant will be at least 18 years of age or older on election day.
- (d) A statement instructing the applicant not to complete the application if the applicant checked "no" in response to the question set forth in paragraph (b) or (c).
- (e) A statement informing the applicant that if the application is submitted by mail and the applicant is registering to vote for the first time, the applicant must submit the information set forth in paragraph (a) of subsection 2 of NRS 293.2725 to avoid the requirements of subsection 1 of NRS 293.2725 upon voting for the first time.

- 11. Except as otherwise provided in subsection 5 of NRS 293.518, the county clerk shall not register a person to vote pursuant to this section unless that person has provided all of the information required by the application.
- 12. The county clerk shall mail, by postcard, the notices required pursuant to subsections 5 and 6. If the postcard is returned to the county clerk by the United States Postal Service because the address is fictitious or the person does not live at that address, the county clerk shall attempt to determine whether the person's current residence is other than that indicated on the application to register to vote in the manner set forth in NRS 293.530.
- 13. A person who, by mail, registers to vote pursuant to this section may be assisted in completing the application to register to vote by any other person. The application must include the mailing address and signature of the person who assisted the applicant. The failure to provide the information required by this subsection will not result in the application being deemed incomplete.
- 14. An application to register to vote must be made available to all persons, regardless of political party affiliation.
- 15. An application must not be altered or otherwise defaced after the applicant has completed and signed it. An application must be mailed or delivered in person to the office of the county clerk within 10 days after it is completed.
- 16. A person who willfully violates any of the provisions of subsection 13, 14 or 15 is guilty of a category E felony and shall be punished as provided in NRS 193.130.
- 17. The Secretary of State shall adopt regulations to carry out the provisions of this section.
 - **Sec. 26.** NRS 293.565 is hereby amended to read as follows:
- 293.565 1. Except as otherwise provided in subsection [2,] 3, sample ballots must include:
 - (a) If applicable, the statement required by NRS 293.267;
- (b) The fiscal note or description of anticipated financial effect, as provided pursuant to NRS 218D.810, 293.250, 293.481, 293.482, 295.015 or 295.095 for each proposed constitutional amendment, statewide measure, measure to be voted upon only by a special district or political subdivision and advisory question;
- (c) An explanation, as provided pursuant to NRS 218D.810, 293.250, 293.481, 293.482 or 295.121, of each proposed constitutional amendment, statewide measure, measure to be voted upon only by a special district or political subdivision and advisory question;
- (d) Arguments for and against each proposed constitutional amendment, statewide measure, measure to be voted upon only by a special district or political subdivision and advisory question, and rebuttals to each argument,

as provided pursuant to NRS 218D.810, 293.250, 293.252, 293.481, 293.482 or 295.121; and

- (e) The full text of each proposed constitutional amendment.
- 2. If, pursuant to the provisions of NRS 293.2565, the word "Incumbent" must appear on the ballot next to the name of the candidate who is the incumbent, the word "Incumbent" must appear on the sample ballot next to the name of the candidate who is the incumbent.
- 3. Sample ballots that are mailed to registered voters may be printed without the full text of each proposed constitutional amendment if:
- (a) The cost of printing the sample ballots would be significantly reduced if the full text of each proposed constitutional amendment were not included;
- (b) The county clerk ensures that a sample ballot that includes the full text of each proposed constitutional amendment is provided at no charge to each registered voter who requests such a sample ballot; and
- (c) The sample ballots provided to each polling place include the full text of each proposed constitutional amendment.
- [3.] 4. Before the period for early voting for any election begins, the county clerk shall cause to be mailed to each registered voter in the county a sample ballot for his or her precinct, with a notice informing the voter of the location of his or her polling place. If the location of the polling place has changed since the last election:
- (a) The county clerk shall mail a notice of the change to each registered voter in the county not sooner than 10 days before mailing the sample ballots; or
- (b) The sample ballot must also include a notice in bold type immediately above the location which states:

NOTICE: THE LOCATION OF YOUR POLLING PLACE HAS CHANGED SINCE THE LAST ELECTION

- [4.] 5. Except as otherwise provided in subsection [5,] 6, a sample ballot required to be mailed pursuant to this section must:
 - (a) Be printed in at least 12-point type; and
- (b) Include on the front page, in a separate box created by bold lines, a notice printed in at least 20-point bold type that states:

NOTICE: TO RECEIVE A SAMPLE BALLOT IN LARGE TYPE, CALL (Insert appropriate telephone number)

[5.] 6. A portion of a sample ballot that contains a facsimile of the display area of a voting device may include material in less than 12-point type to the extent necessary to make the facsimile fit on the pages of the sample ballot.

- [6.] 7. The sample ballot mailed to a person who requests a sample ballot in large type by exercising the option provided pursuant to NRS 293.508, or in any other manner, must be printed in at least 14-point type, or larger when practicable.
- [7.] 8. If a person requests a sample ballot in large type, the county clerk shall ensure that all future sample ballots mailed to that person from the county are in large type.
- [8-] 9. The county clerk shall include in each sample ballot a statement indicating that the county clerk will, upon request of a voter who is elderly or disabled, make reasonable accommodations to allow the voter to vote at his or her polling place and provide reasonable assistance to the voter in casting his or her vote, including, without limitation, providing appropriate materials to assist the voter. In addition, if the county clerk has provided pursuant to subsection 4 of NRS 293.2955 for the placement at centralized voting locations of specially equipped voting devices for use by voters who are elderly or disabled, the county clerk shall include in the sample ballot a statement indicating:
 - (a) The addresses of such centralized voting locations;
- (b) The types of specially equipped voting devices available at such centralized voting locations; and
- (c) That a voter who is elderly or disabled may cast his or her ballot at such a centralized voting location rather than at his or her regularly designated polling place.
- [9.] 10. The cost of mailing sample ballots for any election other than a primary or general election must be borne by the political subdivision holding the election.
 - **Sec. 27.** NRS 293.567 is hereby amended to read as follows:
- 293.567 After the close of registration for each primary election but not later than the [second Friday next] Friday preceding the primary election and after the close of registration for each general election but not later than the [second Friday next] Friday preceding the general election, the county clerk shall ascertain by precinct and district the number of registered voters in the county and their political affiliation, if any, and shall transmit that information to the Secretary of State.
 - **Sec. 28.** NRS 293.710 is hereby amended to read as follows:
- 293.710 1. It is unlawful for any person, in connection with any election, [or] petition [,] or registration of voters, whether acting himself or herself or through another person in his or her behalf, to:
- (a) Use or threaten to use any force, *intimidation*, coercion, violence, restraint or undue influence:
- (b) Inflict or threaten to inflict any physical or mental injury, damage, harm or loss upon the person or property of another;

- (c) Expose or publish or threaten to expose or publish any fact concerning another in order to induce or compel such other to vote or refrain from voting for any candidate or any question;
- (d) Impede or prevent, by abduction, duress or fraudulent contrivance, the free exercise of the franchise by any voter, or thereby to compel, induce or prevail upon any elector to give or refrain from giving his or her vote; or
- (e) Discharge or change the place of employment of any employee with the intent to impede or prevent the free exercise of the franchise by such employee.
- 2. [Unless a greater penalty is provided by law, any violation] A person who violates a provision of this section is [a gross misdemeanor.] guilty of a category E felony and shall be punished as provided in NRS 193.130.
 - **Sec. 29.** NRS 293.730 is hereby amended to read as follows:
 - 293.730 1. A person shall not:
- (a) Remain in or outside of any polling place so as to interfere with the conduct of the election.
- (b) Except an election board officer, receive from any voter a ballot prepared by the voter.
 - (c) Remove a ballot from any polling place before the closing of the polls.
- (d) Apply for or receive a ballot at any election precinct or district other than the one at which the person is entitled to vote.
- (e) Show his or her ballot to any person, after voting, so as to reveal any of the names voted for.
- (f) Inside a polling place, ask another person for whom he or she intends to vote.
 - (g) Except an election board officer, deliver a ballot to a voter.
- (h) Except an election board officer in the course of the election board officer's official duties, inside a polling place, ask another person his or her name, address or political affiliation.
 - 2. A voter shall not:
 - (a) Receive a ballot from any person other than an election board officer.
- (b) Deliver to an election board or to any member thereof any ballot other than the one received.
- (c) Place any mark upon his or her ballot by which it may afterward be identified as the one voted by the person.
- 3. Any person who violates any provision of this section is guilty of a [gross misdemeanor.] category E felony and shall be punished as provided in NRS 193.130.
 - **Sec. 30.** NRS 293.740 is hereby amended to read as follows:
- 293.740 1. Except as otherwise provided in subsection 2, it is unlawful inside a polling place or within 100 feet from the entrance to the building or other structure in which a polling place is located:

- (a) For any person to solicit a vote or speak to a voter on the subject of marking the voter's ballot.
- (b) For any person, including an election board officer, to do any electioneering on election day.
- → The county clerk or registrar of voters shall ensure that, at the outer limits of the area within which electioneering is prohibited, notices are continuously posted on which are printed in large letters "Distance Marker: No electioneering between this point and the entrance to the polling place."
- 2. The provisions of subsection 1 do not apply to the conduct of a person in a private residence or on commercial or residential property that is within 100 feet from the entrance to a building or other structure in which a polling place is located. The provisions of subsection 1 are not intended to prohibit a person from voting solely because he or she is wearing a prohibited political insigne and is reasonably unable to remove the insigne or cover it. In such a case, the election board officer shall take such action as is necessary to allow the voter to vote as expediently as possible and then assist the voter in exiting the polling place as soon as is possible.
- 3. Any person who violates any provision of this section is guilty of a gross misdemeanor.
- 4. As used in this section, "electioneering" means campaigning for or against a candidate, ballot question or political party by:
- (a) Posting signs relating to the support of or opposition to a candidate, ballot question or political party;
- (b) Distributing literature relating to the support of or opposition to a candidate, ballot question or political party;
- (c) Using loudspeakers to broadcast information relating to the support of or opposition to a candidate, ballot question or political party;
- (d) Buying, selling, wearing or displaying any badge, button or other insigne which is designed or tends to aid or promote the success or defeat of any political party or a candidate or ballot question to be voted upon at that election; *or*
- (e) [Polling or otherwise soliciting from a voter information as to whether the voter intends to vote or has voted for or against a particular political party, candidate or ballot question; or
 - (f) Soliciting signatures to any kind of petition.
 - Sec. 31. (Deleted by amendment.)
 - **Sec. 32.** NRS 293.755 is hereby amended to read as follows:
- 293.755 1. A person who tampers or interferes with, or attempts to tamper or interfere with, a mechanical voting system, mechanical voting device or any computer program used to count ballots with the intent to prevent the proper operation of that device, system or program is guilty of a category D felony and shall be punished as provided in NRS 193.130.

- 2. A person who tampers or interferes with, or attempts to tamper or interfere with, a mechanical voting system, mechanical voting device or any computer program used to count ballots with the intent to influence the outcome of an election is guilty of a category B felony and shall be punished by imprisonment in the state prison for a minimum term of not less than 2 years and a maximum term of not more than 20 years.
- 3. The county or city clerk shall report any alleged violation of this section to the district attorney who shall cause appropriate proceedings to be instituted and prosecuted in a court of competent jurisdiction without delay.
 - **Sec. 33.** NRS 293.800 is hereby amended to read as follows:
- 293.800 1. A person who, for himself, herself or another person, willfully gives a false answer or answers to questions propounded to the person by the registrar or field registrar of voters relating to the information called for by the application to register to vote, or who willfully falsifies the application in any particular, or who violates any of the provisions of the election laws of this State or knowingly encourages another person to violate those laws is guilty of a category E felony and shall be punished as provided in NRS 193.130.
- 2. A public officer or other person, upon whom any duty is imposed by this title, who willfully neglects his or her duty or willfully performs it in such a way as to hinder the objects and purposes of the election laws of this State, except where another penalty is provided, is guilty of a category E felony and shall be punished as provided in NRS 193.130.
- 3. If the person is a public officer, his or her office is forfeited upon conviction of any offense provided for in subsection 2.
- 4. A person who causes or endeavors to cause his or her name to be registered, knowing that he or she is not an elector or will not be an elector on or before the day of the next ensuing election in the precinct or district in which he or she causes or endeavors to cause the registration to be made, and any other person who induces, aids or abets the person in the commission of either of the acts is guilty of a category E felony and shall be punished as provided in NRS 193.130.
- 5. A field registrar or other person who [:] provides to an elector an application to register to vote and who:
- (a) Knowingly falsifies [an] *the* application [to register to vote] or knowingly causes an application to be falsified; [or]
- (b) Knowingly provides money or other compensation to another for a falsified application [to register to vote,]; or
- (c) Intentionally fails to submit to the county clerk a completed application,
- \rightarrow is guilty of a category E felony and shall be punished as provided in NRS 193.130.

- **Sec. 34.** NRS 293C.265 is hereby amended to read as follows:
- 293C.265 1. Except as otherwise provided in subsection 2 and in NRS 293.2725 and 293.3083, a person who registered *by mail or computer* to vote [pursuant to the provisions of NRS 293.5235] shall, for the first city election in which the person votes at which that registration is valid, vote in person unless he or she has previously voted in the county in which he or she is registered to vote.
 - 2. The provisions of subsection 1 do not apply to a person who:
- (a) Is entitled to vote in the manner prescribed in NRS 293C.342 to 293C.352, inclusive;
- (b) Is entitled to vote an absent ballot pursuant to federal law or NRS 293C.317 or 293C.318;
 - (c) Is disabled;
- (d) Submits or has previously submitted a written request for an absent ballot that is signed by the registered voter before a notary public or other person authorized to administer an oath; or
 - (e) Requests an absent ballot in person at the office of the city clerk.
 - **Sec. 35.** NRS 293C.342 is hereby amended to read as follows:
- 293C.342 1. A registered voter who resides in an election precinct in which there were not more than 200 voters registered for the last preceding city general election, or in a precinct in which it appears to the satisfaction of the city clerk *and Secretary of State* that there are not more than 200 registered voters, may vote at any election regulated by this chapter in the manner provided in NRS 293C.345 to 293C.352, inclusive.
- 2. Whenever the city clerk has designated a precinct as a mailing precinct, registered voters residing in that precinct may vote at any election regulated by this chapter in the manner provided in NRS 293C.345 to 293C.352, inclusive.
 - **Sec. 36.** NRS 293C.530 is hereby amended to read as follows:
- 293C.530 1. Before the period for early voting for any election begins, the city clerk shall cause to be mailed to each registered voter in the city a sample ballot for his or her precinct, with a notice informing the voter of the location of his or her polling place. If the location of the polling place has changed since the last election:
- (a) The city clerk shall mail a notice of the change to each registered voter in the city not sooner than 10 days before mailing the sample ballots; or
- (b) The sample ballot must also include a notice in bold type immediately above the location which states:

NOTICE: THE LOCATION OF YOUR POLLING PLACE HAS CHANGED SINCE THE LAST ELECTION

- 2. Except as otherwise provided in subsection [3,] 4, a sample ballot required to be mailed pursuant to this section must:
 - (a) Be printed in at least 12-point type;
- (b) Include the description of the anticipated financial effect and explanation of each citywide measure and advisory question, including arguments for and against the measure or question, as required pursuant to NRS 293.481, 293.482, 295.205 or 295.217; and
- (c) Include on the front page, in a separate box created by bold lines, a notice printed in at least 20-point bold type that states:

NOTICE: TO RECEIVE A SAMPLE BALLOT IN LARGE TYPE, CALL (Insert appropriate telephone number)

- 3. The word "Incumbent" must appear on the sample ballot next to the name of the candidate who is the incumbent, if required pursuant to NRS 293.2565.
- **4.** A portion of a sample ballot that contains a facsimile of the display area of a voting device may include material in less than 12-point type to the extent necessary to make the facsimile fit on the pages of the sample ballot.
- [4.] 5. The sample ballot mailed to a person who requests a sample ballot in large type by exercising the option provided pursuant to NRS 293.508, or in any other manner, must be printed in at least 14-point type, or larger when practicable.
- [5.] 6. If a person requests a sample ballot in large type, the city clerk shall ensure that all future sample ballots mailed to that person from the city are in large type.
- [6.] 7. The city clerk shall include in each sample ballot a statement indicating that the city clerk will, upon request of a voter who is elderly or disabled, make reasonable accommodations to allow the voter to vote at his or her polling place and provide reasonable assistance to the voter in casting his or her vote, including, without limitation, providing appropriate materials to assist the voter. In addition, if the city clerk has provided pursuant to subsection 4 of NRS 293C.281 for the placement at centralized voting locations of specially equipped voting devices for use by voters who are elderly or disabled, the city clerk shall include in the sample ballot a statement indicating:
 - (a) The addresses of such centralized voting locations;
- (b) The types of specially equipped voting devices available at such centralized voting locations; and
- (c) That a voter who is elderly or disabled may cast his or her ballot at such a centralized voting location rather than at the voter's regularly designated polling place.

- [7.] 8. The cost of mailing sample ballots for a city election must be borne by the city holding the election.
 - **Sec. 37.** (Deleted by amendment.)
 - **Sec. 38.** (Deleted by amendment.)
- **Sec. 39.** (Deleted by amendment.)
- **Sec. 40.** (Deleted by amendment.)
- Sec. 40.5. Chapter 294A of NRS is hereby amended by adding thereto a new section to read as follows:
- 1. A foreign national shall not, directly or indirectly, make a contribution or a commitment to make a contribution to:
 - (a) A candidate;
 - (b) A committee for political action;
 - (c) A committee for the recall of a public officer;
- (d) A person who is not under the direction or control of a candidate, of a group of candidates or of any person involved in the campaign of the candidate or group who makes an expenditure that is not solicited or approved by the candidate or group;
- (e) A political party, committee sponsored by a political party or business entity that makes an expenditure on behalf of a candidate or group of candidates;
- (f) An organization made up of legislative members of a political party whose primary purpose is to provide support for their political efforts;
- (g) A personal campaign committee or the personal representative of a candidate who receives contributions or makes expenditures that are reported as contributions or expenditures by the candidate; or
- (h) A nonprofit corporation that is registered or required to be registered pursuant to NRS 294A.225.
- 2. Except as otherwise provided in subsection 3, a candidate, person, group, committee, political party, organization, business entity or nonprofit corporation described in subsection 1 shall not knowingly solicit, accept or receive a contribution or a commitment to make a contribution from a foreign national.
- 3. For the purposes of subsection 2, if a candidate, person, group, committee, political party, organization, business entity or nonprofit corporation is aware of facts that would lead a reasonable person to inquire whether the source of a contribution is a foreign national, the candidate, person, group, committee, political party, organization, business entity or nonprofit corporation shall be deemed to have not knowingly solicited, accepted or received a contribution in violation of subsection 2 if the candidate, person, group, committee, political party, organization, business entity or nonprofit corporation requests and obtains from the source of the contribution a copy of current and valid United States

- passport papers. This subsection does not apply to any candidate, person, group, committee, political party, organization, business entity or nonprofit corporation if the candidate, person, group, committee, political party, organization, business entity or nonprofit corporation has actual knowledge that the source of the contribution solicited, accepted or received is a foreign national.
- 4. If a candidate, person, group, committee, political party, organization, business entity or nonprofit corporation discovers that the candidate, person, group, committee, political party, organization, business entity or nonprofit corporation received a contribution in violation of this section, the candidate, person, group, committee, political party, organization, business entity or nonprofit corporation shall, if at the time of discovery of the violation:
- (a) Sufficient money received as contributions is available, return the contribution received in violation of this section not later than 30 days after such discovery.
- (b) Except as otherwise provided in paragraph (c), sufficient money received as contributions is not available, return the contribution received in violation of this section as contributions become available for this purpose.
- (c) Sufficient money received as contributions is not available and contributions are no longer being solicited or accepted, not be required to return any amount of the contribution received in violation of this section that exceeds the amount of contributions available for this purpose.
 - 5. A violation of any provision of this section is a gross misdemeanor.
 - 6. As used in this section:
 - (a) "Foreign national" has the meaning ascribed to it in 2 U.S.C. § 441e.
- (b) "Knowingly" means that a candidate, person, group, committee, political party, organization, business entity or nonprofit corporation:
- (1) Has actual knowledge that the source of the contribution solicited, accepted or received is a foreign national;
- (2) Is aware of facts which would lead a reasonable person to conclude that there is a substantial probability that the source of the contribution solicited, accepted or received is a foreign national; or
- (3) Is aware of facts which would lead a reasonable person to inquire whether the source of the contribution solicited, accepted or received is a foreign national, but failed to conduct a reasonable inquiry.
 - **Sec. 41.** NRS 294A.0055 is hereby amended to read as follows:
- 294A.0055 1. "Committee for political action" means any group of natural persons or entities that solicits or receives contributions from any other person, group or entity and:

- (a) Makes or intends to make contributions to candidates or other persons; or
 - (b) Makes or intends to make expenditures,
- ⇒ designed to affect the outcome of any primary [.] election, primary city election, general [or] election, general city election, special election or question on the ballot.
 - 2. "Committee for political action" does not include:
- (a) An organization made up of legislative members of a political party whose primary purpose is to provide support for their political efforts.
- (b) An entity solely because it provides goods or services to a candidate or committee in the regular course of its business at the same price that would be provided to the general public.
 - (c) An individual natural person.
- (d) An individual corporation or other business organization who has filed articles of incorporation or other documentation of organization with the Secretary of State pursuant to title 7 of NRS.
 - (e) A labor union.
- (f) A personal campaign committee or the personal representative of a candidate who receives contributions or makes expenditures that are reported as campaign contributions or expenditures by the candidate.
 - (g) A committee for the recall of a public officer.
 - Sec. 42. NRS 294A.007 is hereby amended to read as follows:
- 294A.007 1. "Contribution" means a gift, loan, conveyance, deposit, payment, transfer or distribution of money or of anything of value other than the services of a volunteer, and includes:
- (a) The payment by any person, other than a candidate, of compensation for the personal services of another person which are rendered to a:
 - (1) Candidate;
- (2) Person who is not under the direction or control of a candidate or group of candidates or of any person involved in the campaign of the candidate or group who makes an expenditure on behalf of the candidate or group which is not solicited or approved by the candidate or group; *or*
- (3) Committee for political action, political party, committee sponsored by a political party or business entity which makes an expenditure on behalf of a candidate or group of candidates, [; or
- (4) Person or group of persons organized formally or informally, including a business entity, who advocates the passage or defeat of a question or group of questions on the ballot,]
- → without charge to the candidate, person, committee or political party.
- (b) The value of services provided in kind for which money would have otherwise been paid, such as paid polling and resulting data, paid direct mail, paid solicitation by telephone, any paid paraphernalia that was printed or

otherwise produced to promote a campaign and the use of paid personnel to assist in a campaign.

- 2. As used in this section, "volunteer" means a person who does not receive compensation of any kind, directly or indirectly, for the services provided to a campaign.
 - **Sec. 43.** NRS 294A.100 is hereby amended to read as follows:
- 294A.100 1. A person shall not make *or commit to make* a contribution or contributions to a candidate for any office, except a federal office, in an amount which exceeds \$5,000 for the primary election or primary city election, regardless of the number of candidates for the office, and \$5,000 for the general election or general city election, regardless of the number of candidates for the office, during the period:
- (a) Beginning from 30 days before the regular session of the Legislature immediately following the last election for the office and ending 30 days before the regular session of the Legislature immediately following the next election for the office, if that office is a state, district, county or township office; or
- (b) Beginning from 30 days after the last election for the office and ending 30 days before the next general city election for the office, if that office is a city office.
- 2. A candidate shall not accept a contribution *or commitment to make a contribution* made in violation of subsection 1.
- 3. A person who willfully violates any provision of this section is guilty of a category E felony and shall be punished as provided in NRS 193.130.
 - Sec. 44. (Deleted by amendment.)
 - **Sec. 45.** (Deleted by amendment.)
 - **Sec. 46.** (Deleted by amendment.)
 - **Sec. 47.** (Deleted by amendment.)
 - **Sec. 48.** NRS 294A.150 is hereby amended to read as follows:
- 294A.150 1. [Except as otherwise provided in NRS 294A.283, every person or group of persons organized formally or informally, including a business entity, who] Every committee for political action that advocates the passage or defeat of a question or group of questions on the ballot at a primary election, primary city election, general election or general city election [and who receives or expends money in an amount in excess of \$10,000 to advocate the passage or defeat of such question or group of questions] shall, not later than January 15 of each year that the provisions of this subsection apply to the [person, group of persons or business entity,] committee for political action, for the period from January 1 of the previous year through December 31 of the previous year, report each campaign contribution in excess of \$1,000 [\$100] received during that period and contributions received during the period from a contributor which

cumulatively exceed \$1,000. [\$100.] The report must be completed on the form designed and provided by the Secretary of State pursuant to NRS 294A.373. The form must be signed by [the person or] a representative of the [group or business entity] committee for political action under penalty of perjury. The provisions of this subsection apply to [the person, group of persons or business entity:] a committee for political action:

- (a) Each year in which [:
- (1) An] an election or city election is held for each question for which the [person, group of persons or business entity] committee for political action advocates passage or defeat; [or
- (2) A person, group of persons or business entity receives or expends money in excess of \$10,000 to advocate the passage or defeat of a question or group of questions on the ballot at a primary election, primary city election, general election or general city election;] and
 - (b) The year after [each] the year described in paragraph (a).
- 2. If a question is on the ballot at a primary election or primary city election and the general election or general city election immediately following that primary election or primary city election is held on or after January 1 and before the July 1 immediately following that January 1, every [person or group of persons organized formally or informally, including a business entity, who] committee for political action that advocates the passage or defeat of the question or a group of questions that includes the question fand who receives or expends money in an amount in excess of \$10,000 to advocate the passage or defeat of such question or group of questions] shall comply with the requirements of this subsection. If a question is on the ballot at a general election or general city election held on or after January 1 and before the July 1 immediately following that January 1, every [person or group of persons organized formally or informally, including a business entity, who] committee for political action that advocates the passage or defeat of the question or a group of questions that includes the question fand who receives or expends money in an amount in excess of \$10,000 to advocate the passage or defeat of such question or group of questions] shall comply with the requirements of this subsection. A [person, group of persons or business entity] committee for political action described in this subsection shall, not later than:
- (a) Seven days before the primary election or primary city election, for the period from the January 1 immediately preceding the primary election or primary city election through 12 days before the primary election or primary city election;
- (b) Seven days before the general election or general city election, for the period from 11 days before the primary election or primary city election through 12 days before the general election or general city election; and

- (c) July 15 of the year of the general election or general city election, for the period from 11 days before the general election or general city election through June 30 of that year,
- report each campaign contribution in excess of \$\frac{\\$1,000}{\}1,\frac{\\$1,001}{\}1\$ received during the period and contributions received during the period from a contributor which cumulatively exceed \$\frac{\\$1,000}{\}1,\frac{\\$1,000}{\}1\$. The report must be completed on the form designed and provided by the Secretary of State pursuant to NRS 294A.373 and signed by \$\frac{\{\}1001\}{\}1\$ representative of the \$\frac{\{\}2001\}{\}2001\} or business entity\$\frac{\}{2}\$ committee for political action under penalty of perjury.
- 3. The name and address of the contributor and the date on which the contribution was received must be included on the report for each contribution in excess of \$1,000 [\$100] and contributions which a contributor has made cumulatively in excess of that amount since the beginning of the current reporting period.
- 4. If a question is on the ballot at a primary election or primary city election and the general election or general city election immediately following that primary election or primary city election is held on or after July 1 and before the January 1 immediately following that July 1, every [person or group of persons organized formally or informally, including a business entity, who] committee for political action that advocates the passage or defeat of the question or a group of questions that includes the question fand who receives or expends money in an amount in excess of \$10,000 to advocate the passage or defeat of such question or group of questions] shall comply with the requirements of this subsection. [Except as otherwise provided in NRS 294A.283, if] If a question is on the ballot at a general election or general city election held on or after July 1 and before the January 1 immediately following that July 1, every [person or group of persons organized formally or informally, including a business entity, whol committee for political action that advocates the passage or defeat of the question or a group of questions that includes the question fand who receives or expends money in an amount in excess of \$10,000 to advocate the passage or defeat of such question or group of questions] shall comply with the requirements of this subsection. A [person, group of persons or business entity] committee for political action described in this subsection shall, not later than:
- (a) Seven days before the primary election or primary city election, for the period from the January 1 immediately preceding the primary election or primary city election through 12 days before the primary election or primary city election; and

- (b) Seven days before the general election or general city election, for the period from 11 days before the primary election or primary city election through 12 days before the general election or general city election,
- report each campaign contribution in excess of \$1,000 [\$100]\$ received during the period and contributions received during the period from a contributor which cumulatively exceed \$1,000. [\$100.]\$ The report must be completed on the form designed and provided by the Secretary of State pursuant to NRS 294A.373. The form must be signed by [the person or] a representative of the [group or business entity] committee for political action under penalty of perjury.
- 5. Except as otherwise provided in subsection 6, every [person or group of persons organized formally or informally, including a business entity, who] committee for political action that advocates the passage or defeat of a question or group of questions on the ballot at a special election [and who receives or expends money in an amount in excess of \$10,000 to advocate the passage or defeat of such question or group of questions] shall, not later than:
- (a) Seven days before the special election, for the period from the date that the question qualified for the ballot through 12 days before the special election; and
- (b) Thirty days after the special election, for the remaining period through the special election,
- report each campaign contribution in excess of \$1,000 [\$100] received during the period and contributions received during the period from a contributor which cumulatively exceed \$1,000. [\$100.] The report must be completed on the form designed and provided by the Secretary of State pursuant to NRS 294A.373. The form must be signed by [the person or] a representative of the [group or business entity] committee for political action under penalty of perjury.
- 6. Every [person or group of persons organized formally or informally, including a business entity, who] committee for political action that advocates the passage or defeat of a question or group of questions on the ballot at a special election to determine whether a public officer will be recalled [and who receives or expends money in an amount in excess of \$10,000 to advocate the passage or defeat of such question or group of questions] shall report each of the contributions received on the form designed and provided by the Secretary of State pursuant to NRS 294A.373 and signed by [the person or] a representative of the [group or business entity] committee for political action under penalty of perjury, 30 days after:
- (a) The special election, for the period from the filing of the notice of intent to circulate the petition for recall through the special election; or
- (b) If the special election is not held because a district court determines that the petition for recall is legally insufficient pursuant to subsection 6 of

NRS 306.040, for the period from the filing of the notice of intent to circulate the petition for recall through the date of the district court's decision.

- 7. The reports required pursuant to this section must be filed with:
- (a) If the question is submitted to the voters of one county, the county clerk of that county;
- (b) If the question is submitted to the voters of one city, the city clerk of that city; or
- (c) If the question is submitted to the voters of more than one county or city, the Secretary of State.
- 8. A person may mail or transmit the report to the appropriate officer by regular mail, certified mail, facsimile machine or electronic means. A report shall be deemed to be filed with the officer:
 - (a) On the date that it was mailed if it was sent by certified mail; or
- (b) On the date that it was received by the officer if the report was sent by regular mail, transmitted by facsimile machine or electronic means, or delivered personally.
- 9. If the [person or group of persons, including a business entity,] *committee for political action* is advocating passage or defeat of a group of questions, the reports must be itemized by question or petition.
- 10. Each county clerk or city clerk who receives a report pursuant to this section shall file a copy of the report with the Secretary of State within 10 working days after receiving the report.
 - Sec. 49. (Deleted by amendment.)
 - **Sec. 50.** NRS 294A.160 is hereby amended to read as follows:
- 294A.160 1. It is unlawful for a candidate to spend money received as a campaign contribution for the candidate's personal use.
- 2. Every candidate for a state, district, county, city or township office at a primary, general, primary city, general city or special election who is elected to that office and received contributions that were not spent or committed for expenditure before the primary, general, primary city, general city or special election shall:
 - (a) Return the unspent money to contributors;
- (b) Use the money in the candidate's next election or for the payment of other expenses related to public office or his or her campaign, regardless of whether he or she is a candidate for a different office in the candidate's next election;
 - (c) Contribute the money to:
- (1) The campaigns of other candidates for public office or for the payment of debts related to their campaigns;
 - (2) A political party; or
- (3) [A person or group of persons advocating the passage or defeat of a question or group of questions on the ballot; or

- (4)] Any combination of persons or groups set forth in subparagraphs (1) [,] and (2); [and (3);]
 - (d) Donate the money to any tax-exempt nonprofit entity; [or]
- (e) Donate the money to any governmental entity or fund of this State or a political subdivision of this State [that is authorized to receive donations of money] and may request that the money be used for a specific purpose; or
- 3. Every candidate for a state, district, county, city or township office at a primary, general, primary city, general city or special election who withdraws after filing a declaration of candidacy or an acceptance of candidacy or is [not elected to] defeated for that office and who received contributions that were not spent or committed for expenditure before the primary, general, primary city, general city or special election shall, not later than the 15th day of the second month after the [candidate's defeat:] election:
 - (a) Return the unspent money to contributors;
 - (b) Contribute the money to:
- (1) The campaigns of other candidates for public office or for the payment of debts related to their campaigns;
 - (2) A political party; or
- (3) [A person or group of persons advocating the passage or defeat of a question or group of questions on the ballot;] Any combination of persons or groups set forth in subparagraphs (1) [,] and (2); [and (3);]
 - (c) Donate the money to any tax-exempt nonprofit entity; [or]
- (d) Donate the money to any governmental entity or fund of this State or a political subdivision of this State [that is authorized to receive donations of money] and may request that the money be used for a specific purpose; or
- (e) Dispose of the money in any combination of the methods provided in paragraphs (a) [, (b) and (c).] to (d), inclusive.
- 4. Every candidate for a state, district, county, city or township office who withdraws after filing a declaration of candidacy or an acceptance of candidacy or is defeated for that office at a primary or primary city election and received a contribution from a person in excess of \$5,000 shall, not later than the 15th day of the second month after the [candidate's defeat,] election, return any money in excess of \$5,000 to the contributor.
- 5. [Every] Except as otherwise provided in subsection 6, every public officer who:
 - (a) Holds a state, district, county, city or township office;
- (b) Does not run for reelection *to that office* and is not a candidate for any other office; and

- (c) Has contributions that are not spent or committed for expenditure remaining from a previous election,
- → shall, not later than the 15th day of the second month after the expiration of the public officer's term of office, dispose of those contributions in the manner provided in subsection 3.
 - 6. A public officer who:
 - (a) Holds a state, district, county, city or township office;
- (b) Does not run for reelection to that office and is a candidate for any other office; and
- (c) Has contributions that are not spent or committed for expenditure remaining from a previous election,
- → may use the unspent campaign contributions in a future election. Such a public officer is subject to the reporting requirements set forth in NRS 294A.120, 294A.125, 294A.128, 294A.200, 294A.360 and 294A.362 for as long as the public officer is a candidate for any office.
- 7. In addition to the methods for disposing the unspent money set forth in subsections 2, 3 [and 4,], 4 and 6, a Legislator may donate not more than \$500 of that money to the Nevada Silver Haired Legislative Forum created pursuant to NRS 427A.320.
- [7.] 8. Any contributions received before a candidate for a state, district, county, city or township office at a primary, general, primary city, general city or special election dies that were not spent or committed for expenditure before the death of the candidate must be disposed of in the manner provided in subsection 3.
- [8.] 9. The court shall, in addition to any penalty which may be imposed pursuant to NRS 294A.420, order the candidate or public officer to dispose of any remaining contributions in the manner provided in this section.
- [9.] 10. As used in this section, "contributions" include any interest and other income earned thereon.
 - Sec. 51. NRS 294A.200 is hereby amended to read as follows:
- 294A.200 1. Every candidate for state, district, county or township office at a primary or general election shall, not later than January 15 of each year, for the period from January 1 of the previous year through December 31 of the previous year, report each of the campaign expenses in excess of \$100 incurred and each amount in excess of \$100 disposed of pursuant to NRS 294A.160 *or subsection 4 of NRS 294A.286* during the period on the form designed and provided by the Secretary of State pursuant to NRS 294A.373. The form must be signed by the candidate under penalty of perjury. The provisions of this subsection apply to the candidate:
- (a) Beginning the year of the general election for that office through the year immediately preceding the next general election for that office; and

- (b) Each year immediately succeeding a calendar year during which the candidate disposes of contributions pursuant to NRS 294A.160 [.] or 294A.286.
- 2. Every candidate for state, district, county or township office at a primary or general election shall, if the general election for the office for which he or she is a candidate is held on or after January 1 and before the July 1 immediately following that January 1, not later than:
- (a) Seven days before the primary election for that office, for the period from the January 1 immediately preceding the primary election through 12 days before the primary election;
- (b) Seven days before the general election for that office, for the period from 11 days before the primary election through 12 days before the general election; and
- (c) July 15 of the year of the general election for that office, for the period from 11 days before the general election through June 30 of that year,
- report each of the campaign expenses in excess of \$100 incurred during the period on the form designed and provided by the Secretary of State pursuant *to* NRS 294A.373. Each form must be signed by the candidate under penalty of perjury.
- 3. Every candidate for state, district, county or township office at a primary or general election shall, if the general election for the office for which he or she is a candidate is held on or after July 1 and before the January 1 immediately following that July 1, not later than:
- (a) Seven days before the primary election for that office, for the period from the January 1 immediately preceding the primary election through 12 days before the primary election; and
- (b) Seven days before the general election for that office, for the period from 11 days before the primary election through 12 days before the general election,
- report each of the campaign expenses in excess of \$100 incurred during the period on the form designed and provided by the Secretary of State pursuant to NRS 294A.373. The form must be signed by the candidate under penalty of periury.
- 4. Except as otherwise provided in subsection 5, every candidate for a district office at a special election shall, not later than:
- (a) Seven days before the special election, for the period from the candidate's nomination through 12 days before the special election; and
- (b) Thirty days after the special election, for the remaining period through the special election,
- report each of the campaign expenses in excess of \$100 incurred during the period on the form designed and provided by the Secretary of State

pursuant to NRS 294A.373. Each form must be signed by the candidate under penalty of perjury.

- 5. Every candidate for state, district, county, municipal or township office at a special election to determine whether a public officer will be recalled shall report each of the campaign expenses in excess of \$100 incurred on the form designed and provided by the Secretary of State pursuant to NRS 294A.373 and signed by the candidate under penalty of perjury, 30 days after:
- (a) The special election, for the period from the filing of the notice of intent to circulate the petition for recall through the special election; or
- (b) If the special election is not held because a district court determines that the petition for recall is legally insufficient pursuant to subsection 6 of NRS 306.040, for the period from the filing of the notice of intent to circulate the petition for recall through the date of the district court's decision.
- 6. Reports of campaign expenses must be filed with the officer with whom the candidate filed the declaration of candidacy or acceptance of candidacy. A candidate may mail or transmit the report to that officer by regular mail, certified mail, facsimile machine or electronic means. A report shall be deemed to be filed with the officer:
 - (a) On the date that it was mailed if it was sent by certified mail; or
- (b) On the date that it was received by the officer if the report was sent by regular mail, transmitted by facsimile machine or electronic means, or delivered personally.
- 7. County clerks who receive from candidates for legislative or judicial office, including, without limitation, the office of justice of the peace or municipal judge, reports of campaign expenses pursuant to this section shall file a copy of each report with the Secretary of State within 10 working days after receiving the report.
 - **Sec. 52.** (Deleted by amendment.)
 - Sec. 53. (Deleted by amendment.)
 - Sec. 54. NRS 294A.220 is hereby amended to read as follows:
- 294A.220 1. [Except as otherwise provided in NRS 294A.283, every person or group of persons organized formally or informally, including a business entity, who] Every committee for political action that advocates the passage or defeat of a question or group of questions on the ballot at a primary election, primary city election, general election or general city election [and who receives or expends money in an amount in excess of \$10,000 to advocate the passage or defeat of such question or group of questions] shall, not later than January 15 of each year that the provisions of this subsection apply to the [person or group of persons,] committee for political action, for the period from January 1 of the previous year through December 31 of the previous year, report each expenditure made during the

period on behalf of or against the question, the group of questions or a question in the group of questions on the ballot in excess of \$1,000 [\$100] on the form designed and provided by the Secretary of State pursuant to NRS 294A.373. The form must be signed by [the person or] a representative of the [group or business entity] committee for political action under penalty of perjury. The provisions of this subsection apply to the [person, group of persons or business entity:] committee for political action:

- (a) Each year in which [:
- (1) An] an election or city election is held for a question for which the [person, group of persons or business entity] committee for political action advocates passage or defeat; [or
- (2) A person, group of persons or business entity receives or expends money in excess of \$10,000 to advocate the passage or defeat of a question or group of questions on the ballot at a primary election, primary city election, general election or general city election;] and
 - (b) The year after [each] the year described in paragraph (a).
- 2. If a question is on the ballot at a primary election or primary city election and the general election or general city election immediately following that primary election or primary city election is held on or after January 1 and before the July 1 immediately following that January 1, every [person or group of persons organized formally or informally, including a business entity, who] committee for political action that advocates the passage or defeat of the question or a group of questions that includes the question fand who receives or expends money in an amount in excess of \$10,000 to advocate the passage or defeat of such question or group of questions] shall comply with the requirements of this subsection. If a question is on the ballot at a general election or general city election held on or after January 1 and before the July 1 immediately following that January 1, every [person or group of persons organized formally or informally, including a business entity, who] committee for political action that advocates the passage or defeat of the question or a group of questions that includes the question fand who receives or expends money in an amount in excess of \$10,000 to advocate the passage or defeat of such question or group of questions] shall comply with the requirements of this subsection. A [person, group of persons or business entity] committee for political action described in this subsection shall, not later than:
- (a) Seven days before the primary election or primary city election, for the period from the January 1 immediately preceding the primary election or primary city election through 12 days before the primary election or primary city election;

- (b) Seven days before the general election or general city election, for the period from 11 days before the primary election or primary city election through 12 days before the general election or general city election; and
- (c) July 15 of the year of the general election or general city election, for the period from 11 days before the general election or general city election through the June 30 immediately preceding that July 15,
- → report each expenditure made during the period on behalf of or against the question, the group of questions or a question in the group of questions on the ballot in excess of \$1,000 [\$1,000] on the form designed and provided by the Secretary of State pursuant to NRS 294A.373 and signed by [the person or] a representative of the [group or business entity] committee for political action under penalty of perjury.
- 3. If a question is on the ballot at a primary election or primary city election and the general election or general city election immediately following that primary election or primary city election is held on or after July 1 and before the January 1 immediately following that July 1, every [person or group of persons organized formally or informally, including a business entity, who] committee for political action that advocates the passage or defeat of the question or a group of questions that includes the question fand who receives or expends money in an amount in excess of \$10,000 to advocate the passage or defeat of such question or group of questions] shall comply with the requirements of this subsection. [Except as otherwise provided in NRS 294A.283, if If a question is on the ballot at a general election or general city election held on or after July 1 and before the January 1 immediately following that July 1, every [person or group of persons organized formally or informally, including a business entity, whol committee for political action that advocates the passage or defeat of the question or a group of questions that includes the question [and who receives or expends money in an amount in excess of \$10,000 to advocate the passage or defeat of such question or group of questions] shall comply with the requirements of this subsection. A [person, group of persons or business entity] committee for political action described in this subsection shall, not later than:
- (a) Seven days before the primary election or primary city election, for the period from the January 1 immediately preceding the primary election or primary city election through 12 days before the primary election or primary city election; and
- (b) Seven days before the general election or general city election, for the period from 11 days before the primary election or primary city election through 12 days before the general election or general city election,
- report each expenditure made during the period on behalf of or against the question, the group of questions or a question in the group of questions on

the ballot in excess of \$1,000 [\$100] on the form designed and provided by the Secretary of State pursuant to NRS 294A.373. The form must be signed by [the person or] a representative of the [group or business entity] committee for political action under penalty of perjury.

- 4. Except as otherwise provided in subsection 5, every [person or group of persons organized formally or informally, including a business entity, who] committee for political action that advocates the passage or defeat of a question or group of questions on the ballot at a special election shall, not later than:
- (a) Seven days before the special election, for the period from the date the question qualified for the ballot through 12 days before the special election; and
- (b) Thirty days after the special election, for the remaining period through the special election,
- → report each expenditure made during the period on behalf of or against the question, the group of questions or a question in the group of questions on the ballot in excess of \$1,000 (\$1,000) (\$1,000
- 5. Every [person or group of persons organized formally or informally, including a business entity, who] committee for political action that advocates the passage or defeat of a question or group of questions on the ballot at a special election to determine whether a public officer will be recalled [and who receives or expends money in an amount in excess of \$10,000 to advocate the passage or defeat of such question or group of questions] shall list each expenditure made during the period on behalf of or against the question, the group of questions or a question in the group of questions on the ballot in excess of \$1,000 [\$100] on the form designed and provided by the Secretary of State pursuant to NRS 294A.373 and signed by [the person or] a representative of the [group or business entity] committee for political action under penalty of perjury, 30 days after:
- (a) The special election, for the period from the filing of the notice of intent to circulate the petition for recall through the special election; or
- (b) If the special election is not held because a district court determines that the petition for recall is legally insufficient pursuant to subsection 6 of NRS 306.040, for the period from the filing of the notice of intent to circulate the petition for recall through the date of the district court's decision.
- 6. Expenditures made within the State or made elsewhere but for use within the State, including expenditures made outside the State for printing, television and radio broadcasting or other production of the media, must be included in the report.

- 7. The reports required pursuant to this section must be filed with:
- (a) If the question is submitted to the voters of one county, the county clerk of that county;
- (b) If the question is submitted to the voters of one city, the city clerk of that city; or
- (c) If the question is submitted to the voters of more than one county or city, the Secretary of State.
- 8. If an expenditure is made on behalf of a group of questions, the reports must be itemized by question or petition. A person may mail or transmit the report to the appropriate filing officer by regular mail, certified mail, facsimile machine or electronic means. A report shall be deemed to be filed with the filing officer:
 - (a) On the date that it was mailed if it was sent by certified mail; or
- (b) On the date that it was received by the filing officer if the report was sent by regular mail, transmitted by facsimile machine or electronic means, or delivered personally.
- 9. Each county clerk or city clerk who receives a report pursuant to this section shall file a copy of the report with the Secretary of State within 10 working days after receiving the report.
 - **Sec. 55.** (Deleted by amendment.)
 - **Sec. 56.** NRS 294A.230 is hereby amended to read as follows:
- 294A.230 1. Each committee for political action shall, before it engages in any activity in this State, register with the Secretary of State on forms supplied by the Secretary of State.
 - 2. The form must require:
 - (a) The name of the committee;
 - (b) The purpose for which it was organized;
 - (c) The names, addresses and telephone numbers of its officers;
- (d) If the committee for political action is affiliated with any other organizations, the name, address and telephone number of each organization;
 - (e) The name, address and telephone number of its registered agent; and
 - (f) Any other information deemed necessary by the Secretary of State.
- 3. A committee for political action shall file with the Secretary of State [an]:
- (a) An amended form for registration within 30 days after any change in the information contained in the form for registration.
- (b) A form for registration on or before January 15 of each year, regardless of whether there is a change in the information contained in the most recent form for registration filed by the committee for political action with the Secretary of State.
- 4. The Secretary of State shall include on the Secretary of State's Internet website the information required pursuant to subsection 2.

- **Sec. 57.** (Deleted by amendment.)
- **Sec. 58.** (Deleted by amendment.)
- **Sec. 59.** NRS 294A.286 is hereby amended to read as follows:
- 294A.286 1. A person who administers a legal defense fund shall:
- (a) Within 5 days after the creation of the legal defense fund, notify the Secretary of State of the creation of the fund on a form provided by the Secretary of State; and
- (b) For the same period covered by the report filed pursuant to NRS 294A.120, 294A.200 or 294A.360, report any contribution received by or expenditure made from the legal defense fund.
- 2. The reports required by paragraph (b) of subsection 1 must be submitted on the form designed and provided by the Secretary of State pursuant to NRS 294A.373. Each form must be signed by the administrator of the legal defense fund under penalty of perjury.
- 3. The reports required by paragraph (b) of subsection 1 must be filed in the same manner and at the same time as the report filed pursuant to NRS 294A.120, 294A.200 or 294A.360.
- 4. Not later than the 15th day of the second month after the conclusion of all civil, criminal or administrative claims or proceedings for which a candidate or public officer established a legal defense fund, the candidate or public officer shall:
 - (a) Return the unspent money to contributors;
 - (b) Donate the money to any tax-exempt nonprofit entity; or
- (c) Dispose of the money in any combination of the methods provided in paragraphs (a) and (b).
 - Sec. 60. NRS 294A.287 is hereby amended to read as follows:
- 294A.287 1. A person shall not make *or commit to make* a contribution or contributions to the legal defense fund of a candidate or public officer in an amount which exceeds \$10,000 during the applicable period prescribed in NRS 294A.100 pertaining to the office the candidate is seeking or that the public officer holds.
- 2. A candidate or public officer shall not accept a contribution *or commitment to make a contribution* to his or her legal defense fund that is made in violation of subsection 1.
- 3. A person who willfully violates any provision of this section is guilty of a category E felony and shall be punished as provided in NRS 193.130.
 - **Sec. 61.** NRS 294A.300 is hereby amended to read as follows:
- 294A.300 1. It is unlawful for a member of the Legislature, the Lieutenant Governor, the Lieutenant Governor-Elect, the Governor or the Governor-Elect to solicit or accept any monetary contribution, or solicit or accept a commitment to make such a contribution for any political purpose during the period beginning:

- (a) Thirty days before a regular session of the Legislature and ending 30 days after the final adjournment of a regular session of the Legislature;
- (b) Fifteen days before a special session of the Legislature is set to commence and ending 15 days after the final adjournment of a special session of the Legislature, if the Governor sets a specific date for the commencement of the special session that is more than 15 days after the Governor issues the proclamation calling for the special session; or
- (c) The day after the Governor issues a proclamation calling for a special session of the Legislature and ending 15 days after the final adjournment of a special session of the Legislature if the Governor sets a specific date for the commencement of the special session that is 15 or fewer days after the Governor issues the proclamation calling for the special session.
- 2. A person shall not make or commit to make a contribution or commitment prohibited by subsection 1.
- 3. This section does not prohibit the payment of a salary or other compensation or income to a member of the Legislature, the Lieutenant Governor or the Governor during a session of the Legislature if it is made for services provided as a part of his or her regular employment or is additional income to which he or she is entitled.
- [3.] 4. As used in this section, "political purpose" includes, without limitation, the establishment of, or the addition of money to, a legal defense fund.
 - **Sec. 62.** NRS 294A.347 is hereby amended to read as follows:

294A.347 1. A statement which:

- (a) Is published within 60 days before a general election, general city election or special election or 30 days before a primary election or primary city election;
- (b) Expressly advocates the election or defeat of a clearly identified candidate for a state or local office; and
- (c) Is published by a person who receives compensation from the candidate, an opponent of the candidate [,] or a person, party [, committee or business entity required to report expenditures pursuant to NRS 294A.210,] or committee for political action,
- \rightarrow must contain a disclosure of the fact that the person receives compensation pursuant to paragraph (c) and the name of the person, party [, committee or business entity] or committee for political action providing that compensation.
 - 2. A statement which:
- (a) Is published by a candidate within 60 days before a general election, general city election or special election or 30 days before a primary election or primary city election; and
 - (b) Contains the name of the candidate,

- ⇒ shall be deemed to comply with the provisions of this section.
 - 3. As used in this section, "publish" means the act of:
 - (a) Printing, posting, broadcasting, mailing or otherwise disseminating; or
- (b) Causing to be printed, posted, broadcasted, mailed or otherwise disseminated.
 - **Sec. 63.** NRS 294A.360 is hereby amended to read as follows:
- 294A.360 1. Every candidate for city office at a primary city election or general city election shall file the reports in the manner required by NRS 294A.120, 294A.128 and 294A.200 for other offices not later than January 15 of each year, for the period from January 1 of the previous year through December 31 of the previous year. The provisions of this subsection apply to the candidate:
- (a) Beginning the year of the general city election for that office through the year immediately preceding the next general city election for that office; and
- (b) Each year immediately succeeding a calendar year during which the candidate disposes of contributions pursuant to NRS 294A.160 [.] or subsection 4 of NRS 294A.286.
- 2. Every candidate for city office at a primary city election or general city election, if the general city election for the office for which he or she is a candidate is held on or after January 1 and before the July 1 immediately following that January 1, shall file the reports in the manner required by NRS 294A.120, 294A.128 and 294A.200 for other offices not later than:
- (a) Seven days before the primary city election for that office, for the period from the January 1 immediately preceding the primary city election through 12 days before the primary city election;
- (b) Seven days before the general city election for that office, for the period from 11 days before the primary city election through 12 days before the general city election; and
- (c) July 15 of the year of the general city election for that office, for the period from 11 days before the general city election through the June 30 of that year.
- 3. Every candidate for city office at a primary city election or general city election, if the general city election for the office for which he or she is a candidate is held on or after July 1 and before the January 1 immediately following that July 1, shall file the reports in the manner required by NRS 294A.120, 294A.128 and 294A.200 for other offices not later than:
- (a) Seven days before the primary city election for that office, for the period from the January 1 immediately preceding the primary city election through 12 days before the primary city election; and

- (b) Seven days before the general city election for that office, for the period from 11 days before the primary city election through 12 days before the general city election.
- 4. Except as otherwise provided in subsection 5, every candidate for city office at a special election shall so file those reports:
- (a) Seven days before the special election, for the period from the candidate's nomination through 12 days before the special election; and
- (b) Thirty days after the special election, for the remaining period through the special election.
- 5. Every candidate for city office at a special election to determine whether a public officer will be recalled shall so file those reports 30 days after:
- (a) The special election, for the period from the filing of the notice of intent to circulate the petition for recall through the special election; or
- (b) If the special election is not held because a district court determines that the petition for recall is legally insufficient pursuant to subsection 6 of NRS 306.040, for the period from the filing of the notice of intent to circulate the petition for recall through the date of the district court's decision.
 - **Sec. 64.** (Deleted by amendment.)
 - **Sec. 65.** NRS 294A.365 is hereby amended to read as follows:
- 2. The categories of expense or expenditure for use on the report of expenses or expenditures are:
 - (a) Office expenses;
 - (b) Expenses related to volunteers;
 - (c) Expenses related to travel;
 - (d) Expenses related to advertising;
 - (e) Expenses related to paid staff;
 - (f) Expenses related to consultants;
 - (g) Expenses related to polling;
 - (h) Expenses related to special events;
- (i) Except as otherwise provided in NRS 294A.362, goods and services provided in kind for which money would otherwise have been paid; and
 - (j) Other miscellaneous expenses.

- 3. Each report of expenses or expenditures described in subsection 1 must list the disposition of any unspent campaign contributions using the categories set forth in subsection 2 of NRS 294A.160 [-] or subsection 4 of NRS 294A.286.
 - **Sec. 66.** NRS 294A.373 is hereby amended to read as follows:
- 294A.373 1. The Secretary of State shall design a single form to be used for all reports of campaign contributions and expenses or expenditures that are required to be filed pursuant to NRS 294A.120, 294A.125, 294A.128, 294A.140, 294A.150, 294A.200, 294A.210, 294A.220, 294A.270, 294A.280, [294A.283,] 294A.360 and 294A.362 and reports of contributions received by and expenditures made from a legal defense fund that are required to be filed pursuant to NRS 294A.286.
- 2. The form designed by the Secretary of State pursuant to this section must only request information specifically required by statute.
- 3. Upon request, the Secretary of State shall provide a copy of the form designed pursuant to this section to each person, committee, political party [, group] and business entity that is required to file a report described in subsection 1.
- 4. The Secretary of State must obtain the advice and consent of the Legislative Commission before providing a copy of a form designed or revised by the Secretary of State pursuant to this section to a person, committee, political party [, group] or business entity that is required to use the form.
 - **Sec. 67.** (Deleted by amendment.)
 - **Sec. 68.** NRS 294A.382 is hereby amended to read as follows:
- 294A.382 The Secretary of State shall not request or require a candidate, person, [group of persons,] committee, political party or business entity to list each of the expenditures or campaign expenses of \$100 or less on a form designed and provided pursuant to NRS 294A.373.
 - **Sec. 69.** NRS 294A.390 is hereby amended to read as follows:
- 294A.390 The officer from whom a candidate or entity requests a form for:
 - 1. A declaration of candidacy;
 - 2. An acceptance of candidacy;
- 3. The registration of a committee for political action pursuant to NRS 294A.230, a committee for the recall of a public officer pursuant to NRS 294A.250 or a business entity that wishes to engage in certain political activity pursuant to NRS 294A.377;
- 4. The reporting of the creation of a legal defense fund pursuant to NRS 294A.286; or
- 5. The reporting of campaign contributions, expenses or expenditures pursuant to NRS 294A.120, 294A.128, 294A.140, 294A.150, 294A.200,

294A.210, 294A.220, 294A.270, 294A.280 [, 294A.283] or 294A.360 and the reporting of contributions received by and expenditures made from a legal defense fund pursuant to NRS 294A.286,

⇒ shall furnish the candidate with the necessary forms for reporting and copies of the regulations adopted by the Secretary of State pursuant to this chapter. An explanation of the applicable provisions of NRS 294A.100, 294A.120, 294A.128, 294A.140, 294A.150, 294A.200, 294A.210, 294A.220, 294A.270, 294A.280 [, 294A.283] or 294A.360 relating to the making, accepting or reporting of campaign contributions, expenses or expenditures and the penalties for a violation of those provisions as set forth in NRS 294A.100 or 294A.420, and an explanation of NRS 294A.286 and 294A.287 relating to the accepting or reporting of contributions received by and expenditures made from a legal defense fund and the penalties for a violation of those provisions as set forth in NRS 294A.287 and 294A.420, must be developed by the Secretary of State and provided upon request. The candidate or entity shall acknowledge receipt of the material.

Sec. 70. (Deleted by amendment.)

Sec. 71. NRS 294A.400 is hereby amended to read as follows:

294A.400 The Secretary of State shall, within 30 days after receipt of the reports required by NRS 294A.120, 294A.125, 294A.128, 294A.140, 294A.150, 294A.200, 294A.210, 294A.220, 294A.270, 294A.280 $\frac{1}{5}$, 294A.283 and 294A.286, prepare and make available for public inspection a compilation of:

- 1. The total campaign contributions, the contributions which are in excess of \$100 and the total campaign expenses of each of the candidates from whom reports of those contributions and expenses are required.
- 2. The total amount of loans to a candidate guaranteed by a third party, the total amount of loans made to a candidate that have been forgiven and the total amount of written commitments for contributions received by a candidate.
- 3. The contributions made to a committee for the recall of a public officer in excess of \$100.
 - 4. The expenditures exceeding \$100 made by a:
 - (a) Person on behalf of a candidate other than the person.
- (b) Group of persons or business entity advocating the election or defeat of a candidate.
 - (c) Committee for the recall of a public officer.
 - 5. The contributions in excess of \$100 made to:
- (a) A person who is not under the direction or control of a candidate or group of candidates or of any person involved in the campaign of the candidate or group who makes an expenditure on behalf of the candidate or group which is not solicited or approved by the candidate or group.

- (b) A committee for political action, political party, committee sponsored by a political party or business entity which makes an expenditure on behalf of a candidate or group of candidates.
- 6. [The contributions in excess of \$1,000 made to and the expenditures exceeding \$1,000 made by a:
- (a) Person or group of persons organized formally or informally, including a business entity who advocates the passage or defeat of a question or group of questions on the ballot and who receives or expends money in an amount in excess of \$10,000 for such advocacy, except as otherwise provided in paragraph (b).
- (b) Person or group of persons organized formally or informally, including a business entity, who advocates the passage or defeat of a constitutional amendment or statewide measure proposed by an initiative or referendum, including, without limitation, the initiation or circulation thereof, and who receives or expends money in an amount in excess of \$10,000 for such advocacy.
- 7.] The total contributions received by and expenditures made from a legal defense fund.
 - **Sec. 72.** NRS 294A.420 is hereby amended to read as follows:
- 294A.420 1. If the Secretary of State receives information that a person, *committee* or entity that is subject to the provisions of NRS 294A.120, 294A.128, 294A.140, 294A.150, 294A.200, 294A.210, 294A.220, 294A.227, 294A.230, *294A.250*, 294A.270, 294A.280, [294A.283,] 294A.286 or 294A.360 has not filed a report or form for registration pursuant to the applicable provisions of those sections, the Secretary of State may, after giving notice to that person , *committee* or entity, cause the appropriate proceedings to be instituted in the First Judicial District Court.
- 2. Except as otherwise provided in this section, a person, *committee* or entity that violates an applicable provision of NRS 294A.112, 294A.120, 294A.128, 294A.130, 294A.140, 294A.150, 294A.160, 294A.200, 294A.210, 294A.220, 294A.227, 294A.230, **294A.250**, 294A.270, 294A.280, [294A.283,] 294A.286, 294A.300, 294A.310 or 294A.360 *or section 40.5 of this act* is subject to a civil penalty of not more than \$5,000 for each violation and payment of court costs and attorney's fees. The civil penalty must be recovered in a civil action brought in the name of the State of Nevada by the Secretary of State in the First Judicial District Court and deposited by the Secretary of State for credit to the State General Fund in the bank designated by the State Treasurer.
- 3. If a civil penalty is imposed because a person, *committee* or entity has reported its contributions, expenses or expenditures after the date the report is due, except as otherwise provided in this subsection, the amount of the civil penalty is:

- (a) If the report is not more than 7 days late, \$25 for each day the report is late.
- (b) If the report is more than 7 days late but not more than 15 days late, \$50 for each day the report is late.
- (c) If the report is more than 15 days late, \$100 for each day the report is late.
- → A civil penalty imposed pursuant to this subsection against a public officer who by law is not entitled to receive compensation for his or her office or a candidate for such an office must not exceed a total of \$100 if the public officer or candidate received no contributions and made no expenditures during the relevant reporting periods.
- 4. For good cause shown, the Secretary of State may waive a civil penalty that would otherwise be imposed pursuant to this section. If the Secretary of State waives a civil penalty pursuant to this subsection, the Secretary of State shall:
- (a) Create a record which sets forth that the civil penalty has been waived and describes the circumstances that constitute the good cause shown; and
- (b) Ensure that the record created pursuant to paragraph (a) is available for review by the general public.
 - **Sec. 73.** NRS 306.040 is hereby amended to read as follows:
- 306.040 1. Upon determining that the number of signatures on a petition to recall is sufficient pursuant to NRS 293.1276 to 293.1279, inclusive, the Secretary of State shall notify the county clerk, the officer with whom the petition is to be filed pursuant to subsection 4 of NRS 306.015 and the public officer who is the subject of the petition.
- 2. After the verification of signatures is complete, but not later than the date a complaint is filed pursuant to subsection 5 or the date the call for a special election is issued, whichever is earlier, a person who signs a petition to recall may request the Secretary of State to strike the person's name from the petition. If the person demonstrates good cause therefor and the number of such requests received by the Secretary of State could affect the sufficiency of the petition, the Secretary of State shall strike the name of the person from the petition.
- 3. Not sooner than 10 days nor more than 20 days after the Secretary of State completes the notification required by subsection 1, if a complaint is not filed pursuant to subsection 5, the officer with whom the petition is filed shall issue a call for a special election in the jurisdiction in which the public officer who is the subject of the petition was elected to determine whether the people will recall the public officer.
- 4. The call for a special election pursuant to subsection 3 or 6 must include, without limitation:

- (a) The last day on which a person may register to vote to qualify to vote in the special election; [and]
- (b) The last day on which a petition to nominate other candidates for the office may be filed $[\cdot, \cdot]$; and
- (c) Whether any person is entitled to vote in the special election pursuant to NRS 293.343 to 293.355, inclusive.
- 5. The legal sufficiency of the petition may be challenged by filing a complaint in district court not later than 5 days, Saturdays, Sundays and holidays excluded, after the Secretary of State completes the notification required by subsection 1. All affidavits and documents in support of the challenge must be filed with the complaint. The court shall set the matter for hearing not later than 30 days after the complaint is filed and shall give priority to such a complaint over all other matters pending with the court, except for criminal proceedings.
- 6. Upon the conclusion of the hearing, if the court determines that the petition is sufficient, it shall order the officer with whom the petition is filed to issue a call for a special election in the jurisdiction in which the public officer who is the subject of the petition was elected to determine whether the people will recall the public officer. If the court determines that the petition is not sufficient, it shall order the officer with whom the petition is filed to cease any further proceedings regarding the petition.
 - Sec. 74. (Deleted by amendment.)
 - Sec. 75. (Deleted by amendment.)
 - Sec. 76. (Deleted by amendment.)
 - Sec. 77. (Deleted by amendment.)
 - Sec. 78. (Deleted by amendment.)
 - **Sec. 79.** (Deleted by amendment.)
 - Sec. 80. (Deleted by amendment.)
 - Sec. 81. (Deleted by amendment.)
 - Sec. 82. (Deleted by amendment.)
 - Sec. 83. (Deleted by amendment.)
- **Sec. 84.** NRS 294A.281, 294A.282, 294A.283 and 294A.284 are hereby repealed.
- **Sec. 85.** 1. This section, sections 1 to 37, inclusive, <u>40.5</u>, 41, 42, 43, 48, 50, 51, 54, 56, 59 to 63, inclusive, 65, 66, 68, 69, 71, 72, 73 and 84 of this act become effective on July 1, 2011.
- 2. Sections 38, 39, 40, 44 to 47, inclusive, 49, 52, 53, 55, 57, 58, 64, 67, 70 and 74 to 83, inclusive, of this act become effective on January 16, 2013.

LEADLINES OF REPEALED SECTIONS

- 294A.281 Registration.
- 294A.282 Registered agent.

294A.283 Reporting of contributions and expenditures; period covered; deadline; form; filing.

294A.284 Reporting of certain information concerning compensation of persons to circulate petitions.

Assemblyman Segerblom moved that the Assembly concur in the Senate Amendment No. 567 to Assembly Bill No. 82.

Remarks by Assemblymen Segerblom, Hickey, and Flores.

Motion carried by a constitutional majority.

Bill ordered enrolled.

UNFINISHED BUSINESS

SIGNING OF BILLS AND RESOLUTIONS

There being no objections, the Speaker and Chief Clerk signed Assembly Bills Nos. 2, 117, 152, 202, 204, 212, 232, 291, 294, 358, 374, 384, 388, 390, 463, 489, 530; Assembly Joint Resolution No. 6; Assembly Resolution No. 9; Senate Bill No. 257.

GUESTS EXTENDED PRIVILEGE OF ASSEMBLY FLOOR

On request of Assemblyman Daly, the privilege of the floor of the Assembly Chamber for this day was extended to the following students and chaperones from Roger Corbett Elementary School: Carlos Acosta, Mario Alverez, Diego Aguilar, Noemi Almanza, Ulises Beccerra, Nestor Aguirre, Daniel Bailey, Jareth Bernardino, Miah Barnes, Lonnie Billoups, Diego DeLaTorre, Axel Becerra, Angelo Buenrostro, Juan Flores, Bryan Chagolla, Jason Escobar, Litzi Gonzalez, Jesus DeDios, Fatima Estrada, Anthony Guerrero, Alexandria Erwin, Manuel Flores, Luis Hernandez, Juan Gallegos, Maria Jackie Flores, Jonathan Lopez, Julia Huerta, Laurel Grabot-Fraser, Valeria Lopez, Yasmin Lopez, Jose Guerrero, Miguel Lopez, Juan Martinez, Jorge Martinez, Sergio Mellin, Anthony Martinez, Kimberly Moreno, Eufemia Perez, Ricardo Morales, Juan Mosqueda, Jorge Ramos, Erika Nava Duran, Fanny Munoz, Amiyah Ringer-Drumm, Fernando Ojeda, Juan Perez, Erika Rios, Jessica Pintor-Villanueva, Jeremiah Rogers, Stephanie Soria Padilla, Steffany Rios, Alejandra Sanchez, Miguel Valencia, Karina Rubio, Tony Soriano, Irlanda Vazquez, Jordin Saldana, Marisol Vargas, Jimmy Vega, Luis Silva, Paola Zamora, Valarie Solorio, Karen Shaw, Don Bounds, and Lori Pomajzl.

On request of Assemblyman Hickey, the privilege of the floor of the Assembly Chamber for this day was extended to Shinae Watterson.

On request of Assemblyman Livermore, the privilege of the floor of the Assembly Chamber for this day was extended to the following students and

chaperones from Freemont Elementary School: Juana Casteneda, Siana Escobosa, Lesly Flores, Khayman Garland, Kahle Good, Baylee Hale, Jodi Ignacio, Thomas Kreider, Jarrod Meyer, Andrew Moreira, Jennifer Munoz Marquez, Skylor Olshefsky, Destinae Ordaz, McKaylynn Petersen, Sophia Peterson, Kevin Pinto-Negrete, Caitlin Rose, Vyctoria Sessions, Kelvin Smith, Isaac Stuhlmiller, Joshua Verde Cabrera, Gustavo Villegas Espino, Karen Hellman, Mary Berge, Laina Joy Aquino, Angel Chavez, Evan Cherpeski, Ariana Chipres, Garret Damico, Andrew Deleon, Ashley Harris, Kristian Jachimiak, Axel Llamas Aranda, Regan Lowe, Chris Luna, Mackenzie Malone, Cassandra McGregor, Erika Nauyoks-Bolanos, Megan Owens, Amber Parman, Hunter Rauh, John Rupert, Megan Smith, Aryanna Thierry, William Turner, Angel Valdez, Leahsena Vitale, Nathan White, and Trinity Woodward.

On request of Assemblywoman Mastroluca, the privilege of the floor of the Assembly Chamber for this day was extended to Addie Schiller.

Assemblyman Conklin moved that the Assembly adjourn until Friday, June 3, 2011, at 11 a.m.

Motion carried.

Assembly adjourned at 12:08 p.m.

Approved:

JOHN OCEGUERA
Speaker of the Assembly

Attest: SUSAN FURLONG

Chief Clerk of the Assembly